Ferroelectric Materials and Artificial Dielectric Layer Structures for Microwave Integrated Circuit Technologies

Yue Ma

Ferroelectric Materials and Artificial Dielectric Layer Structures for Microwave Integrated Circuit Technologies

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Technische Universiteit Delft, op gezag van de Rector Magnificus Prof. Ir. K. C. A. M. Luyben, voorzitter van het College voor Promoties, in het openbaar te verdedigen

op 22 maart 2011 om 15:00 uur

door

Yue MA

Master of Engineering van Tsinghua University, Beijing, China geboren te Tianjin, China.

Dit proefschrift is goedgekeurd door de promotor(en): Prof. Dr.-Ing. J. N. Burghartz

Copromotor: Prof. Dr. B. Rejaei-Salmassi

Samenstelling promotiecommissie:

Rector Magnificus, voorzitter
Prof. Dr.-Ing. J. N. Burghartz, Institute for Microelectronics Stuttgart (IMS CHIPS), promotor
Prof. Dr. B. Rejaei-Salmassi, Sharif University of Technology, copromotor
Prof. Dr. J. R. Long, Technische Universiteit Delft
Prof. Dr. Ir. P. M. Sarro, Technische Universiteit Delft
Prof. Dr. Y. Zhuang, Wright State University
Prof. Dr. Ir. L. K. Nanver, Technische Universiteit Delft
Dr. R. Woltjer, NXP Semiconductors

Yue Ma,

Ferroelectric Materials and Artificial Dielectric Layer Structures for Microwave Integrated Circuit Technologies,

Ph.D. Thesis Delft University of Technology.

ISBN: 978-90-8570-732-5

Copyright © 2011 by Yue Ma

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means without the prior written permission of the copyright owner.

Printed by CPI Wöhrmann Print Service in the Netherlands.

Contents

1 Introduction

1.1	Motivation and background	1
1.2	Ferroelectrics and their applications in microwave engineering	5
1	1.2.1 Dielectrics and ferroelectrics	5
]	1.2.2 Applications of ferroelectrics	9
1	1.2.3 Microwave ferroelectric components	.10
1.3	Artificial dielectric materials	13
1.4	Outline of the thesis	15
Refe	erences	16

2 Ferroelectric Materials for Integrated Tunable Microwave Components

2.1	Introduction	23
2.2 thir	Chemical solution deposition of Barium Strontium Titanate	(BST) 26
2.3	BST thin films characterization	28
	2.3.1 Structural characterization2.3.2 Electrical characterization	28 31
2.4	Tunable components by substrate transfer technology (STT)	36
	2.4.1 Introduction2.4.2 STT process for ferroelectric tunable components	36 37
2.5	Experimental results and discussions	39
2.6	Conclusions	43

3 Tunability Enhancement of Ferroelectric Varactors Using Nano-dot Electrodes

3.1	Introduction	47
3.2	Basic principles of the ferroelectric nano-dot varactor	49
3.3	Nano-dot varactor arrays	59
3.4	Discussion and conclusion	65
Refer	rences	66

4 On-chip Passives with Artificial Dielectric Layer (ADL) Shields

4.1 Introduction
4.2 Integrated coplanar waveguide (CPW) transmission lines with ultra-high-k dielectric shields
4.3 ADL as a shield
4.3.1 Concept of ADL764.3.2 Shielding effect of ADL81
4.4 Experimental results of CPW lines
4.4.1 CPW with ordinary ADL shields824.4.2 CPW with micro-patterned ADL shields904.4.3 CPW with multi-layered ADL shields94
4.5 Applications of shielded CPW lines
4.5.1 Principles of directional couplers974.5.2 Branch-line couplers.994.5.3 Ring hybrids103
4.6 Application of ADL shields to spiral inductors106

4.6.1 Principles of operation	
4.6.2 Results and discussions	
References	

5 On-chip Artificial Dielectric Resonators for Microwave Applications

5.1	Introduction			
5.2	Theoretical analysis of artificial dielectric resonator			
5	.2.1 Configuration and theoretical model	119		
5	.2.2 Equivalent circuit model			
5	.2.3 Numerical simulations			
5	5.2.4 Effects of electric and magnetic polarization loss of ADL			
5	.2.5 Restrictions of the model	137		
5.3	Full wave simulations of ADL resonators			
5.4	Experimental results and discussions	147		
5.5	Conclusions			
Refe	References			

6 Conclusions and Recommendations

6.1	Conclusions	153
6.2	Recommendations	155

A	Ferroelectric Phase Transition	157
	References	.160

B Process	Flow for	the Fabricati	on of Microw	ave Ferroelectric
Devices	5	•••••	•••••	

С	Derivation of effective magnetic susceptibility of ADL 165	
	References	
D	Derivation of shielding effect of ADL	
E	Extraction of Transmission Line Parameters	
	References	
Su	ımmary	
Sa	menvatting	
Ac	cknowledgements	
Ał	bout the Author	

Chapter 1

Introduction

1.1 Motivation and background

The emergence of monolithic microwave integrated circuits (MMICs) in the past two decades has caused a dramatic change in our everyday life and experience. Today, MMIC technology is a multi-billion dollar industry and the backbone of our information and communication society. With the efforts of thousands of microwave scientists and engineers, cell phone has gone from an expensive, heavy and large item to a portable communication tool in everyone's pocket, even as small as a wrist watch (see Fig.1.1), which allows people to communicate from almost anywhere. The impact of the MMIC revolution, however, has not remained confined to the sphere of personal communication. Modern ultra-compact communication systems provide a wide variety of services from high-quality audio and video to high-speed internet and wireless access. The transfer of such high data rates requires high-speed data links which could not have been realized without miniature MMIC transceivers. Well known examples are local area networks (WLAN) and wireless personal area networks (WPAN) used by modern personal digital

1

assistants (PDAs) and notebook computers.

The advent of low-cost commercial radar for sensing and position detection is another major driver of the MMIC technology. Automotive wireless (e.g., radar) sensors are revolutionizing driving safety and automated traffic control [1-5]. Already, radar-based systems providing adaptive cruise control are being fitted to luxury cars, and this market alone is expected to witness a nearly 20-fold increase in near future [6]. The ever increasing popularity of cheap, light and compact navigation instruments utilizing the global positioning system (GPS) is another application area where MMICs are indispensable.



Figure 1.1: One of world's first cell phones held by inventor Martin Cooper, as compared to the wrist watch mobile phone packing a 1.5-inch touchscreen with tiny built-in stylus, bluetooth 2.0, MP4 and MP3 players.

Historically, the origin of the MMIC technology can be traced back to the development of planar transmission lines and solid state devices [7]. Over the last 50 years, driven by the IC-technology trend of smaller size, lighter weight, lower cost and increased complexity, huge amounts of research interests have been attracted in this direction [7,8]. Initially, GaAs was the substrate of choice for microwave integrated circuits because of its high electron mobility, leading to faster transistors, and its semi-insulating electrical characteristics, making it a suitable vehicle for microwave passive components. However, GaAs device manufacturing is very difficult and transistors on GaAs are very fragile and easily damaged by overheating or electrical overload [9].

Due to these shortcomings and the already existing enormous market for digital silicon-based ICs, silicon has been replacing GaAs as the mainstream MMIC technology. The maturity of silicon technology makes it extremely cost effective for mass production [10] and offers the further possibility to combine microwave circuits with digital circuits on the same chip. Moreover, excellent microwave device characteristics have been demonstrated with silicon-based SiGe and CMOS transistors with cut-off frequencies exceeding 500 GHz [11]. These results are very promising for active devices to operate at microwave frequencies, while being manufactured on a low cost, highly integrated silicon platform. Nevertheless, the shift towards silicon technology has come at a price. What distinguishes microwave integrated circuits from digital VLSI circuitry is the extensive use of passive components such as capacitors, inductors and transmission lines to implement various functions like impedance matching, phase shifting, and filtering [9]. But the migration of integrated passive microwave components to silicon has caused a number of serious hurdles.

First, unlike CMOS transistors, passive components do not readily shrink with each new generation of processing technology. Lumped inductors and capacitors are per se energy storage devices, which by nature consume relatively large chip area. Besides, the length of transmission lines used in microwave circuit, such as quarter wavelength impedance transformer, is normally comparable with the wavelength at frequencies of interest, sometimes even comparable or larger than the chip size. In addition, integrated passive components on ordinary silicon suffer from substrate loss and crosstalk due to the electromagnetic filed penetration into the conductive silicon, which results in severe performance degradation at microwave frequencies [12]. Finally, compared to conventional logic application, the new functionality of electrical tuning is required for microwave integrated circuits. Tunable components like varactors are widely used in impedance matching, voltage controlled oscillators (VCO), filters and phase arrayed antennas. Although silicon-based semiconductor varactors are available for tuning purposes, their quality factors (Q) typically decrease with increasing frequency. As a result, semiconductor varactors with sufficiently high Q and tunability are still a big challenge for frequencies above 10-20GHz [13].

Over the past decade, various solutions have been proposed to address these issues. In particular, attention has been focused on reducing the loss and crosstalk caused by the conducting silicon material through the use of high-resistivity substrates [14-16], local

removal of silicon (silicon micromachining) [17,18], substrate transfer]19,20], and the use of esoteric technologies such as very thick dielectric layers [21-23] and proton bombardment [24]. Micromachining techniques have also been employed to realize micro-electro-mechanically tunable components (MEMs) [25,26] and potentially to improve the quality of conventional silicon-based varactors.

Nevertheless, by comparison, much less attention has been paid to the use of materials with extraordinary electromagnetic properties as a potential remedy for the issues mentioned above. One exception is the recent effort to reduce the area consumption of on-chip inductors and transmission lines by the implementation of integrated ferromagnetic thin films with high permeability. Such films have been demonstrated to significantly increase the inductance of on-chip solenoid [27] and spiral [28,29] inductors compared with their air core counterparts. Integrated microstrip transmission lines with thin ferromagnetic cores have also been shown to exhibit a shorter propagation wavelength, therefore leading to a smaller device size [30,31].

The objective of this thesis is to explore the application of two other classes of materials in integrated microwave passive components: ferroelectrics and artificial dielectrics. The interest in ferroelectric materials in microwave integrated circuits is primarily due to a high dielectric constant which also depends on the electric field. With dielectric constant possibly as high as a few hundreds even in thin film form, the metal-insulator-metal (MIM) capacitor can be made substantially small while still maintaining an adequate capacitance density value. The recognition of potential applications of ferroelectric materials for microwave tunable devices dates back about fifty years ago. It's until the last decade that intensive research efforts were made in this direction, driven by the miniaturization of microwave components and possible integration in microwave integrated circuits, due to the development of ferroelectric thin and thick film deposition technologies [32].

Likewise, the utilization of artificial dielectrics in microwave engineering is not a new concept. Since the 2nd world war, artificial dielectrics have been studied for use in light weight lenses for antenna beam shaping [33-38]. These materials comprise three dimensional arrays of miniature metallic particles immersed in an insulating dielectric medium [39]. Their dielectric constant can be (locally) engineered by changing the size, shape, and spacing of the particles. However, monolithic integration of artificial dielectrics offers the additional benefit of enhancing the range of material constants and

realizing novel media by exploiting the (sub-) micron patterning and thin film deposition techniques. As such, the exploration of integrated artificial dielectrics and devices based on them is a very new research area which will be explored in this work.

1.2 Ferroelectrics and their applications in microwave engineering

1.2.1 Dielectrics and ferroelectrics

In the so-called dielectric materials, when an electric field is applied, the ions (cations and anions) move in different directions and/or electron clouds deform, causing electric dipole moment (see Fig.1.2). This phenomenon is known as electric polarization of the dielectric, and the polarization (P) is expressed quantitatively as the sum of the electric dipole moments per unit volume. According to Poisson's equation, free charges are sources of the electric displacement (D) field

$$\nabla \cdot \boldsymbol{D} = \boldsymbol{\rho}_{\text{free}} \,, \tag{1.1}$$

where ρ_{free} is the density of free charges. The electric displacement **D** is related to the electric field (**E**) and polarization **P** by the following expression:

$$\boldsymbol{D} = \boldsymbol{\varepsilon}_0 \boldsymbol{E} + \boldsymbol{P} \,. \tag{1.2}$$

Here, ε_0 is known as the permittivity of vacuum. Most dielectric materials polarize linearly with external electric filed, nonlinearities are negligible. This is usually expressed mathematically as:

$$\boldsymbol{P} = \varepsilon_0 \chi_{ij} \boldsymbol{E} , \qquad (1.3)$$

where χ_{ij} is called the dielectric susceptibility of the material and is generally a constant second rank tensor. It follows from (1.2) and (1.3) that

$$\boldsymbol{D} = \varepsilon_0 \boldsymbol{E} + \boldsymbol{P} = \varepsilon_0 \boldsymbol{E} + \varepsilon_0 \chi_{ij} \boldsymbol{E} = \varepsilon_0 (\delta_{ij} + \chi_{ij}) \boldsymbol{E} = \varepsilon_0 \varepsilon_{ij} \boldsymbol{E}, \qquad (1.4)$$

where $\varepsilon_{ij} = \delta_{ij} + \chi_{ij}$ is called the relative permittivity or relative dielectric constant and δ_{ij} is Kronecker's delta (ie. $\delta_{ij} = 1$ for i = j and $\delta_{ij} = 0$ for $i \neq j$). In homogeneous isotropic materials, both χ_{ij} and ε_{ij} reduce to scalar constants known as χ and ε_r respectively, which appear in most textbooks [40].



Figure 1.2: Schematics of ionic polarization and electronic polarization.

Depending on the crystalline structures, the positive and negative charge centers of certain materials may not coincide with each other even without an external electric field. Such materials are said to possess electric spontaneous polarization. Ferroelectric materials are commonly defined as crystals which show an electric spontaneous polarization and in which the direction of the polarization can be reversed by an external electric field, yielding a hysteresis loop. The term 'ferroelectric' is used in analogy to

'ferromagnetic', in which the material possesses permanent magnetic moment. Historically, this is because ferroelectricity was first experimentally identified in Rochelle salt (KNaC₄H₄O₆·4H₂O) by Valasek in 1920, when ferromagnetism was already known. Although the prefix ferro means iron, most ferroelectric materials do not have iron atoms in their lattice. Besides the hysteretic polarization similarity, also akin to ferromagnetic materials is the phase transition temperature of ferroelectrics, denoted by the Curie point T_C . Most ferroelectric materials undergo a structural phase transition from a high-temperature (above the Curie point) non-ferroelectric (also known as paraelectric) phase into a low-temperature (below the Curie point) ferroelectric phase (Fig.1.3). The crystalline symmetry of ferroelectric phase is always lower than the paraelectric phase. Above the Curie point, the relative dielectric constant ε_r falls off with temperature over a wide range in the paraelectric phase according to Curie-Weiss law

$$\mathcal{E}_r = \frac{C}{T - T_0},\tag{1.5}$$

where C is the Curie-Weiss constant, T_0 is Curie-Weiss temperature which is equal or slightly lower than the exact phase transition temperature T_C .

Ferroelectric phase transitions are often characterized as of being either displacive or order-disorder type, though often phase transitions will have behavior that contains elements of both types. The distinction is made in terms of whether the paraelectric phase is microscopically non-polar (displacive) or only non-polar in a macroscopic or thermally average sense (order-disorder). Among the many different ferroelectric materials, the ones with perovskite crystal structure are of particular interest for practical applications. Taking BaTiO₃ as an example, which is a typical displacive type and is well studied, Fig.1.4 shows both the cubic perovskite structure of paraelectric phase and tetragonal structure of ferroelectric phase (although cubic BaTiO₃ crystal transforms successfully to three ferroelectric phases as the temperature goes down, only the first one below the Curie point with tetragonal structure is shown here). In the cubic phase, there is obviously no spontaneous polarization from the symmetry consideration. In the tetragonal phase, the Ti and O ions move relative to Ba at the origin from their cubic positions, thus spontaneous polarization along [001] direction occurs [40].



Figure 1.3: Schematics of (a) relative dielectric constant as a function of temperature in different phases; (b) hysteresis loop of polarization-electric filed in a ferroelectric phase and (c) nonlinear polarization-electric field relation in a paraelectric phase.



Figure 1.4: Crystal structure of BaTiO₃ for a cubic paraelectric phase without spontaneous polarization and a tetragonal ferroelectric phase with spontaneous polarization along [001] direction.

1.2.2 Applications of ferroelectrics

Ferroelectric materials are widely used in various devices especially in modern integrated circuits. A brief summary of applications of ferroelectric materials with an emphasis of thin film and IC technology is presented below.

The first impressive property of ferroelectric material to come to mind is perhaps the high permittivity. For example, single crystal BaTiO₃ shows a relative dielectric constant of more than 5000 at room temperature. Although dielectric constant of only a few hundred is observed for thin film form, it is still much higher than that of typical dielectric materials like SiO₂ (with a relative dielectric constant of around 4) widely used in IC technologies. It is the high dielectric constant that makes ferroelectric thin film very attractive for dynamic random access memory (DRAM) applications. As is well known, DRAM capacitor gets physically smaller with scaling, yet adequate capacitance per cell has to be maintained to get enough charge storage ability. For trench capacitor, it means deeper trench must be formed to increase the capacitor surface area, which substantially increase the process complexity and cost. For stacked capacitor, it is difficult to make the surface area indefinitely, therefore, ferroelectric materials like Barium Strontium Titanate (BST) are considered as candidates to replace SiO₂ as the dielectric layer in DRAM capacitor [41].

Ferroelectric materials in ferroelectric phase show hysteresis loop as depicted in Fig. 1.3(b). Basically, two different polarization states of different polarization directions can be identified. Because these two polarization states can be switched between each other with an external electric field, these two states can be used to represent binary states. Furthermore, since a ferroelectric maintains a remanent polarization at zero bias, this memory would require no refreshing, leading to non-volatile ferroelectric memory. A device incorporating a ferroelectric film as a capacitor to hold data is called ferroelectric random access memory (FeRAM). Lead zirconate titanate (PZT) has been proposed to be a good candidate for FeRAM applications [40]. However, both the PZT ferroelectric layer and the noble metals used as electrodes are not compatible with CMOS process integration and may cause contamination issues. Other applications of ferroelectric materials include piezoelectric devices, pyroelectric sensors, electrooptic devices etc. These are beyond the interests of this thesis and will not be discussed here.

Over the last twenty years, with the fast development in IC technologies, ferroelectric materials were extensively investigated and implemented in modern MMICs, mainly due to their extraordinary properties of high permittivity and/or strong field dependence of dielectric constant. One major application of ferroelectric materials in microwave regime is for lumped capacitors. As energy storage devices, capacitors are not readily scaled with logic circuit elements, therefore consume relatively large chip area. The implementation of ferroelectric materials with very high permittivity could potentially solve the problem since the stored electric energy in a material is proportional to its dielectric constant. In conventional GaAs MMIC's, off-chip large capacitors were provided with additional pins, leading to higher pin count and larger package size [42,43]. By integrating these capacitors on chip using conventional SiN as dielectric, the chip size would be extremely large. In contrast, on-chip BST thin film employed by metal organic decomposition (MOD) technique has been demonstrated to have a relative dielectric constant of 300, which is 50 times higher than that of SiN [43]. Therefore, substantial area reduction and more compact package size can be achieved with on-chip BST capacitors, resulting in miniaturization of the RF front-end block of mobile phone sets [43]. The comparisons of area consumption and package size of aforementioned technologies are illustrated in Fig.1.5.

As shown in Fig.1.3, nonlinear relations between electric polarization and electric field are observed in both ferroelectric phase and paraelectric phase, resulting in field dependant permittivity. This characteristic is commonly described by the material tunability, which is defined as the ratio of material permittivity at zero applied electric field to its permittivity at the maximum electric field [32]. This property, combined with relatively low loss at high frequencies, makes ferroelectrics attractive for tunable microwave devices, which is another major application of ferroelectric materials at microwave frequencies. In principle, all forms of ferroelectric materials, *i.e.* bulk, thick film and thin film, can be used for microwave tunable applications. Bulk or single crystalline ferroelectrics are preferred for tunable disc resonators where the largest permittivity can be achieved [44,45]. Thick ferroelectric films can be processed on common microwave substrates like Al₂O₃, in a cost effective way, by screen printing techniques or low temperature co-fired ceramics (LTCC) technology [46,47]. This is attractive to phase arrayed antennas where thousands of phase shifters can be easily

implemented by ferroelectric transmission lines. However, devices based on bulk or thick film ferroelectrics require very high voltages for effective tuning, *i.e.* of the order of hundreds of volts to tens of kilovolts, which hinders their integration in modern MMICs



Figure 1.5: Area comparison of MMIC in case of using SiN and BST capacitors. Adapted from [43].

Over the last decade, ferroelectrics in thin film form have attracted high research interest because of its low tuning voltage and its compatibility with modern IC technologies. Microwave tunable components, including varactors, tunable filters, voltage controlled oscillators, phase shifters and delay lines, have been realized already by using thin film ferroelectric materials [32]. Among the various ferroelectric materials, BST thin film has been investigated extensively. Compared to other ferroelectric

materials, BST thin film offers high tuning range (a tunability greater than 13 has been demonstrated [48]), low microwave losses (loss tangent could be as low as 0.005 at 20 GHz [49]), various standard thin film deposition technologies available on different substrates, and Curie temperature control by simply changing the Ba/Sr ratio.

In many works, BST thin film microwave tunable devices have been fabricated on single crystal substrates such as MgO, LaAlO₃ and sapphire (see for example [50] and [51]). This is because these substrates have extremely low loss at microwave frequencies. Epitaxial growth of BST thin film on these substrates offers the best quality of the film. However, these substrates are expensive and often only available at small sizes, in contrast silicon is the substrate choice for mass production. Besides, silicon also offers the possibility to co-integrate microwave and digital logic circuits. Recently, much effort has been made to realize BST thin film tunable devices on silicon.

The performance of BST thin film tunable microwave devices (tunability and loss) on silicon depends on various factors, *i.e.*, substrate conductivity, film property, stress, film-electrode interface, device topology and electrodes. Since ordinary silicon loss features high loss at microwave frequencies, high resistivity silicon (HRS) is usually chosen to minimize substrate loss contribution, and thus to achieve high quality factors. Large efforts have been made to optimize BST based tunable devices while considering the aforementioned aspects. For example, the film property can be optimized, from processing aspect, through morphology control [52], compositional design [53] and post deposition treatment [54]. Normally, the electrodes of parallel plate BST varactors are made by platinum (Pt) to withstand high temperature processing step and to offer minimal lattice mismatch. However, Pt has a relatively high resistivity which contributes to high conductor loss. It has been shown that by using a thick bottom electrode consisting of $Pt(50nm)/Au(0.5\mu m)$, the microwave losses associated with metal layers can be substantially reduced. BST varactors based on this technique have shown record Q factors of about 40 at 45 GHz, which is much higher than that of a silicon abrupt junction varactor diode [55]. Delay lines, phase shifters and voltage controlled oscillator (VCO) based on the high-O BST varactor technologies have been demonstrated [56-58]. Due to the enormous progresses made, ferroelectrics are gaining attention for microwave tunable applications and becoming commercialized [13]. However, the deposition of ferroelectric materials mostly requires high temperature process steps and platinum as electrode material, which are not well compatible with modern CMOS back-end-of-line (BEOL) technologies. Another issue related to ferroelectric based tunable components, especially when a thick ferroelectric film is used or a planar capacitor structure is adopted, is the high DC bias voltage (tens to hundreds of volt) required to gain a reasonable tuning range.

1.3 Artificial dielectric materials

Artificial materials, from a broad point of view, are man-made materials in contrast to natural materials which are products of nature. Plastics are perhaps the most well known man-made materials and are surrounding us in our everyday life.

In a narrow sense, artificial materials, sometimes also called metamaterials, are man-made structured composite materials that exhibit electromagnetic properties not available in natural materials. (It is interesting that the term metamaterial is even narrowly referred to artificial material with negative refractive index.) Artificial materials thus gain their electromagnetic properties from their structure rather than directly from their composites. In this thesis, we adopt this narrow meaning of artificial materials.

It is well known that the response of a system to an electromagnetic excitation is strongly dependent on the properties of the materials involved. Normally, we use the macroscopically defined parameters of permittivity and permeability to characterize materials under an electromagnetic field. In an ordinary material, the permittivity and permeability per se originate from the induced electric and magnetic dipole moment at the atomic level respectively by an external electromagnetic field. These two parameters are in general frequency dependent. In artificial materials, well-designed microscopic structures, which could generate electric and/or magnetic dipole moment (such as metal spheres in a three dimensional lattice), are embedded in another natural material. Although these structures are generally much larger than the size of an atom, as long as the wavelength of the electromagnetic wave is far more than the dimension of the individual structure, the induced dipole moments behave very similar to their atomic counterpart. As a result, the artificial material globally can be viewed as a continuous medium with averaged (effective) permittivity and permeability. The microscopic structures are easily realized by today's micro-fabrication technologies. The essence of artificial materials is to ignore the detail of electric and magnetic field distribution of the individual elements, but to consider the system as a continuous medium under electromagnetic field.

- 13 -

The first attempt of artificial materials may date back to the late nineteenth century [59]. The artificial materials today are widely investigated for a large frequency range, from radio-frequency to visible light, and can be roughly categorized according to their response to incident electromagnetic field as artificial dielectrics and artificial magnetics. Some artificial media, such as the so-called "double negative (DNG)" metamaterials (also the terms of "left-handed", "negative refraction index" and "backward-wave" media are commonly used for this kind of metamaterials), are both artificial dielectrics and artificial magnetics, this is because both the macroscopic permittivity and the permeability of these composites are originated from the structures rather than from the materials [39]. Furthermore, since the microscopic elements introduced as metallic spheres normally generate both electric and magnetic dipole moments under high frequency electromagnetic field, the differences between these two kinds of artificial materials are not stringent. Nevertheless, within a suitable frequency range, it is still valuable to make these distinctions to emphasize certain aspect of the artificial media in reaction to external electromagnetic field. Related to the scope of this thesis, only a short review of artificial dielectric is given below.

It is commonly recognized that Kock first developed artificial dielectrics in the late 1940s [39]. An artificial dielectric is physically a large scale model of a real dielectric. Normally, it consists of a large number of identical conducting elements periodically arranged in a three-dimensional or two-dimensional pattern within a host conventional dielectric. Under an electric field, the positive and negative charges within each single conducting obstacle displace with one another. Since the conducting obstacle is overall electrically neutral, the dominant part of the induced field by the displacement of the charges is a dipole field. Thus, each obstacle simulates the behaviors of an atom or a molecule in a natural dielectric and exhibits an electrical dipole moment [39].

Typical artificial dielectric structures include metallic spheres or disks in a three dimensional lattice, and metallic strips or rods in a two dimensional periodic pattern [39]. It should be noticed that, unless the metallic obstacles are of spherical shape and are embedded in a cubic or random lattice pattern, the induced polarization value is different for a fixed applied electric field in different directions. Thus artificial dielectrics in general are anisotropic.

Artificial dielectrics are extensively applied in microwave engineering. As for Kock's work, the early period of the development of artificial dielectrics is for lens antenna

applications. Compared with a lens antenna using conventional dielectrics, a lens antenna based on artificial dielectrics offers the advantages of low cost and low weight [39]. In recent years, the DNG metamaterials have attracted attention of a wide audience. In a sense, a DNG metamaterial is an artificial dielectric as well as an artificial magnetic, showing both negative value of permittivity and permeability in a certain frequency range [59]. Recently, artificial dielectric resonator, composed of small metal strips and put in a waveguide type cavity, was proposed by Kubo's group [60]. A Q factor over 1000 and effective permittivity over 800 of the artificial dielectric resonator was demonstrated. On chip $\lambda/4$ transmission line resonator built on top of artificial dielectric was demonstrated by Huang et al., and was shown to substantially reduce the chip area occupied by the resonator because the high permittivity of the artificial dielectric and thus a short wavelength of the transmission line [61].

15

In this thesis, a new type of artificial dielectric structure is introduced. The building blocks are small metal squares embedded in conventional dielectrics. The structures can be easily realized by modern micro-fabrication technologies. We will show that an in plane effective dielectric constant of tens of thousands (could be called super high-k) can be achieved by the proposed structures. Applications of the artificial dielectric structure to on-chip microwave passives shield and on-chip high-Q resonator will be demonstrated.

1.4 Outline of the thesis

This thesis focuses on: firstly, tunable functionality of microwave passive components based on ferroelectric materials, including both the material preparation and possible device realization; and secondly, loss reduction of on-chip passive components with artificial dielectric structure. The contents of the thesis can be summarized as follows:

Chapter 2 describes chemical solution deposition (CSD) method for BST thin film preparation. The microstructure of deposited BST thin film is characterized by atomic force microscopy (AFM) and scanning electron microscopy (SEM). The dielectric constant as a function of applied voltage is obtained by C-V measurement of parallel plate capacitor. It also presents possible tunable device designs based on BST thin film. A device fabrication flow based on substrate transfer technology is proposed. Issues related to the device performance are analyzed.

Chapter 3 presents a new ferroelectric based varactor design by implementing nano-dot electrodes which is aimed to enhance the tunability of the varactor at relatively low bias

voltage. Theoretical studies of single dot varactor and nano-dot varactor arrays are carried out. It is shown that near the dot electrode, the electric field distribution is determined by the dot dimension rather than the ferroelectric material thickness. The large value of electric field adjacent to the dot gives rise the high tuning range.

Chapter 4 introduces a new artificial dielectric structure consisting of a conventional dielectric thin film sandwiched between two patterned metal layers. It is shown that substrate losses of on-chip microwave passive components, such as CPW transmission lines and spiral inductors, built on ordinary medium resistivity silicon substrate can be effectively reduced by the proposed structure. The artificial dielectric structure acts as an electric shield to substantially reduce undesired electric field penetrating into the conductive silicon substrate.

Chapter 5 presents a new design of on-chip artificial dielectric resonator coupled with a metallic ring. Simulations using effective media parameters of the artificial dielectric structure yield promising high Q on microwave substrate like glass and HRS. The artificial dielectric resonator is theoretically studied and experimentally verified, although the achieved Q is less than expected.

Chapter 6 provides a summary of conclusions of the presented research topics, along with a number of recommendations for future research.

Appendix A provides a short introduction of basic physics of ferroelectric phase transition.

Appendix B presents the flowchart for the fabrication of BST thin film based tunable devices used in this work.

Appendix C and D presents detail calculations and derivations which have not been included in the main text for brevity reasons.

Finally, **Appendix E** describes details of transmission line parameters extraction procedures.

References

 T. Shimura, Y. Kawasaki, Y. Ohashi, K. Shirakawa, T. Hirose, S. Aoki, H. Someta, K. Makiyama, S. Yokokawa, "76GHz flip-chip MMICs for automotive radars", IEEE RFIC Symp. Digest, pp. 25-28, 1998.

- [2] I. Gresham, N. Jain, T. Budka, A. Alexanian, N. Kinayman, B. Ziegner, S. Brown, P. Staecker, "A 76-77GHz pulsed-doppler radar module for autonomous cruise control applications", IEEE MTT-S Digest, pp. 1551-1554, 2000.
- [3] M. Walden, A. Garrod, "A european low cost mmic based millimetre-wave radar module for automotive applications", 33rd European Microwave Conference, pp. 1195-1198, Munich, Germany, 2003.
- [4] C.N. Rheinfelder, S. Lindenmeier, J.-F. Luy, C. willner, A. Schuppen, "A sensor system based on SiGe MMIC's for 24GHz automotive applications", IEEE MTT-S Digest, pp. 2239-2242, 2001.
- [5] I. Gresham, A. Jenkins, R. Egri, C. Eswarappa, F. Kolak, R. Wohlert, J. Bennett, J. Lanteri, "Ultra wide band 24GHz automotive radar front-end", IEEE RFIC Symp. Digest, pp. 505-508, 2003.
- [6] J. Powell, D. Bannister, "Business prospects for commercial mm-wave MMICs", IEEE Microwave Magazine, Vol. 6, No. 4, pp. 34-43, 2005.
- [7] H. Howe, "Microwave integrated circuits an historical perspective", IEEE Trans. Microwave Theory Tech., Vol. 32, No. 9, pp. 991-996, 1984.
- [8] E.C. Niehenke, R.A. Pucel, I.J. Bahl, "Microwave and millimeter-wave integrated circuits", IEEE Trans. Microwave Theory Tech., Vol. 50, No. 3, pp. 846-857, 2002.
- [9] I.D. Robertson, S. Lucyszyn, "RFIC and MMIC Design and Technology", Antony Rowe Ltd, Chippenham, Wiltshire, 2001.
- [10] J.N. Burghartz, "Status and trends of silicon RF technology", Technical Digest ESSDERC, pp. 56-63, 1999.
- [11] R. Krithivasan, Y. Lu, J.D. Cressler, J.S. Rieh, M.H. Khater, D. Ahlgren, G. Freeman, "Half-terahertz operation of SiGe HBTs", IEEE Electron Device Lett., Vol 27, No. 7, pp. 567-569, 2006.
- [12] J.N. Burghartz, "RF passives on silicon the intended and the unintended", 2006 BCTM Short Course; Course on IEEE Expert Now, 2008.
- [13] N. Setter, D. Damjanovic, L. Eng, G. Fox, S. Gevorgian, S. Hong, A. Kingon, H. Kohlstedt, N.Y. Park, G.B. Stephenson, I. Stolitchnov, A.K. Taganstev, D.V. Taylor, T. Yamada, S. Streiffer, "Ferroelectric thin films: review of materials, properties, and applications", J. Appl. Phys., Vol.100, pp. 051606/1-46, 2006.
- [14] J.N. Burghartz, D.C. Edelstein, K.A. Jenkins, Y.H. Kwart, "Spiral inductors and

transmission lines in silicon technology using copper-damascene interconnects and low-loss substrate", IEEE Trans. Microwave Theory Tech., Vol.45, No.10, pp. 1961-1968, 1997.

- [15] B. Rong, J.N. Burghartz, L.K. Nanver, B. Rejaei, M. v.d. Zwan, "Surface-passivated high- resistivity silicon substrates for RFICs", IEEE Electron Dev. Lett., Vol. 25, No. 4, pp. 176-178, 2004.
- [16] A.C. Reyes, S.M. El-Ghazaly, S.J. Dorn, M. Dydyk, D.K. Schroder, H. Patterson, "Coplanar waveguides and microwave inductors on silicon substrates", IEEE Trans. Microwave Theory Tech., Vol. 43, No. 9, pp. 2016-2022, 1995.
- [17] J.W. Lin, C.C. Chen, Y.T. Cheng, "A robust high-Q micromachined RF inductor for RFIC applications", IEEE Trans. Electron Dev., Vol. 52, No. 7, pp. 1489-1496, July 2005.
- [18] T. Wang, Y.S. Lin, S.S. Lu, "An ultra low-loss and broadband micromachined RF inductor for RFIC input-matching applications", IEEE Trans. Electron Dev., Vol. 53, No. 3, pp. 568-570, 2006.
- [19] R. Dekker, P. Baltus, H. Mass, "Substrate transfer for RF technologies", IEEE Trans. Electron Dev., Vol. 50, No. 3, pp. 747-757, 2003.
- [20] K. Buisman, L.K. Nanver, T.L.M. Scholtes, H. Schellevis, L.C.N. de Vreede, "High-performance varactor diodes integrated in a silicon-on-glass technology", Proc. ESSDERC, pp. 117-120, 2005.
- [21] B.K. Kim, B.K. Ko, K. Lee, J.W. Jeong, K.S. Lee, S.C. Kim, "Monolithic planar RF inductor and waveguide structures on silicon with performance comparable to those in GaAs MMIC", Techn. Dig. IEDM, pp. 717-720, 1995.
- [22] X. Huo, K. J. Chen, P.C.H. Chan, "Silicon-based high-Q inductors incorporating electroplated copper and low-k BCB dielectric", IEEE Electron Dev. Lett., Vol. 23, No. pp. 520-522, 2002.
- [23] G.J. Carchon, W. De Raedt, E. Beyne, "Wafer-level packaging technology for high-Q on-chip inductors and transmission lines", IEEE Trans. Microwave Theo. Tech., Vol. 52, No. 4, pp. 1244-1251, 2004.
- [24] K.T. Chan, C.H. Huang, A. Chin, M.F. Li, D.L. Kwong, S.P. McAlister, D.S. Duh, W.J. Lin, "Large Q-factor improvement for spiral inductors on silicon using proton implantation", IEEE Microwave Wireless Compon. Lett., Vol. 13, No. 11, pp.

460-462, 2003.

- [25] L. Dussopt, G.M. Rebeiz, "Intermodulation distortion and power handling in RF MEMS switches, varactors and tunable filters", IEEE Trans. Microwave Theo. Tech., Vol. 51, No. 4, pp. 1247-1256, 2003.
- [26] D. Girbau, N. Otegi, L. Pradell, A. Lazaro, "Study of intermodulation in RF MEMS variable capacitors", IEEE Trans. Microwave Theo. Tech., Vol. 54, No. 3, pp. 1120-1130, 2006.
- [27] Y. Zhuang, M. Vroubel, B. Rejaei, J.N. Burghartz, "Integrated RF inductors with micro-patterned NiFe core", Solid-State Electron., Vol. 51, No. 3, pp. 405-413, 2007.
- [28] M. Yamaguchi, M. Baba, K.-I. Arai, "Sandwich-type ferromagnetic RF integrated inductor", IEEE Trans. Microwave Theory Tech., Vol. 49, No. 12, pp.2331-2335, 2001.
- [29] A.M. Crawford, D. Gardner, S.X. Wang, "High-frequency microinductors with amorphous magnetic ground planes", IEEE Trans. Magn., Vol. 38, No. 5, pp. 3168-3170, 2002.
- [30] P.K. Amiri, B. Rejaei, Y. Zhuang, M. Vroubel, J.N. Burghartz, "Ferromagnetic thin films for loss reduction in on-chip transmission lines", IEEE Trans. Magn., Vol. 43, No. 6, pp. 2630-2632, 2007.
- [31] P.K. Amiri, B. Rejaei, Y. Zhuang, M. Vroubel, D.W. Lee, S.X. Wang, J.N. Burghartz, "Integrated on-chip microstrip lines with Co-Ta-Zr magnetic films", IEEE Intermag Digest, pp. 514-, 2008.
- [32] A.K. Tagantsev, V.O. Sherman, K.F. Astafiev, J. Venkatesh, N. Setter, "Ferroelectric materials for microwave tunable applications", J. Electroceram., Vol. 11, No. 1-2, pp. 5-66, 2003.
- [33] W.E. Kock, "Metal-lens antennas", Proc. IRE, Vol. 34, No. 11, pp. 828-836, 1946.
- [34] W.E. Kock, "Metallic delay lenses", Bell Syst. Tech. J., Vol. 27, pp. 58-82, 1948.
- [35] J. Brown, "Artificial dielectrics", Prog. Dielect., Vol. 2, pp. 195-225, 1960.
- [36] I.J. Bahl, K.C. Gupta, "A leaky wave antenna using an artificial dielectric medium", IEEE Trans. Antennas Propag., Vol. 22, No. 1, pp. 119-122, 1974.
- [37] I.J. Bahl, K.C. Gupta, "Frequency scanning leaky-wave antenna using artificial dielectrics", IEEE Trans. Antennas Propag., Vol. 23, No. 4, pp. 584-589, 1975.

- [38] A.C. Brown, "Pattern shaping with a metal-plate lens", IEEE Trans. Antennas Propag., Vol. 28, No. 4, pp. 564-568, 1980.
- [39] R.E. Collin, "Field Theory of Guided Waves", 2nd Edition, IEEE Press, New York, 1992.
- [40] R. Waser (Ed.), "Nanoelectronics and Information Technology", Wiley-VCH, Weinheim, Germany, 2003.
- [41] S. Ezhilvalavan, T.-Y. Tseng, "Progress in the developments of (Ba,Sr)TiO3 (BST) thin films for Gigabit era DRAMs", Mater. Chem. Phys., Vol. 65, No. 3, pp. 227-248, 2000.
- [42] H. Koizumi, A. Noma, T. Tanaka, K. Kanazawa, D. Ueda, "A GaAs MMIC chip-set for mobile communications using on-chip ferroelectric capacitors", IEEE J. Solid-State Circuits, Vol. 31, No. 6, pp. 835-840, 1996.
- [43] D. Ueda, "Implementation of GaAs monolithic microwave integrated circuits with on-chip BST capacitors", J. Electroceram., Vol. 3, No. 2, pp. 105-113, 1999.
- [44] A. Deleniv, A. Eriksson, S. Gevorgian, "Design of narrow-band tunable band-pass filters based on dual mode SiTiO₃ disc resonators", 2002 IEEE MTT-S Digest, Vol.2, pp. 1197-1200.
- [45] A. Deleniv, A. Eriksson and S. Gevorgian, "Four-pole tunable band-pass filter based on two dual mode SiTiO₃ disc resontors", 32nd Europ. Microw. Conf. 2002, pp. 1-3.
- [46] W. Hu, D. Zhang. M.J. Lancaster, T.W. Button, B. Su, "Investigation of ferroelectric thick-film varactors for microwave phase shifters", IEEE Trans. Microw. Theo. Tech., Vol. 55, No. 2, pp. 418-424, 2007.
- [47] A. Deleniv, S. Gevorgian, H. Jantunnen, T. Hu, "LTCC compatible ferroelectric phase shifters", 2005 IEEE MTT-S Digest, Vol. 2, pp. 599-602.
- [48] N.K. Pervez, P.J. Hansen, R.A. York, "High tunability barium strontium titanate thin films for rf circuit applications", Appl. Phys. Lett., Vol. 85, No. 19, pp. 4451-4453, 2004.
- [49] D.E. Kotecki, J.D. Baniecki, H. Shen, R.B. Laibowitz, K.L.Saenger, J.J. Lian, T.M. Shaw, S.D. Athavale, C. Cabral, Jr., P.R. Duncombe, M. Gutsche, G. Kunkel, Y.-J. Park, Y.-Y. Wang, R. Wise, "(Ba,Sr)TiO₃ dielectrics for future stacked-capacitor DRAM", IBM J. Research Development, Vol. 43, No. 3, pp. 367-382, 1999.
- [50] J. Bellotti, E.K. Akdogan, A. Safari, W. Chang, S. Kirchoefer, "Tunable dielectric

properties of BST thin films for RF/MW passive components", Integr. Ferroelectr., Vol. 49, No. 1, pp. 113-122, 2002.

- [51] D. Kim, Y. Choi, M. Ahn, M.G. Allen, J.S. Kenney, P. Marry, "2.4GHz continuously variable ferroelectric phase shifters using all-pass networks", IEEE Microw. Wirel. Compon. Lett., Vol. 13, No. 10, pp. 434-436, 2003.
- [52] S. Hoffmann, R. Waser, "Control of the morphology of CSD-prepared (Ba,Sr)TiO3 thin films", J. Euro. Ceram. Society, Vol. 19, No. 6, pp. 1339-1343, 1999.
- [53] M.W. Cole, W.D. Nothwang, C. Hubbard, E. Ngo, M. Ervin, "Low dielectric loss and enhanced tunability of Ba_{0.6}Sr_{0.4}TiO₃ based thin films via material compositional design and optimized film processing methods", J. Appl. Phys., Vol. 93, No. 11, pp. 9218-9225, 2003.
- [54] S. Halder, T. Schneller, R. Meyer, R. Waser, "Effect of ozone treatment on the electrical properties of (Ba0.7Sr0.3)TiO3 thin films", J. Appl. Phys., Vol. 97, pp. 114904/1-5, 2005.
- [55] A. Vorobiev, P. Rundqvist, K. Khamchane, S. Gevorgian, "Silicon substrate integrated high Q-factor paralle-plate ferroelectric varactors for microwave/millimeterwave applications", Appl. Phys. Lett., Vol. 83, No. 15, pp. 3144-3146, 2003.
- [56] D. Kuylenstierna, A. Vorobiev, P. Linner, S. Gevorgian, "Ultrawide-band tunable true-time delay lines using ferroelectric varactors", IEEE Trans. Microw. Theo. Tech., Vol. 53, No. 6, pp. 2164-2170, 2005.
- [57] D. Kuylenstierna, A. Vorobiev, P. Linner, S. Gevorgian, "Composite right/left handed transmission line phase shifter using ferroelectric varactors", IEEE Microw. Wirel. Compon. Lett., Vol. 16, No. 4, pp. 167-169, 2006.
- [58] M. Norling, A. Vorobiev, H. Jacobsson, S. Gevorgian, "A low-noise K-band VCO based on room-temperature ferroelectric varactors", IEEE Trans. Microw. Theo. Tech., Vol. 55, No. 2, pp. 361-369, 2007.
- [59] N. Engheta, R.W. Ziolkowski (Ed.), "Metamaterials", IEEE Press, NJ, 2006.
- [60] H. Kubo, T. Mukai, A. Sanada, "High Q or high effective permittivity artificial dielectric resonator in a waveguide", IEICE Trans. Electron., Vol. E88-C, No. 7, 2005.
- [61] D. Huang, T. LaRocca, M.-C.F. Chang, "Low phase noise millimetre-wave

frequency generation using embedded artificial dielectric", Electron. Lett., Vol. 43, No. 18, pp. 983-984, 2007.

Chapter 2

Ferroelectric Materials for Integrated Tunable Microwave Components

2.1 Introduction

Electrically tunable microwave devices, such as varactors and phase shifters, are indispensable in modern microwave integrated circuits. Taking a multi-band, multi-mode transceiver as an example, simple adaptive blocks employing tunable components offer the advantages of compact size and low loss [1] in comparison with the traditional method of using parallel, non-adaptive circuit blocks. For next generation wireless communication systems to support high data transfer rates, the operating frequency is desired to rise to broadband channels at Ku-band (12-18 GHz), K-band (18-26 GHz) and even higher [2]. In spite of the progress in transistor technologies, no semiconductor varactors with high enough quality factor (Q-factor) and tunability are available for

frequency range higher than 10-20 GHz [3].

Due to the electric field dependence of the permittivity, ferroelectric (FE) materials have been proposed for tunable microwave devices over 40 years ago [4]. However, extensive research on FE devices is fairly recent [5]. In theory and practice, all forms of FE materials, bulks, thin films and thick films can be used in tunable microwave devices. Particularly, FE thin films have attracted large amount of interests due to their broad tuning range, low tuning voltage, high breakdown voltage and especially low loss at microwave frequencies. Already thin-film based varactors with a Q-factor of ~40 at 45 GHz have been demonstrated [6]. Besides, cost effective deposition methods for FE thin film are readily available and the strong potential for integration into existing IC processes make it a competitive candidate for high frequency applications.

Even though FE materials in ferroelectric phase can not be completely ruled out for microwave applications, it is commonly believed that only in the paraelectric phase the desired properties of high tunability and low loss can be achieved simultaneously [5]. Moreover, it has been verified that only near the Curie point does a FE material show highest tunability [5]. Therefore, it is desired that FE material has a Curie point T_c slightly below the normal operating temperature, *i.e.* at room temperature. Among the various FE materials, Barium Strontium Titianate (BST) is the most widely investigated one for tunable microwave applications [7]. BST is a solid solution of $BaTiO_3$ and $SrTiO_3$ and is commonly denoted as $Ba_xSr_{1-x}TiO_3$, where x represents the relative concentration of the two composites. By the formation of the solid solution, the Curie point can be shifted to any value between about 400K (T_c of BaTiO₃) and 105K (T_c of SrTiO₃), depending on the x value [5]. As an example, $Ba_0 rSr_0 TO_3$ (x=0.7) bulk ceramic undergoes a paraelectric-ferroelectric phase transition near room temperature. Although in contrast to its bulk form, BST thin film shows much lower permittivity due to fine grain size, interfacial capacitance and residual stress [8], reasonable tunning range and dielectric loss suitable for microwave tunable applications can still be realized.

In this chapter, different techniques for BST thin film deposition are discussed and compared. Chemical solution deposition (CSD) method is chosen for preparation of BST thin film in the lab facilities of Delft Institute of Microsystems and Nanoelectronics (DIMES). Results for structural and electrical characterization of deposited films are given.

From the device point of view, ferroelectric thin-film based tunable components are

widely investigated for a variety of applications in microwave (sub-)circuits, such as phase shifters [9,10], tunable filters [11], matching networks [12] and voltage controlled oscillators (VCO) [13]. The tunable components roughly fall into two categories: lumped elements like varacors and distributed components such as tunable transmission lines (this classification depends on the wavelength of electromagnetic waves compared to the device dimensions). In both cases, either a parallel-plate or a planar structure can be adopted for fabrication of tunable devices.

Planar configurations, like finger print varactors and coplanar waveguide (CPW) transmission lines, with both electrodes on the top of the ferroelectric film are easy to fabricate, but at a disadvantage. The electric field in a planar device is applied over lateral distances of the order of microns [14]. The dc bias voltage required for changing the dielectric constant of the ferroelectric film (and, thus, the electric characteristics of the device) should now be applied over such distances. Given the dc fields required for a significant modification of the dielectric constant of typical ferroelectric materials (tens of V/µm in the case of BST thin film), relatively high dc voltages (hundreds or even thousands of Volt) are needed to realize a reasonable tuning range.

On the other hand, tunable components using ferroelectric thin films in a parallel-plate configuration offer low tuning voltages since the voltage is now applied over the small thickness of the ferroelectric layer, inducing high electric fields (as shown in Fig.3.1). However, the on-chip integration of parallel-plate ferroelectric devices faces a major difficulty: in order to be able to grow ferroelectric films of microwave quality on top of the bottom electrode, the latter should satisfy stringent requirements. Proper crystal growth of the ferroelectric film requires small lattice mismatch as well as small thermal expansion coefficient mismatch between the ferroelectric and the underlying metallic electrode. Besides, the electrode should withstand the high temperatures required for the ferroelectric deposition and annealing. The integrated ferroelectric varactors reported so far mostly use Platinum (Pt) electrodes which satisfy these requirements. However, Pt is not compatible with a modern silicon (Si) integrated circuit (IC) process where Aluminum and Copper are the metals of choice.

In this chapter, substrate transfer technology (STT) is also proposed for the fabrication of BST thin film varactors and tunable microstrip transmission lines with parallel-plate configuration. The use of STT circumvents the restriction of Pt electrodes since the high temperature process step for BST deposition precedes the formation of the

25

two electrodes. In principle, based on STT, BST thin film can be deposited on any substrate as far as the whole or part of the substrate materials can be removed later. However, lattice mismatch and thermal expansion coefficient mismatch between BST and substrate should also be taken into consideration for choosing a substrate material. As a preliminary step, oxidized Si wafer was used as the substrate since due to the relatively mature STT process based on Si wafer. However, devices fabricated were found to be short circuited. Possible causes are investigated which indicating that BST deposition on silicon oxide is likely not a feasible solution. Comments on future works are finally given.

2.2 Chemical solution deposition of Barium Strontium Titanate (BST) thin film

A variety of deposition techniques for BST thin films, including pulsed laser deposition (PLD), RF sputtering, metalorganic chemical vapor deposition (MOCVD), molecular beam epitaxy (MBE) and chemical solution deposition (CSD), are available. A comparison of the advantages and limitations of these methods is given in Table 2.1.

CSD methods have been widely investigated for perovskite thin films since the mid-1980s, mainly aimed at FE thin films for electronic applications [17]. One major advantage of CSD methods over other deposition techniques is an easier stoichiometry control as different film compositions (Ba/Sr ratio in the case of BST deposition for example) can be readily achieved by changing the composition of starting chemicals. Besides, the compositional homogeneity of deposited film can be guaranteed by using an appropriate mixture of starting chemicals. Therefore, high quality samples with different compositions can be quickly tested in a cost effective way. In addition, the low investment capital makes it attractive not only for device fabrication but also for scientific research.

Conventionally, the morphology, or micro structure of BST thin films deposited by CSD method is a randomly oriented polycrystalline structure. However, the device performances (dielectric constant, tunability and loss) are generally considered to degrade from the grain boundaries in the film. Therefore, especially in the case of parallel plate tunable capacitors, the morphology control is normally regarded as a drawback of CSD method. Nevertheless, Hoffman and co-workers have shown that by controlling the precursor chemistry and deposition parameters, it is possible to change the microstructure of BST thin films from a randomly oriented polycrystalline configuration to a columnar structure [18]. Based on these characteristics, we establish CSD method for BST thin film deposition by using the facilities of DIMES.

Normally, CSD deposition of BST thin film includes (1) precursor preparation, (2) film deposition and (3) pyrolysis and crystallization by high temperature annealing. The precursor preparation generally involves dissolving metalorganic compounds (typically metal alkoxide compounds) into a common solvent. The solution is then deposited on a substrate by spin coating. The as-deposited film is amorphous in nature and contains a significant organic fraction. To convert the amorphous film to a crystalline one, either the so called *two-step* or *one-step* pyrolysis and crystallization processes will follow. In the *two-step* case, the as-deposited film is subject to a separate pyrolysis step for bake out of the organic species, at relatively low temperature (normally 200-400 °C), prior to a high temperature crystallization annealing. In the *one-step* method, the film is directly sent to a pre-heated furnace at crystallization temperature, which results in both organic pyrolysis and crystallization.

Based on works of Hoffmann [18] and Hasenkox [19], a process flow for Ba_{0.7}Sr_{0.3}TiO₃ is developed as sketched in Fig.2.1. The composition (Ba/Sr ratio) is chosen so that the Curie point is shifted to room temperature, where the film is believed to possess highest dielectric constant at zero bias voltage. Since the trend "the higher the dielectric constant, the higher the tunability" is observed in many dielectrics, highest value of dielectric constant is therefore favorable for microwave tunable applications [5]. The reason of choosing Ba acetate and Sr acetate as starting materials is the formation of an intermediate carbonate phase [18]. The acetylacetone (AcAc) is reacted with titanium-tetra-n-butoxide or Ti(OBu)₄ in Fig.2.1 to prevent the formation of nanosized TiO_2 particles in the solution, which will degrade the film performance [19]. The mixed starting materials are diluted to 0.1M in 2-methylpropionates (2MOE). A crystallization step is taken after every single spin coating which will result in a thickness of ~10 nm per layer. According to Hoffmann, it is the intermediate phase, suitable concentration of the final solution (resulting in thinner film per layer) and the crystallization strategy that produce a columnar structure of the final BST film, which is favorable for tunable devices with a metal-insulator-metal (MIM) structure like parallel plate varactors and microstrip transmission lines.

Table 2.1 Advantages and limitations of process techniques for BST thin film deposition[15,16]

Method	Advantages	Limitations	
	Rapid sampling of materials	Morphology	
	Quickly produce new materials	Point defect concentration	
PLD	Low temperature epitaxial growth	Limited sample scale	
		Uniformity	
		High residual stress	
	Cost	Point defect concentration	
DE	Uniformity	Residual stress	
Sputtering	Low temperature	Stoichiometry control	
Sputtering	Scalability	Slow deposition rate for oxide	
	Standard IC processing		
	Uniformity	Immature technology	
	Morphology	Precursor stability	
MOCVD	Composition control	Precursor availability	
	Low point defect concentration	Expensive	
	Scalability	Down-time	
	Uniformity	Expensive	
	Composition control	Complexity	
MBE	Precise atomic layering	Immature for oxide	
	Extreme flexibility		
	Scalability		
	Inexpensive, low capital investment	Phase control	
	Rapid sampling of materials	Morphology	
CSD	Homogeneity	Reproducibility	
	Stoichiometry control		
	Simple preparation route		

2.3 BST thin films characterization

2.3.1 Structural characterization

BST thin films are developed by CSD method following the aforementioned process flow
(Fig.2.1) on 100nmPt/20nmTi/SiO₂/Si substrate. Titanium is added for good adhesion between Pt and SiO₂. 40 nm-thick BST thin films. deposited by four coating-crystallization cycles, were investigated in a FEI XL30 scanning electron microscopy (SEM) to characterize the micro-structure during the film evolution. The SEM sample was coated by a thin gold layer to prevent charge accumulation during experiment. Atomic force microscopy (AFM) (NTEGRA, NT-MDT) operating at semi-contact mode was used to study the surface morphology and roughness of BST thin film after fifteen coating-crystallization cycles.



Figure 2.1: Process flow of BST thin film preparation by CSD method.



Figure 2.2: SEM images of ~40nm BST thin films by CSD method. The film is dense and without crack and fine grain size (30-50nm) is clearly demonstrated.

Fig.2.2 shows the SEM images of the 40 nm-thick BST films at different magnifications. The films are dense and crack-free with grain sizes in the 30-50 nm range.

In Fig.2.3, the AFM image of BST thin film after fifteen coating-crystallization cycles is shown. The thickness of the film is measured by Tencor alpha step surface profiler after patterning the film in 2.5%HF solution and is found to be 150 ± 5 nm. Therefore, approximately a 10 nm layer is obtained for each coating-crystallization cycle. The films are again homogeneous and crack-free and the surface roughness is found from the AFM measurement to have a root mean square value of 0.96 nm.



Figure 2.3: Surface morphology of 150 nm BST thin film by AFM. Smooth surface is identified.

2.3.2 Electrical characterization

To characterize the tunable properties of BST thin films, *i.e.* the dielectric constant as a function of applied bias voltage, parallel plate capacitors were fabricated for the 150 nm BST thin films on a platinized wafer. A Pt layer of 100 nm in thickness was used as the bottom electrode with 20 nm Ti film as adhesion layer between Pt and Si substrate. After the BST deposition, a layer of 1.4 μ m Aluminum (Al) was sputtered on top of it. The Al layer was then patterned and etched to form the top electrode of the parallel-plate capacitor. (Fig.2.4).



Figure 2.4: Cross section view of of Al/BST/Pt thin film varactor.



Figure 2.5: Equivalent circuit model of BST thin film capacitor.

Impedance measurements as a function of bias voltage at room temperature (25 °C) and -30 °C were performed with a HP4156A precision semiconductor parameter analyzer on a Cascade probing station equipped with a thermo-chuck at 1 MHz. An equivalent circuit model of BST parallel capacitor is developed and shown in Fig.2.5, in which R_{el} represents the resistances of the two electrodes as well as the contact resistance, R_{BST} models the losses of BST thin film capacitor due to dielectric loss and leaky current, and C_{BST} is the capacitance of the film. The values of R_{BST} and C_{BST} are calculated from the impedance measurement based on the equivalent circuit model by assuming R_{el} equals to 1.5 Ω .

Fig. 2.6 shows the capacitance density and the capacitor loss which is represented by the inverse of the device quality factor (1/Q). The capacitor is built using a 150 nm-thick BST film sandwiched between a bottom Pt electrode and a top Al electrode. The applied voltage is varied from -5 V to +5 V (bottom with respect to top electrode). The



Figure 2.6: Capacitance density and capacitor losses of Al/BST/Pt thin film varactor as a function of applied bias voltage on Pt electrode.

measurements were carried out under two different temperatures: room temperature (25 °C) and -30 °C. From the figure, we can clearly identify the asymmetry of both capacitance and loss with respect to the bias voltage. Furthermore, the following observations can be made:

- At most negative bias voltages, the temperature does not affect the capacitance and quality factor of the capacitor. The effect of temperature at positive bias voltages, however, is significant.
- (2) The maximum capacitance value at room temperature is obtained at bias voltage of 1 V instead of 0 V.
- (3) The loss at negative bias voltage is relatively flat. It slightly increases with increasing the voltage.
- (4) At positive bias voltage, the losses increase much faster with the applied (positive) voltage. This is especially significant at room temperature.



Figure 2.7: Band diagrams of Al/BST/Pt capacitor, (a) the work functions of Al and Pt, the electron affinity and band gap of BST; (b) band diagram of Al/BST/Pt when contacts are formed and equilibrium is established; (c) band diagram of Al/BST/Pt when negative bias voltage is applied at Pt; (d) band diagram of Al/BST/Pt when positive bias voltage is applied at Pt.

Note, that by increasing and then decreasing the bias voltage, no hysterisis was observed. Therefore, the film is in its paraelectric phase and the polarity behavior is not due to the possible formation of ferroelectric phase of the film. Thus, the observed asymmetry may be attributed to the different electrodes used (Pt vs. Al), and can be understood by the band structure of Al/BST/Pt layers which is shown in Fig.2.7. As shown in Fig.2.7 (a), Al and Pt have work functions of 4.3 eV [20] and 5.3 eV [21] respectively, while BST has an electron affinity of 4.0 eV and a band gap of 3.6 eV [21]. The Fermi level of BST is normally believed to be lightly above the center of the band gap [21]. When the three materials form the capacitor structure with zero bias voltage and thermal equilibrium is achieved, the Fermi levels of the three regions must be equalized.

As shown in Fig.2.7 (b), due to the small differences of the work function of Al and the electron affinity of BST, an ohmic contact is formed at the Al/BST interface. While in case of Pt/BST, a Schottky barrier is formed. Therefore space charge regions are developed in BST at both interfaces and an interfacial voltage (V_{int}) is built under zero

bias. This voltage can be expressed as

$$V_{\rm int} = V_{\rm bi(AI/BST)} + V_{\rm bi(BST/Pt)} = \frac{1}{e} (\phi_{\rm Pt} - \phi_{\rm Al}), \qquad (2.1)$$

where $V_{\rm bi(AI/BST)}$ and $V_{\rm bi(BST/Pt)}$ are the built-in voltage at the Al/BST and Pt/BST interfaces, respectively, and $\phi_{\rm Pt}$ and $\phi_{\rm Al}$ are work functions of Pt, respectively, Al [22].

Taking the values of $\phi_{Pt} = 5.3 \text{eV}$ and $\phi_{Al} = 4.3 \text{eV}$, the interfacial voltage is calculated to be 1.0 V, which is consistent with the measurement result, where the maximum value of the capacitance occurs when 1 V is applied on Pt electrode to annihilate the interfacial voltage V_{int} . When negative bias voltage is applied on Pt electrode, the energy band is bended as shown in Fig.2.7 (c), the leakage current is controlled by the Schottky barrier at Pt/BST interface. Therefore, the losses at low negative bias voltage are expected to be dominated by the dielectric loss due to the 1.4 eV potential barrier [20]. However, if the Pt electrode is positively biased, as shown in Fig.2.7 (d), much leakage current will flow due to the ohmic contact at the Al/BST interface. As the bias voltage increases, the potential barrier is reduced by the Schottky effect [23]. The thermally excited electrons will now easily overcome the potential barrier and flow across the BST film into the Pt electrode, resulting in a rapid growth of the capacitor loss at room temperature. At -30 °C, the concentration of thermally excited electrons is much less than that at room temperature, therefore the losses due to leakage current are lower as clearly exhibited in Fig.2.6.

The relative dielectric constant and the loss tangent of the BST thin film is extracted from the calculated R_{BST} and C_{BST} in Fig.2.5 for negative bias supplies, where the capacitor loss is mainly due to the dielectric loss of the BST thin film. Since the thickness of the film (150 nm) is much less than the device dimensions, the fringe field is safely ignored in the extraction of dielectric constant from capacitance value. The dielectric constant of 380 and loss tangent of 0.05 are obtained at zero bias, and a tunability of 2.05 is achieved with a bias voltage as low as 5 V for 150 nm BST thin film. However, compared to the results of Hoffmann [18], the dielectric constant is still lower and the dielectric loss higher than their reported values, which makes our prepared BST thin film not of sufficient quality for tunable microwave applications. One possible reason is the use of atmospheric ambient in the crystallization step where an oxygen ambient environment is preferred to reduce oxygen vacancies in the film. The quality of the solution precursor is another important factor determining the film properties. For example, the more particles in the precursor, the more defects in the deposited film. Extensive efforts are needed to determine the optimized conditions to prepare and purify the precursors. Furthermore, the possible contamination of the annealing environment, the wafer cleaning before each coating steps, the annealing strategies, etc. should also be taken into consideration to identify the optimal conditions to prepare BST thin films for making good quality devices.

2.4 Tunable components by substrate transfer technology (STT)

2.4.1 Introduction

As mentioned in Section 2.1, parallel-plate BST tunable devices normally require Pt as bottom and/or top electrodes. However, Pt is not IC compatible and is difficult to etch. Therefore, we propose an alternative method to fabricate parallel-plate components which utilize metals common in IC technology. The method is based on the substrate transfer technique (STT).

STT was first introduced by Dekker *et al.* in 1990's [24] for radio frequency (RF) circuit applications. The motivations behind STT are lower power consumption of the RF circuits, higher quality of passive components, and suppression of cross-talk between adjacent devices at RF frequencies [25]. STT achieves this by substituting the major part of the silicon substrate by a glass substrate. Already, low-power, highly integrated GPS receiver frontends [25] and high-performance varactor diodes [26] have been demonstrated using the STT. The bio-compatibility of glass has triggered new research on STT in integrated bio-electronic circuits [27], in addition to its (initially intended) RF application.

Essentially, STT consists of gluing the silicon wafer, face down, onto an alternative

substrate like glass. The process can be summarized as follows: First, active devices are fabricated on silicon wafer using ordinary silicon IC technologies. After that, both surfaces of silicon and glass substrate are treated with special primer to promote the adhesion between the substrate and glue. Then both substrates are glued together. The wafer sandwich is then cured by ultraviolet (UV) light. The silicon wafer is now partially or completely removed depending on the type of silicon substrate, bulk or silicon on isolator (SOI). After cleaning, the wafer is now ready for further processing, such as the formation of passive components and interconnects.

2.4.2 STT process for ferroelectric tunable components

The use of STT allows one to carry out the inevitable high temperature annealing in the deposition of BST thin films prior to the metallization of the film on both sides. As a result, IC compatible metals such as Al and Cu can be used to form electrodes or signal/ground layers of tunable devices. The process flowchart of fabrication of BST tunable microwave devices is designed (see Appendix B) and summarized in Fig.2.8.

The process starts with BST thin film deposition by CSD method on an oxidized Si substrate (step #1). Notice that the high temperature (750 °C) annealing is contained within this step. Step #2 involves patterning of BST thin film by 2.5% HF solution as well as the deposition and structuring of the first metal layer (Al is shown in the figure). This metal layer will serve, e.g., as the bottom electrode of the parallel-plate varactors and ground plane of microstrip transmission lines. The gluing of Si on the glass (Schott AF45 in this case) substrate followed by UV light curing is done in step #3. The Si substrate is then completely removed by wet etching in 33% Tetramethylammonium hydroxide (TMAOH) solution at 85 °C (step #4). Due to the high etching selectivity of SiO₂ over Si in the TMAOH solution, the etching will stop at the Si oxide layer. Finally, the Si oxide layer is patterned for contacting the BST film with the second metal to complete the device fabrication.



Figure 2.8 Sketch of STT process flow of ferroelectric tunable devices.



Figure 2.9: Micrographs of as-fabricated (a) parallel plate varactor and (b) microstrip transmission line.

2.5 Experimental results and discussions

Following the STT process flow described in the previous section, BST thin film tunable parallel-plate capacitors and microstrip transmission lines were fabricated at the DIMES cleanroom facilities. The thickness of BST thin film is 200 nm which is deposited by 20-fold repetition of the coating-crystallization cycles, as described section 2.2. Micrographs of fabricated tunable devices are shown in Fig.2.9. The ground plane of the microstrip is patterned to a defected ground structure to reduce the induced current flow in the ground layer, therefore to increase the inductance per unit length value (L')[28-30]. On the other hand, the capacitance per unit length value (C') is reduced due to the patterning of the ground since the overlap area of signal line and ground is apparently reduced by the patterning. Since the characteristic impedance of a transmission line (Z_0) is given by

$$Z_0 = \sqrt{\frac{L'}{C'}}, \qquad (2.2)$$

 Z_0 can be significantly increased by patterning of the ground, which overcomes one major drawback of microstrip configuration of transmission line with thin film between two metal layers, *i.e.* the very low characteristic impedance. Besides, by ignoring the dielectric loss (which is valid in most cases), the quality factor of the transmission line is

$$Q = \frac{\omega L'}{R}, \qquad (2.3)$$

where ω is the angular frequency. It can be seen that, by patterning of the ground layer, the quality factor of the transmission line is also increased.

However, the electrical measurement of these devices showed that they were short circuited. Hence, we then analyzed the surface of the BST film (those potions which are open to the air after device fabrication) by AFM. Micro-cracks in the micrometer range, uniformly distributed over the whole BST film, were identified (Fig.2.10). Therefore, the two metal layers are in contact with each other through these micro cracks resulting in direct connection between two electrodes of the varactor or between the signal and ground lines of the microstrip line.

To trace the origin of the aforementioned micro cracks of the BST thin film, AFM was used to monitor the surface morphology of the deposited film after every three coating-crystallization cycles (Fig.2.11). Fig.2.11 (a-c) shows the AFM images of BST thin film after 3, 6 and 9 coating-crystallization cycles. Notice that in Fig.2.11c, bright line segments were observed. These segments form a network throughout the sample as revealed by a larger area scan (Fig.2.11d). The dimensions of the line segments were in the micron range, which were at the same level as the micro crack in Fig.2.10. The sample was then etched in 0.55% HF solution for one minute to partly remove a certain thickness of BST at the surface. AFM image (Fig.2.11e) then showed clearly micro cracks coincided with the position of the bright line segments.

From the AFM images of Fig.2.11, it can be seen that the micro cracks are formed at the SiO₂/BST interface. After each annealing step, the temperature of the wafer will drop from 750 °C to room temperature before the next coating. Due to the large thermal expansion coefficient mismatch between SiO₂ ($5x10^{-7}$ K⁻¹ [31]) and BST thin film ($7.8x10^{-6}$ K⁻¹ [32]), the former and the latter will experience compressive and tensile stress respectively. If the thickness of BST thin film is higher than a critical value, cracks will originate at the interface between SiO₂ and BST to release the stresses. Although at the beginning, as in Fig.2.11a and Fig.2.11b, the cracks are not observable at surface, further experiments have shown that the micro cracks already exists in 40 nm thick BST thin film (with 4 coating-crystallization cycles). As a new BST layer is deposited on top of the existing ones, continuous film can be formed on the top, which is verified by Fig2.11c and Fig.2.11d, where only bright line segments instead of cracks are observed. However, the micro cracks underneath will grow and propagate. Therefore, after removing the top surface layer of BST by wet etching, micro cracks are exposed as can



Figure 2.10: AFM image of BST thin film surface after STT processing shows micro cracks.

be seen in Fig2.11e.

As the process flow of Fig.2.8 shows, the SiO_2 layer is patterned after substrate transfer and Si removal in order to contact the BST film with the second metallic electrode. Micro-cracks at this moment are exposed to the air and have developed through the whole thickness of BST thin film during former processing steps (for example, the dry etching of SiO₂). Therefore, the second metal layer will directly contact with the first metal layer, leading to short circuited devices.

The above discussions assert that to realize BST-based tunable devices by STT processing, it is critical to obtain crack-free BST films of practical thickness. From this point of view, the use of SiO_2 as the buffer layer for BST deposition is not appropriate due to the large thermal expansion coefficient differences between the layers. To overcome this problem, a seed layer with small lattice and thermal expansion coefficient mismatch to BST is required to prevent the formation of cracks during the CSD processing steps. On the other hand, this seed layer must allow for selective etching with respect to the BST material in order to implement the second metal contact. Aluminium



Figure 2.11: AFM images of surface morphologies of BST thin film after (a) 3 times, (b) 6 times, (c) 9 times coating-crystallization cycles; (d) large area scan of 9 layers shows network of bright line segments; (e) the line segments were identified to be micro-cracks after wet etching in 0.55HF solution.

oxide (Al_2O_3) and titanium oxide (TiO_2) are possible candidates, while further experiments are necessary for verification.

It is worth mentioning that the essential idea of STT could also be applied to substrates other than Si. Normally, high quality ferroelectric thin films, such as BST, are grown epitaxially on substrates like MgO and LaAlO₃ (LAO) [33]. By virtue of STT, the substrate (MgO or LAO) can be removed if a suitable etchant is found. After processing the top surface of the epitaxial ferroelectric material (e.g. deposition and patterning of the top metal layer) the whole wafer can again be glued onto glass, the substrate can be removed, and the second metallic electrode can be implemented. In this way, high quality ferroelectric thin films can be included in IC-compatible glass (or even Si) processes.

2.6 Conclusions

In summary, crack-free and smooth surface BST thin films have been successfully prepared by CSD method on platinum coated silicon substrates. Tunable parallel plate capacitors with Al/BST/Pt structure were fabricated and characterized by impedance measurement at 1 MHz. The measured asymmetric characteristics of both capacitance and loss with respect to the bias voltage applied between the Pt and Al electrodes is attributed to the formation of an ohmic contact and a Schottky barrier at the Al/BST and Pt/BST interfaces, respectively. A dielectric constant of 380 at zero bias voltage, and a tuning range of 2.05 at only 5 V were observed for the BST thin film.

Besides, ferroelectric thin film tunable microwave devices in parallel-plate configuration are proposed to be realized by substrate transfer technology to lift the stringent requirement on the metallic electrodes. Varactors and microstrip transmission lines with BST thin film sandwiched between two Al layers were fabricated using STT, starting from oxidized Si wafer. Unfortunately, the devices were found to be short circuited. This was casued by micro cracks originating from the SiO₂/BST interface due to the large thermal expansion mismatch.

References

- [1] W.C.E. Neo, Y. Lin, X.D. Liu, L.C.N. de Vreed, L.E. Larson, M. Spirito, M.J. Pelk, K. Buisman, A. Akhnoukh, A. de Graauw, L.K. Nanver, "Adaptive multi-band multi-mode power amplifier using integrated varactor-based tunable matching network", IEEE J. Solid-State Circuits, Vol. 41, No. 9, pp. 2166-2176, 2006.
- [2] C.H. Mueller, F.W. Van Keuls, R.R. Romanofsky, F.A. Miranda, J.D. Warner, "Characterization of (Ba_{0.5}Sr_{0.5})TiO₃ thin films for Ku-band phase shifters", Integr. Ferroelectr., Vol. 28, No. 1-4, 2000.
- [3] N. Setter, D. Damjanovic, L. Eng, G. Fox, S. Gevorgian, S. Hong, A. Kingon, H. Kohlstedt, N.Y. Park, G.B. Stephenson, I. Stolitchnov, A.K. Taganstev, D.V. Taylor, T. Yamada, S. Streiffer, "Ferroelectric thin films: review of materials, properties, and applications", J. Appl. Phys., Vol.100, pp. 051606/1-46, 2006.
- [4] M.D. Domenico, D.A. Johnson, R.H. Pantell, "Ferroelectric harmonic generator and

43

the large-signal microwave characteristics of a ferroelectric ceramic", J. Appl. Phys., Vol. 33, No. 5, pp. 1697-1705, 1962.

- [5] A.K. Tagantsev, V.O. Sherman, K.F. Astafiev, J. Venkatesh, N. Setter, "Ferroelectric materials for microwave tunable applications", J. Electroceram., Vol. 11, No. 1-2, pp. 5-66, 2003.
- [6] A. Vorobiev, P. Rundqvist, K. Khamchane, S. Gevorgian, "Silicon substrate integrated high Q-factor parallel-plate ferroelectric varactors for microwave /millimeterwave applications", Appl. Phys. Lett., Vol. 83, No. 15, pp. 3144-3146, 2003.
- [7] B. Acikel, "High performance barium strontium titanate varactor technology for low cost circuit applications", Ph.D dissertation, University of California Santa Barbara, 2002.
- [8] T.M. Shaw, Z. Suo, M. Huang, E. Liniger, R.B. Laibowitz, J.D. Baniecki, "The effect of stress on the dielectric properties of barium strontium titanate thin films", Appl. Phys. Lett., Vol. 75, No. 14, pp. 2129-2131, 1999.
- [9] A. Kozyrev, V. Osadchy, A. Pavlov, L. Sengupta, "Application of ferroelectrics in phase shifter design", IEEE MTT-S Digest, Vol. 3, pp. 1355-1358, 2000.
- [10] B. Acikel, T.R. Taylor, P.J. Hansen, J.S. Speck, R.A. York, "A new high performance phase shifter using BaxSr1-xTiO3 thin films", IEEE Microw. Wireless. Compon. Lett., Vol. 12, No. 7, pp. 237-239, 2002.
- [11] V.N. Keis, A.B. Kozyrev, M.L. Khazov, J. Sok, J.S. Lee, "20GHz tunable filter based on ferroelectric (Ba,Sr)TiO3 film varactors", Electron. Lett., Vol. 34, No. 11, pp. 1107-1109, 1998.
- [12] L.-Y.V. Chen, R. Forse, D. Chase, R.A. York, "Analog tunable matching network using integrated thin-film BST capacitors", IEEE MTT-S Digest, Vol. 1, pp. 261-264, 2004.
- [13] M. Norling, A. Vorobiev, H. Jacobsson, S. Gevorgian, "A low-noise K-band VCO based on room-temperature ferroelectric varactors", IEEE Trans. Microw. Theo. Tech., Vol. 55, No. 2, pp. 361-369, 2007.
- [14] A.K. Tagantsev, V.O. Sherman, K.F. Astafiev, J. Venkatesh, N. Setter, "Ferroelectric materials for microwave tunable applications", J. Electroceram., Vol. 11, No. 1-2, pp. 5-66, 2003.
- [15] S. Ezhilvalavan, T.-Y. Tseng, "Progress in the developments of (Ba,Sr)TiO3 (BST)

thin films for Gigabit era DRAMs", Mater. Chem. Phys., Vol. 65, No. 3, pp. 227-248, 2000.

- [16] B. York, A. Nagra, J. Speck, O. Auciello, S. Streiffer, "Thin-film ferroelectrics: deposition methods and applications", [Online]. Available: http://my.ece.ucsb.edu /yorklab/Projects/Ferroelectrics/IMS2000%20Workshop/Bob-oral.pdf.
- [17] R.W. Schwartz, "Chemical solution deposition of perovskite thin films", Chem. Mater., Vol. 9, No. 11, pp. 2325-2340, 1997.
- [18] S. Hoffmann, R. Waser, "Control of the morphology of CSD-prepared (Ba,Sr)TiO3 thin films", J. Euro. Ceram. Society, Vol. 19, No. 6, pp. 1339-1343, 1999.
- [19] U. Hasenkox, S. Hoffmann, R. Waser, "Influence of precursor chemistry on the formation of MTiO3 (M=Ba, Sr) ceramic thin films", J. Sol-Gel Sci. Tech., Vol. 12, No. 2, pp. 67-79, 1998.
- [20] H.B. Michaelson, "The work function of the elements and its periodicity", J. Appl. Phys., Vol. 48, No. 11, pp. 4729-4733, 1977.
- [21] J.F. Scott, K. Watanabe, A.J. Hartmann, R.N. Lamb, "Device models for PZT/Pt, BST/Pt, SBT/Pt, and SBT/Bi ferroelectric memories", Ferroelectrics, Vol. 225, No. 1, pp. 83-90, 1999.
- [22] L. Goux, M. Gervais, F. Gervais, A. Gatherinot, C. Champeaux, F. Sabary, "Characterization of pulsed laser deposited Ba_{0.6}Sr_{0.4}TiO₃ on Pt-coated silicon substrates", Mater. Sci. in Semicon. Processing, Vol. 5, No. 2, pp. 189-194, 2003.
- [23] D.A. Neamen, "Semiconductor physics and devices, 3rd ed", McGraw-Hill, New York, 2003.
- [24] R. Dekker, P. Baltus, M. van Deurzen, W. v.d.Einden, H. Mass, A. Wagemans, "An ultra low-power RF bipolar technology on glass", Techn. Dig. IEDM, pp. 921-923, 1996.
- [25] R. Dekker, P. Baltus, H. Mass, "Substrate transfer for RF technologies", IEEE Trans. Electron Dev., Vol. 50, No. 3, pp. 747-757, 2003.
- [26] K. Buisman, L.K. Nanver, T.L.M. Scholtes, H. Schellevis, L.C.N. de Vreede, "High-performance varactor diodes integrated in a silicon-on-glass technology", Proc. ESSDERC, pp. 117-120, 2005.
- [27] H.S. Kim, R.H. Blick, D.M. Kim, C.B. Eom, "Bonding silicon-on-insulator to glass wafers for integrated bio-electronic circuits", App. Phys. Lett., Vol. 85, No. 12, pp.

2370-2372, 2004.

- [28] C.S. Kim, J.S. Park, D. Ahn, J.B. Lim, "A novel 1-D periodic defected ground structure for planar circuits", IEEE Microwave Guided Wave Lett., Vol. 10, No. 4, pp.131-133, 2000.
- [29] Y. Qian, V. Radisic, T. Itoh, "Simulation and experiment of photonic bandgap structures for microstrip circuits", in APMC'97 Proc., Hong Kong, Dec., 1997, pp. 585-588.
- [30] V. Radisic, Y. Qian, R. Coccioli, T. Itoh, "Novel 2-D photonic bandgap structure for microstrip lines", IEEE Microwave Guided Wave Lett., Vol. 8, No. 2, pp.69-71, 1998.
- [31] S.M. Sze, "Physics of Semiconductor Devices", Wiley, New York, 1981.
- [32] T.R. Taylor, P.J. Hansen, B. Acikel, N. Pervez, R.A. York, S.K. Streiffer, J.S. Speck, "Impact of thermal strain on the dielectric constant of sputtered barium strontium titanate thin films", App. Phys. Lett., Vol. 80, No. 11, pp. 1978-1980, 2002.
- [33] C.M. Carlson, T.V. Rivkin, P.A. Parilla, J.D. Perkins, D.S. Ginley, A.B. Kozyrev, V.N. Oshadchy, A.S. Pavlov, "Large dielectric constant (ε/ ε₀>6000) Ba_{0.4}Sr_{0.6}TiO₃ thin films for high-performance microwave phase shifters", App. Phys. Lett., Vol. 76, No. 14, pp. 1920-1922, 2000.

Chapter 3

Tunability Enhancement of Ferroelectric Varactors Using Nano-dot Electrodes

47

3.1 Introduction

In Chapter 2, we focused on the fabrication of parallel-plate varactors based on ferroelectric thin films. There, we proposed the substrate transfer technique to lift off the stringent requirement of Pt as the bottom electrode. Given the technological complexity of this process, a more conventional alternative would be to realize planar varactors with both electrodes on the top of the ferroelectric layer. The ferroelectric could then be grown on any suitable substrate and no restrictions would be imposed on the layer thickness.

The problem with this apparently straightforward solution is that, in a conventional IC process, the lateral distance between coplanar metallic electrodes cannot be made very small due to technological restrictions. These include the lithography resolution as well as

the high aspect ratio involved in etching steps. By contrast, in a parallel-plate varactor this distance is essentially the thickness of the ferroelectric film which may be as small as several nano-meters. Besides, the electric field distribution in a coplanar capacitor is essentially different from that of a parallel-plate device. As a result, compared to parallel-plate varactors, a coplanar varactor requires a high dc voltage to realize reasonable tuning range. A similar issue hinders the integration of tunable microwave components utilizing thick ferroelectric films [1-3]. While tunable resonators and phased-array antennas have already been demonstrated using thick ferroelectric single crystals and ceramics [3,5], these components require very high voltages, of the order of hundreds of volts to tens of kilovolts, for effective tuning.

To address this issue, in this chapter, we propose and theoretically investigate planar ferroelectric varactors and thick-film devices based on nano-dot metallic contacts. The key idea here is that, for any given voltage, the intensity of the electric field surrounding a metallic electrode is dictated by its dimensions if they are small compared to the ferroelectric film thickness or inter-electrode separation. For sub-micron sized electrodes this electric field can become very large and, since the dielectric constant of the ferroelectric is a function of the field intensity, high degrees of tunability can be achieved. This is in sharp contrast with typical parallel-plate configurations where the electric field and, thus, tunability, is governed by the film thickness. The method proposed can potentially be applied to all ferroelectric devices is mostly focused on the optimization of ferroelectric material aspects, while mush less attention has been paid to the role of the device structure [6].

Using realistic material parameters, we will calculate the tuning range of planar nano-dot varactors as function of dot size and film thickness. We show that devices built from electrodes with diameters below 100 nm can exhibit a higher than two-fold change of capacitance under application of relatively low DC bias voltages (<10 V) even when thick ferroelectric films (>5 μ m) are used. Since the capacitance of an individual nano-scale device is usually too small for realistic applications, we extend our analysis to arrays of nano-dots, and investigate the effects of dot size and separation on the overall tunability of the device. We show that below a critical inter-dot separation the enhanced tuning is washed out as the electric field profile starts to resemble the parallel-plate case. Furthermore, in designing the arrays, care has to be taken as to how the individual dots

are contacted with each other. The metallic pad connecting the dots should not be placed too close to the ferroelectric film, but should be separated from the latter by a conventional dielectric film with a thickness exceeding the diameter of each dot.

3.2 Basic principles of the ferroelectric nano-dot varactor

In characterizing the behavior of ferroelectric materials, tunability (also called tuning factor) is a commonly used parameter, which is defined as the ratio of the permittivity of the ferroelectric material at zero electric field to its permittivity at a finite field value,

$$n = \frac{\varepsilon(0)}{\varepsilon(E_0)}.$$
(3.1)

For a linear medium this ratio would be simply equal to unity. Thus, the tunability of a ferroelectric material originates from the non-linear relationship between the applied electric field and the polarization density. However, from a more practical viewpoint, a different definition of tunability for a varactor can be given in terms of the ratio of the capacitance at zero voltage to the capacitance at some nonzero voltage:

$$n_c = \frac{C(0)}{C(V_0)}.$$
 (3.2)

For a parallel plate capacitor, if the fringe field effects are ignored, n and n_c have the same value since the electric field is constant and simply proportional to the voltage. In other configurations, such as the coplanar and interdigited capacitors, the field distribution is not necessarily linearly related to the applied voltage. In such applications, only the tunability of the capacitor (n_c) is of practical importance. However, one should bear in mind that it is the electric field which dictates the (local) value of the dielectric constant in the medium and, in the end, determines the tuning factor. It is the latter consideration which lies at the basis of the idea on using nano-dot electrodes: one may enhance the device tunability by dramatically increasing the electric field locally.

One possibility is to shrink the electrode(s) to sub-micron or even nanometer range (nano-dot contact). Fig.3.1 schematically shows a planar, nano-dot contact (NDC) ferroelectric varactor built from a metallic nano-dot (of radius R_d) and a much larger ground electrode placed above a ferroelectric/Si substrate. A conventional parallel-plate device has also been shown for comparison. Unlike the parallel-plate varactor, the distribution of the electric field in the NDC varactor is highly non-uniform. The electric

field is expected to be very high near the metallic dot, but to drop rapidly into the substrate, in analogy with the problem of a metal-semiconductor nano-junction [7,8]. Furthermore, the intensity of the field near the dot is expected to be mainly determined by the dot size rather than the thickness of the ferroelectric film, in sharp contrast with the parallel-plate configuration. The high intensity of the electric field near the dot contact lies at the basis of the high tunability of the proposed NDC varactor. Applying a relatively small dc bias voltage will cause a major change in the field-dependent dielectric constant of the ferroelectric field in this region, and the high dielectric constant of the ferroelectric, a significant change of the device capacitance can thus be achieved.



Figure 3.1 Schematic view of electric field distribution of a ferroelectric material based (a) parallel plate and (b) planar NDC varactor.

Because of the nonlinear relationship between the electric field and polarization inside a ferroelectric material, the use of numerical techniques is indispensable for an exact analysis of the NDC varactor. However, by making a number of approximations, a semi-analytic model can be derived which offers a qualitative picture of the electric behavior of the NDC varactor. To derive the model we assume that the lateral separation between the dot contact and the ground electrode is much larger than the dot size. The varactor can then be viewed as the series connection of the two capacitances C_d and C_g , representing the self-capacitance of the dot and the ground contact, respectively (Fig.3.2a). Since the area of the ground electrode is taken to be much larger than that of the nano-dot, one has $C_g \gg C_d$ so that the overall varactor capacitance approximately equals C_d itself.

Next, we note that C_d is predominantly determined by the ferroelectric and Si layers because of their high relative dielectric constants. (The thickness of these layers is denoted by t_F and t_{Si} , respectively.) Neglecting the contribution of air above the substrate, one can now make the approximation $C_d \sim C_s / 2$ where C_s is the self capacitance of the same dot when sandwiched between the original substrate and a reversed, but identical substrate as shown in Fig.3.2b. Finally, C_s is approximated by mapping the system unto the spherically symmetric structure of Fig.3.2c, where the metallic dot is replaced by a metallic sphere with the same total area, i.e. with the radius $R_a = R_d / \sqrt{2}$, embedded in a sphere of ferroelectric material with the radius $R_b = R_a + t_F$ (Fig.3.2c), followed by a shell of Si material with the radius $R_c = R_b + t_{Si}$.

At not too high frequencies (<35 GHz, as long as the ferroelectric material does not show much dispersion [9]), electrostatic field analysis can be employed to compute the self capacitance of the structure of Fig.3.2c. Suppose that a total charge of Q resides on the metallic sphere. Employing the rotational symmetry of the structure, the electric displacement can be written as $D = D_r \hat{r}$, where \hat{r} is the unit vector in the radial direction, and

$$D_r(r) = \frac{Q}{4\pi r^2},\tag{3.3}$$

52

with r is the radial distance from the center of the spherical system. Inside the ferroelectric region

$$\boldsymbol{D} = \boldsymbol{\varepsilon}_0 \boldsymbol{E} + \boldsymbol{P} \,, \tag{3.4}$$

where ε_0 is the vacuum dielectric constant, E is the electric field, and P is the polarization density which is a nonlinear function of E, most simply described by a power series expansion of the free energy in terms of P (the order parameter) [10]. For tunable microwave applications, ferroelectric materials of the displacive type in their paraelectric phase are commonly considered since it is believed that only for this case, both high tunability and a relatively low loss can be achieved.6 Following the Landau-Ginzburg-Devonshire (LGD) theory [11], if a homogeneous and isotropic ferroelectric material is assumed for simplicity, the non-linear P-E relationship can be expressed by the equation

$$\boldsymbol{E} = \boldsymbol{\alpha} \boldsymbol{P} + \boldsymbol{\beta} P^2 \boldsymbol{P} \,, \tag{3.5}$$

where $\alpha = (\varepsilon_0 \chi_f)^{-1}$ with χ_f the zero-field susceptibility of the material, β is a (positive) dielectric stiffness constant, and $P^2 = \mathbf{P} \cdot \mathbf{P}$.

Because of the isotropic nature of Eq. (3.5), and the spherical symmetry of the structure, the polarization density only has a radial component, i.e., $P = P_r \hat{r}$, which is given by the implicit equation

$$\left(\varepsilon_f / \chi_f\right) P_r + \varepsilon_0 \beta P_r^3 = \frac{Q}{4\pi r^2}, \qquad (3.6)$$

as can be seen from Eqs. (3.3)-(3.5). Here $\varepsilon_f = 1 + \chi_f$ is the zero-field relative dielectric constant of the ferroelectric material. Using the same equations, the electrostatic potential of the metallic sphere (with respect to the infinity) can now be written as

$$V = \int_{R_a}^{\infty} E_r(r) dr = \frac{Q}{C_0} + \frac{\chi_f \beta}{\varepsilon_f} \int_{R_a}^{R_b} P_r^3(r) dr$$
(3.7)

$$\frac{1}{C_0} = \frac{1}{4\pi\varepsilon_0} \left[\frac{1}{\varepsilon_f} \left(\frac{1}{R_a} - \frac{1}{R_b} \right) + \frac{1}{\varepsilon_{Si}} \left(\frac{1}{R_b} - \frac{1}{R_c} \right) + \frac{1}{R_c} \right],$$
(3.8)

where E_r is the radial component of the electric field. To arrive at this result we have used the fact that inside the Si and air layers $D_r = \varepsilon_0 \varepsilon_{Si} E_r$ and $D_r = \varepsilon_0 E_r$, respectively, with ε_{Si} the relative dielectric constant of Si. Note that C_0 is the zero-field self-capacitance of the structure, i.e., when the nonlinear term in Eq. (3.5) becomes negligibly small.

To further develop the right hand side of Eq. (3.7), we change the integration variable from r to P_r , apply partial integration, and use Eq. (3.3) to arrive at

$$V(Q) = \frac{Q}{C_0} + \frac{\chi_f \beta}{\varepsilon_f} \Big[P_b^3(Q) R_b - P_a^3(Q) R_a \Big] + 2I(Q) \sqrt{Q}$$
(3.9)

$$I(Q) = \frac{3}{4\varepsilon_0} \sqrt{\frac{\varkappa_f P_0}{\varepsilon_f \pi}} \int_{P_b(Q)/P_0}^{P_a(Q)/P_0} \frac{u^{3/2} du}{\sqrt{1+u^2}}.$$
(3.10)

Here $P_0 = \sqrt{\varepsilon_f / (\varepsilon_0 \beta \chi_f)}$, and $P_a(Q) = P_r(R_a, Q)$, $P_b(Q) = P_r(R_b, Q)$ are found by solving (3) for the charge Q. The self-capacitance of the sphere for a given total charge can now be computed as

$$\frac{1}{C_s(Q)} = \frac{dV}{dQ} = \frac{1}{C_0} + \frac{I(Q)}{\sqrt{Q}}.$$
(3.11)

Note that from Eq. (3.7) Q can be found as function of the voltage V which, subsequently, can be used to determine C_s from Eq. (3.11).

Returning to the NDC varactor with the capacitance $C \sim C_d$, and keeping in mind that $C_d \sim C_s / 2$, the overall capacitance as function of the applied bias voltage V can be expressed as



Figure 3.2: (a) Schematic view and simplified equivalent lumped element model of a planar NDC varactor; (b) Self capacitance of the NDC varactor is simplified to a system with a dot sandwiched between two identical ferroelectric/silicon substrate; (c) The system in (b) is further mapped to a spherical model for semi-analytical analysis.

$$\frac{1}{C(V)} = 2 \left\{ \frac{1}{C_0} + \frac{I[Q(V)]}{\sqrt{Q(V)}} \right\},$$
(3.12)

where we have neglected the voltage drop over the ground electrode. In the limit $V \rightarrow 0$ where the nonlinearity of the *E-P* relation become negligible, one has $C \rightarrow C_0 / 2$. Increasing V leads to an increase in the 2nd term on the right hand side of Eq. (3.12), resulting in a decease in C.

The normalized capacitance C(V)/C(0) of the ferroelectric NDC varactor on a ferroelectric/Si substrate is shown in Fig.3.3a as function of voltage for various radii of the nano-dot contact. In the calculation, the constants α and β in Eq. (3.5) were chosen to be $\alpha = 2.4 \times 10^{10}$ cm/F and $\beta = 0.75 \times 10^{20}$ cm⁵/C²F from Acikel's work [11]. These values are typical for Barium Strontium Titanate (BST) film, the most widely investigated ferroelectric material for RF/microwave tunable applications. (The same values of α and β are assumed throughout the whole chapter.) The thickness of the ferroelectric film was 5 µm. As shown in Fig.3.3a, reducing the dot size results in a larger tuning factor [$C_0 / C(V)$] with the same applied bias voltage. With a dot radius below 75 nm, a tuning

factor above 1.5 can be achieved by applying a voltage of merely 10 V. It is worth mentioning that a parallel plate varactor built using an identical ferroelectric film with the same thickness (5 μ m) would require more than 100 V to achieve a tuning factor higher than 1.5.

Note, however, that the degree of tunability of a NDC varactor decreases with decreasing ferroelectric film thickness (Fig.3.3b). With a relatively thin ferroelectric film, the drop of the electric field surrounding the dot-contact mainly takes place inside the Si under-layer. The continuity of the electric displacement across the ferroelectric-Si interface, and the much higher dielectric constant of BST ferroelectric compared to Si (370 vs. 12 at zero field), leads to a high electric field in Si just below the interface. Thus, the overall capacitance of the structure becomes dominated by the Si region whose dielectric constant is independent of the applied dc bias field. To better clarify these results, in Fig.3.4 we have plotted the dependence of the capacitance and tuning factor of the varactor on the dot radius and ferroelectric film thickness for a given bias voltage of 10 V.



Figure 3.3: Calculated *C-V* curves of a planar NDC varactor: (a) as function of the dot radius (R_a) for a ferroelectric film thickness of 5 µm, and (b) as function of the ferroelectric film thickness (t_F) for a dot radius of 50nm.

At this stage it is also instructive to consider the case where a ground plane is placed below the ferroelectric layer. This situation is highly relevant for microwave devices such as tunable microstrip transmission lines and phased-array patch antennas which require a thick ferroelectric substrate on a metallic ground plane. In such cases the conventional parallel-plate configuration demands a very high bias voltage (several hundreds or even a few thousands volts) to gain effective tuning. By contrast, the use of nano-contacts will drastically reduce this voltage as shown in Fig.3.5. For the calculation we used the approximation outlined above, but replaced the Si substrate by a perfect conductor. In effect, the same formula's can be applied except for the fact that the zero-field capacitance given by Eq. (3.8) should now be replaced by

$$\frac{1}{C_0} = \frac{1}{4\pi\varepsilon_0\varepsilon_f} \left(\frac{1}{R_a} - \frac{1}{R_b}\right).$$
(3.13)



Figure 3.4: Calculated capacitance and tunability of a planar NDC varactor on silicon (a) as function of the dot radius (R_a) for a ferroelectric film thickness of 5 µm, and (b) as function of the ferroelectric film thickness (t_F) for a dot radius of 50 nm. The applied dc bias voltage was 10 V in both cases.



Figure 3.5: Calculated capacitance and tunability vs. ferroelectric material thickness for (a) a parallel plate capacitor and (b) a NDC capacitor built from a metallic dot (radius 50 nm) placed above a grounded ferroelectric layer. The applied dc bias voltage was 5 V in both cases.

As shown in Fig.3.5, with a nano-dot electrode with a radius of 50 nm, one can even reach a tuning factor higher than 2 at just 5 V, even on ferroelectric films with a thickness exceeding 50 μ m.

It should be mentioned that the calculation presented above is not strictly rigorous since, due to the high electric field near the nano-dot electrode, higher order of polarization should be included in Eq.(3.5) according to LGD theory [10]. Nevertheless,

it is sufficient to illustrate the essential characteristics of the NDC varactor. For practical applications, however, one should bear in mind that the capacitance of a single NDC varactor is too small due to the latter's small radius (Fig.3.5). To achieve a sufficiently high capacitance while maintaining a high tunability, arrays of nano-contacts should be used, which is the subject of the next section.

3.3 Nano-dot varactor arrays

A realistic nano-dot varactor array is schematically shown in Fig.3.6. The ferroelectric film is covered by a conventional dielectric material such as SiO_2 , which is perforated by an array of holes with diameters of the order of tens of nanometers. Once deposited on top of the dielectric, the metal layer directly contacts the ferroelectric film through these holes, forming nano-contacts. At the same time, the metal connects the individual nano-contacts together, forming a capacitor composed of the shunt connection of individual nano-dots. The second electrode of the capacitor may be built on the same layer (as in Fig.3.2a), or as a ground plane below the ferroelectric. In this section, we restrict ourselves to the latter case.

To analyze the capacitor array, we consider an infinite, 2D lattice of nano-dots in the x-z plane, and compute the capacitance per unit area of the structure. This assumption is justified if the number of dots in the array is large. The analysis can then be reduced to that of a unit cell containing a single dot, provided that appropriate boundary conditions on electric field are imposed on the vertical walls of the unit cell. For 2D lattices which are symmetric under reflections with respect to lines forming the boundary of a unit cell (e.g. triangular and square lattices), it can be shown that the normal component of the electric field should vanish on the boundary walls.

Let us now focus our attention on a triangular array of nano-dot varactors with a hexagonal unit cell (Fig.3.6a). To further simplify our calculations, a rotational symmetry with respect to the axis of one single dot is supposed. This is based on the fact that close to the dot, the field and potential almost preserve the rotational symmetry of the dot. As we see in the previous section, the capacitance is mainly determined by the field near the dot. Therefore, this approximation has little impact on the total capacitance results. Combined with the previous approximation, one ends up with a 2D problem which will



Figure 3.6: (a) 3D and (b) cross section view of arrayed NDC varactor.

be once again studied using the electrostatic approximation. The 2D geometry in consideration is sketched in Fig.3.7. Here, symmetry about the y-axis helps to further simplify the domain by half (only the right half is shown). As shown in Fig.3.7, the radius of the nano-dot is r, the thickness of ferroelectric film is t_F , the thickness of SiO2 is d, and the separation of two adjacent dots (center to center distance) is 2*l*. For electrostatic problem, the electric field is usually expressed as the gradient of an electrostatic potential φ . During the calculation, a voltage $\varphi = V$ is applied to the top electrode, the bottom

electrode is kept at $\varphi = 0$. Zero Neumann boundary conditions for the potential, which implies the normal components of electric field along the boundary walls vanish, is applied to boundary walls (*x*=0, $0 \le y \le t_F$) and (*x*=1, $0 \le y \le t_F + d$) by symmetry consideration.



Figure 3.7: Cross section geometry of a unit cell (only right half is shown) of arrayed NDC varactor.

The Laplace equation

$$-\nabla \bullet (\varepsilon \,\nabla \varphi) = 0 \tag{3.14}$$

is applied to both ferroelectric and SiO₂ regions. In SiO₂ region, the dielectric constant term ε is given by $\varepsilon_0 \varepsilon_r$, where ε_0 is again the vacuum dielectric constant and ε_r is the relative dielectric constant of SiO₂, which is taken as 4. In ferroelectric region, we define ε as a local, field-dependent dielectric constant which is given by

$$\varepsilon = \frac{|\boldsymbol{D}|}{|\boldsymbol{E}|} = \varepsilon_0 \left(1 + \frac{\varepsilon_f - 1}{1 + \varepsilon_0 \chi_f \beta P^2} \right).$$
(3.15)

Notice that ε is a function of electric field, thus a function of the potential. The electrostatic potential was solved iteratively using finite element method. At the beginning, the local dielectric constants in the ferroelectric region were assumed to be a constant value. The potential at every point is then solved based on this uniform dielectric constant value by Eq.3.14. Taking the gradient of the potential, electric field can be calculated. Using this electric field value and Eq.3.3, a new local dielectric constant can be derived by Eq.3.15. Substituting the new local dielectric constants into Eq.3.14, potential is again solved at every point. The whole procedures are repeated until the electrostatic potential converges to certain accuracy. A program was developed to carry out the calculation.

The total amount of charge Q_c for one unit cell is found by integration of the perpendicular component of the electric displacement D on the surface of the bottom electrode over its area A. The capacitance per unit cell is defined as

$$C(V) = (1 / A) (dQ_c / dV), \qquad (3.16)$$

which is evaluated using numerical differentiation. In what follows, we present the calculation results for the capacitance and tunability of a NDC array as function of the radius of the nano-dot electrodes, inter-dot separation (center to center distance), and thickness of the conventional dielectric layer (here SiO₂) separating the top global electrode from the ferroelectric. As in the previous section, we assume the ferroelectric to be BST. Since model calculations for a single NDC varactor show practically no change once the thickness of the (grounded) ferroelectric layer exceeds 1 μ m (see Fig.3.5), the same value was used in the numerical simulations to reduce the computation time.

Fig.3.8 shows the calculated tuning factor at a bias voltage of 15 V, as a function of the dot radius (the calculated zero-bias capacitance density is also shown). The thickness of the SiO₂ layer and the separation between two neighboring dots were assumed to be $d = 1 \mu m$ and $2l = 2 \mu m$, respectively (see Fig.3.7). As shown in Fig.3.8, a tuning factor of 2.02 is obtained with a dot radius below 50 nm. In comparison, a parallel plate varactor with the same BST film thickness and applied bias voltage yields a value of 1.30. This

can be seen from Fig.3.8, as the limit $R_d \rightarrow l$ $(l = 1 \ \mu m)$ corresponds to the parallel-plate case (Fig.3.7). Note also that the parallel-plate varactor exhibits an almost twice higher capacitance density compared to a NDC array composed of dots with a radius of 50 nm. The higher tunability is, therefore, obtained at the cost of a lower capacitance density. Nonetheless, in the majority of applications, varactors are intended for providing adjustability, not high capacitance.



Figure 3.8: Calculated capacitance density (at zero voltage) and tuning factor (at 15 V) of a NDC array as function of dot radius. The BST and SiO₂ layers were both 1 μ m-thick, and the inter-dot separation was $2l = 2 \mu$ m in the calculations. A curvature of 30 nm was assumed at the bottom corner of the metallic dot to prevent the electric field exceeding the break down field of the BST ferroelectric.

Figure 3.9 shows the dependence of the tuning factor (and zero-voltage capacitance density) of the NDC array on the inter-dot separation (2*l*). At small values of 2*l*, when the dot-separation becomes comparable to the dot radius, the NDC array varactor behaves like a parallel-plate capacitor. This is because of the strong electric coupling between adjacent dots: the enhanced tuning is washed out as the electric field profile starts to

resemble the parallel-plate case. At sufficiently large separation, the inter-dot coupling becomes weak, and the tuning factor of the NDC array approaches that of a single dot.



Figure 3.9: Calculated capacitance density (at zero voltage) and tuning factor (at 15V) of a NDC array as function of inter-dot separation. A BST and SiO_2 layer thickness of 1 μ m, and a dot radius of 50nm were assumed in the calculations.

Finally, consider the behavior of the NDC array as function of the thickness of the isolating SiO₂ layer, as shown in Fig.3.10. Reducing the SiO₂ layer thickness increases the capacitance density, but yields a lower tuning factor. This can again be understood in terms of the electric field distribution near the dot contact. When the SiO₂ layer is sufficiently thick, the electric field is mostly concentrated near the dot. This is because, away from the dot, the dc bias applied between the top electrode and the ground plane mostly drops over the SiO₂ layer due to the latter's much lower dielectric field near the dot contact ensures a high tuning factor as we saw before. However, decreasing the SiO₂ layer thickness leads to a more uniform field distribution since, now, a larger portion of the electric field reaches the ferroelectric. Eventually, in the limit where the SiO₂ layer thickness becomes zero, a parallel-plate field profile results.


Figure 3.10: Calculated capacitance density (at zero voltage) and tuning factor (at 15 V) of a NDC array as function of the SiO₂ layer thickness. The BST layer was 1 μ m-thick, the dot radius was 50 nm, and the inter-dot separation was 2*l* = 2 μ m in the calculations.

3.4 Discussion and conclusion

We proposed a method for increasing the tunability of ferroelectric varactors utilizing thick (>1 μ m) ferroelectric films. The method was based on metallic nano-dot electrodes deposited on a ferroelectric layer. Using a semi-analytic model, it was found that the tuning factor of a single nano-dot varactor built on a thick BST layer can reach values higher than 2, even with a dc bias voltage as low as 5 V. To achieve sufficiently high capacitance density, we then considered arrays of nano-dot varactors and studied their behavior as function of the dot radius and separation, as well as the thickness of the isolating layer. It was found that the nano-dot array can maintain the high tuning factor of a single dot, provided that the inter-dot distance does not become too small.

While our focus has been on ferroelectric varactors, the application of nano-dot electrodes can be extended to other types of tunable microwave components utilizing

thick ferroelectric films. In particular, one can consider tunable transmission lines which form the basis of phase shifters and electrically adaptable filters. As an example, consider a microstrip line built on a thick, grounded ferroelectric substrate. By covering the ferroelectric with a conventional dielectric layer (e.g. SiO_2) perforated by an array of sub-micron holes and, subsequently, depositing the microstrip line, a highly tunable transmission line can be formed. The capacitance per unit length and tunability of such a microstrip can be calculated using the static field analysis method presented earlier at not too high frequencies (as far as a quasi-TEM approximation is valid). The detailed results are beyond the scope of this work and will not be presented here.

The proposed nano-dot varactor may offer another potential benefit in comparison with conventional devices. As is well known, one of the major sources of dielectric loss in a ferroelectric is the oxygen vacancies within the grain boundaries of the ferroelectric film [12]. Because the electric field rapidly drops away from the metallic nano-electrode, only a few grains are expected to be involved in determining the capacitance and loss of the device (BST thin film for example has a typical grain size of 60-70 nm). Therefore, lower loss can be expected from nano-dot capacitors and capacitor arrays. However, experiments are required to confirm this argument.

References

- A. Deleniv, S. Gevorgian, H. Jantunnen, T. Hu, "LTCC compatible ferroelectric phase shifters", 2005 IEEE MTT-S Digest, Vol. 2, pp. 599-602.
- [2] W. Hu, D. Zhang. M.J. Lancaster, T.W. Button, B. Su, "Investigation of ferroelectric thick-film varactors for microwave phase shifters", IEEE Trans. Microw. Theo. Tech., Vol. 55, No. 2, pp. 318-323, 2007.
- [3] W. Hu, D. Zhang, M.J. Lancaster, K.S.K. Yeo, T.W. Button, B. Su, "Cost effective ferroelectric thick film phase shifter based on screen-printing technology", 2005 IEEE MTT-S Digest, Vol. 2, pp. 591-593.
- [3] A. Deleniv, A. Eriksson, S. Gevorgian, "Design of narrow-band tunable band-pass filters based on dual mode SiTiO₃ disc resonators", 2002 IEEE MTT-S Digest, Vol.2,

pp. 1197-1200.

- [5] A. Deleniv, A. Eriksson and S. Gevorgian, "Four-pole tunable band-pass filter based on two dual mode SiTiO₃ disc resontors", 32nd Europ. Microw. Conf. 2002, pp. 1-3.
- [6] R. Ott, R. Wordenweber, "Improved design of tunable ferroelectric capacities for microwave applications", Appl. Phys. Lett., Vol. 80, No. 12, pp. 2150-2152, 2002.
- [7] G.D.J. Smit, S. Rogge, T.M. Klapwijk, "Scaling of nano-Schottky-diode", Appl. Phys. Lett., Vol. 81, No. 20, pp. 3852-3853, 2002.
- [8] G.D.J. Smit, M.G. Flokstra, S. Rogge, T.M. Klapwijk, "Scaling of micro-fabricated nanometer-sized Schottky diodes", Microelec. Eng., Vol. 63, No. 1, pp. 329-333, 2002.
- [9] N. Setter, D. Damjanovic, L. Eng, G. Fox, S. Gevorgian, S. Hong, A. Kingon, H. Kohlstedt, N.Y. Park, G.B. Stephenson, I. Stolitchnov, A.K. Taganstev, D.V. Taylor, T. Yamada, S. Streiffer, "Ferroelectric thin films: review of materials, properties, and applications", J. Appl. Phys., Vol.100, pp. 051606/1-36, 2006.
- [10] C. Kittel, "Introduction to Solid State Physics" (8th edition, John Wiley & Sons, Inc, 2005), pp. 375.
- [11] B. Acikel, "High performance barium strontium titanate varactor technology for low cost circuit applications", Ph.D dissertation, University of California Santa Barbara, (2002).
- [12] A. Vorobiev, P. Pundqvist, K. Khamchane, S. Gevorgian, "Microwave loss mechanisms in Ba0.25Sr0.75TiO3 thin film varactors", J. Appl. Phys., Vol. 96, No. 8, pp. 3632-3639, 2003.

Chapter 4

On-chip Passives with Artificial Dielectric Layer (ADL) Shields

4.1 Introduction

Transmission lines are extensively used in microwave circuits, not only as wave guiding structures to transfer signal power from one point to another, but also as passive elements used for processing the microwave signal. Examples are $\lambda/4$ transformers, transmission line resonators, power dividers, and filters [1].

Most people are familiar with coaxial transmission lines which have been widely applied in microwave systems. Over the past 40 years, the development of solid state circuits and IC technology has dramatically changed microwave engineering. With the availability of high quality microwave solid state transistors, the focus of wave guiding components has changed to planar transmission line structures, such as microstrip and coplanar waveguides (CPW). With the current trend and enormous focus on monolithic microwave integrated circuits (MMIC), development of high-quality on-chip planar transmission lines has attracted much attention from the scientific and engineering community [2].

Nonetheless, monolithic integration of high-quality microwave transmission lines still remains a challenge due to performance degradation caused by losses in the underlying silicon (Si) substrate, the main vehicle of the contemporary radio-frequency integrated circuits (IC). It is the conductive nature of Si which allows the flow of electric conduction currents in the substrate in response to the electric field surrounding the transmission line conductors. In turn, the dissipation of microwave power brought about by these currents is manifested as a rise in the overall device loss [3-5].

To illustrate the substrate loss effects, the attenuation of typical CPW lines, with signal line width of 30 μ m and signal to ground width of 30 μ m, was simulated in a microwave simulator, Advanced Design System (ADS). Since our interest is the substrate loss effect, the metallic signal line and ground are assumed to be perfect metal in simulations. In Fig.4.1, the simulated substrate losses are shown as a function of frequency with different values of Si substrate resistivity. With lower substrate resistivity, the line shows higher substrate loss, especially at high frequencies. Although substrate loss can be suppressed by using insulating substrates [3], micromachining [4], and substrate transfer techniques [5], such methods involve modification of the core device integration process, leading to significantly higher fabrication costs.

Electric shields built from metal or poly-silicon layers offer an IC-compatible alternative. Inserted beneath the device, the shield blocks the electric field from entering the conductive Si substrate, preventing the flow of unwanted currents [6-8]. Already coplanar waveguide (CPW) transmission lines with shields built from floating metal strips perpendicular to the line have been demonstrated to have very low substrate loss [8]. However, the inherently large parasitic capacitance between the CPW and the conductive shield substantially reduces the line characteristic impedance. Consequently, shielded lines with high characteristic impedance require narrow signal lines or large signal to ground spacing, which increase the line attenuation or device area. For example, in Cheung's work [8], the signal line width of a shielded line is smaller than un-shielded line by a factor of 3, with the same values of signal to ground space and characteristic impedance. This is a crucial disadvantage in the implementation of resonant tanks, impedance matching networks and couplers, where a wide range of characteristic impedances (30-300 Ω) is required.



Figure 4.1: ADS Simulation results of CPW line attenuation as a function of frequency for different substrate resistivity.

In this chapter we describe a new IC-compatible shielding method for integrated CPW's utilizing an artificial dielectric layer (ADL) consisting of a conventional, thin dielectric film sandwiched between two patterned metal layers [9]. The ADL effectively behaves as a thin homogeneous dielectric with a very high dielectric constant. Using the concept of surface impedance, it can be shown that an ADL layer possesses a very high in-plane dielectric constant. This property of the ADL leads to the prevention of the electric field of the CPW from entering the Si substrate. Our experiments show the ADL shield to yield an up to three-fold reduction of the line attenuation below 30 GHz. Furthermore, the effect of the ADL on the characteristic impedance of the CPW is much smaller than that of conventional shields, allowing the realization of a broad range of characteristic impedances without the need for narrow signal lines or large signal-ground spacing. An additional benefit of the ADL, compared to conventional shielding structures, is the independence of its design of the individual CPW layout, simplifying the mask design and fabrication process [9].

Moreover, we will also show that the shielding effect can be further improved by

micro-patterning the shielding structures [10], i.e. the metal blocks used in ADL, or by using multi-layer ADLs. The former method is aimed at reducing the eddy current loss in the metal blocks which can contribute significantly to the total loss at high frequencies. The latter technique, one the other hand, is employed to increase both the effective dielectric constant and thickness of the ADL. Finally, we will investigate the application of the ADL shield to on-chip spiral inductors, another important and widely used microwave passive component.

In the next section, we describe the principle of shielding by a high-k (high dielectric constant) for integrated transmission lines on lossy substrates.

4.2 Integrated coplanar waveguide (CPW) transmission lines with ultra-high-k dielectric shields

The most widely used transmission lines in microwave IC technology are the microstrip line and the coplanar waveguide transmission line (CPW), which are schematically shown in Fig. 4.2. The microstrip line consists of a narrow stripe of metal placed on top of a grounded substrate. While the microstrip line is common in traditional hybrid circuits built on insulating dielectric substrates, its use in modern silicon ICs is problematic. The conductive nature of conventional silicon material will cause significant losses due to the flow of substrate conduction currents. Moreover, making connections to the metallic ground plane on the back of the silicon wafer requires the use of vertical, through-wafer vias whose implementation in silicon IC's is technologically involved.

An alternative is to implement both the microstrip and ground plane on the wafer front side by using the metal-dielectric stack of a modern IC process. The ground plane in this construction shields the electric field of the microstrip from the silicon substrate, and is easily accessible through commonly used metallic vias [11]. But, due to the small thickness of the dielectric (several um's), the characteristic impedance of such lines is rather small. The only remedy is to reduce the microstrip width which, in turn, causes the resistance per unit length and line attenuation to rise significantly.

In view of these issues, coplanar waveguides constitute a more viable option when it comes to chip integration. The signal and ground conductors of a CPW are built on a single metal layer which is placed above a Si substrate isolated by a silicon oxide layer of a few microns in thickness. Due to its open nature, however, a CPW line is also prone to substrate loss, albeit to a lesser degree compared to conventional microstrip lines as the electric field lines remains close to the conductors and do not traverse the whole substrate thickness. Despite this shortcoming, integrated CPW lines remain popular due to their ease of fabrication and the wide range of characteristic impedances realized.



Figure 4.2 Schematic view of on-chip transmission lines with microstrip (left) configuration and coplanar-waveguide (CPW) configuration (right).

An IC compatible technique to reduce the substrate loss of CPW lines is to use electric shields built underneath the transmission line. Already coplanar waveguide (CPW) transmission lines with shields built from floating metal strips perpendicular to the line have been demonstrated to have very low substrate loss [8]. However, the large parasitic capacitance between the CPW and the conductive shield substantially reduces the line characteristic impedance. Therefore, shielded lines with high characteristic impedance require narrow signal lines or large signal to ground spacing, which increase the line attenuation or device area.

In this section we theoretically investigate an alternative shielding method based on a thin, ultra-high-k dielectric film which is inserted between the CPW and the silicon substrate. To demonstrate the shielding effect, we simulate the electric field distribution on the cross section of CPW line. For CPW lines at not too high frequency, quasi-TEM approximation can be made. Therefore, we look at the quasi-static field distribution by solving two dimensional Laplace's equation for the electric potential on the cross section

73



Figure 4.3 Schematic view of electric field distribution of a CPW transmission line (a) without and (b) with super-k shield.

of the structure. This has been done in Ansoft's Maxwell SV. Fig. 4.3 shows the calculated electric field distribution without (a) and with (b) high-k shield. As shown in Fig. 4.3a, without the high-k shield, a significant portion of electric field lines, and therefore electromagnetic energy will penetrate into Si substrate, leading to substrate losses. With the high-k shield inserted (Fig.4.3b), the electric displacement field, and therefore electric field, tends to be confined in the high-k thin film. As a result, much less electromagnetic energy will leak into Si. The shielding effect can be also understood from a circuit theory point of view: the high-k film can be modeled as a large capacitance and provides an alternative low impedance, low loss route for RF current to flow instead of

entering into the lossy Si substrate. The equivalent circuit model of CPW with high-k shield will be discussed in more detail later.

To better verify the picture presented, simulations were performed for CPW lines on a Si substrate with the high-k shield using the microwave simulator in ADS. In the simulations, CPW lines with signal line width of $w=30 \ \mu m$ and signal-ground space of $s=15 \ \mu m$ were assumed. The substrate stack consists of a 10 Ω cm Si substrate of 500 μm in thick, covered by a 1 μm -thick high-k film 4 μm of SiO₂. For comparison, simulations were also performed on a reference CPW line with the same geometric configurations (wand s), but with the shield replaced by 1 μm of SiO₂. Since we are interested the substrate shielding effect, the conductors and the dielectrics were assumed lossless. The dielectric constants of the super-k layer were chosen to be 500, 1000, 2000 and 5000. The attenuations of the lines were extracted from S-parameter simulations and a comparison of the simulated line attenuations as a function of frequency for different values of dielectric constant is shown in Fig.4.4.



Figure 4.4: ADS Simulation results of CPW line attenuation as a function of frequency with super-k thin film shield for different dielectric constant values. For comparison, CPW line attenuation without shield is also shown.

As shown in Fig.4.4, the ultra-high-k thin film shield can significantly suppress the substrate loss and, thus, reduce line attenuation. The degree of suppression directly depends on the dielectric constant of the shield. At 30 GHz, 1µm thick super-k thin film with a dielectric constant of 500 can reduce the substrate loss by more than 2-fold, and 20-fold reduction of substrate loss can be achieved by super-k film of the same thickness with a dielectric constant of 5000.

Although the simulation shows promising results for loss reduction by super-k materials, the realization of super-k thin films shield on Si is prohibited currently by technological difficulties. Ferroelectric materials are well known to have extremely high dielectric constant values (on the order of 10^5), which make them attractive for this application. However, in thin film forms, ferroelectric materials exhibit much lower value of dielectric constant (3 orders lower) compared to their bulk ceramic forms, especially on Si substrates. Besides, unlike in the simulation presented, realistic ferroelectric films may exhibit high dielectric loss at microwave frequencies which might partially reduce the shielding efficiency. Nevertheless, the idea of electric shield by thin film with super high dielectric constant is still a solid one. This can be accomplished by artificial materials as presented in the next section.

4.3 ADL as a shield

4.3.1 Concept of ADL

As introduced in Chapter 1, an artificial dielectric is a special kind of man-made material which exhibit electromagnetic properties similar to a dielectric material. Physically, it is a large scale model of a real dielectric. Normally, it consists of a large number of identical conducting elements periodically arranged in a three-dimensional or two-dimensional pattern in a host conventional dielectric. A typical structure of artificial dielectric is metallic spheres or disks in a three dimensional lattice. Here we introduce a new kind of artificial dielectric formed by isolated metallic squares embedded in a conventional dielectric material. Since this new kind of artificial material manifests itself as an effective dielectric thin film, we nominate this new structure artificial dielectric layer, or simply, ADL as abbreviation.

Figure 4.5 shows the ADL built from two thin metal layers vertically separated by a thin dielectric film with the thickness t_d and dielectric constant ε_d . Each metal layer is patterned to form a lattice of disconnected, but closely spaced squares. The lattices on the

two metal layers are shifted with respect to each other so that each square on the top lattice partially faces four squares on the bottom lattice and vice versa [9, 12-13]. To reduce the flow of eddy currents inside individual patterns at high frequencies, their dimension is chosen in the micrometer range.



Figure 4.5 (a) and (b): Geometrical layout of the ADL consisting of two thin, patterned metal layers isolated by a thin dielectric. The lattices of squares on the two metal layers are shifted with respect to each other so that each square on the top lattice partially faces four squares on the bottom lattice and vice versa. (c): Cross section of an integrated CPW line shielded by an ADL inserted between the Si and oxide (thickness *h*) layers.

To describe the dielectric properties of the ADL we use an effective medium approach which is justified at the frequencies of interest in this work (<30 GHz) where the electromagnetic wavelength exceeds by several orders of magnitude the dimension and spacing of metallic squares (both in the micrometer range).

The effective properties of the ADL can be determined by modeling the latter as a two-dimensional network of identical capacitors C in series with resistors R (Fig.4.6).



Figure 4.6 Network theoretical model of the ADL based on identical capacitors C and resistors R.

The capacitors are formed between overlapping metallic squares whereas the resistors take account of the series resistance of the patterns. Using the electric network theory and applying the continuum approximation, one finds the surface impedance of the ADL (ratio between the in-plane electric field and the surface density of the RF electric current flowing on the layer) to be given by $Z_s = R + (1/j\omega C)$. This can be understood by the following arguments. Consider the complex power delivered to a unit cell of the ADL network (as shown in Fig.4.6) and suppose the RF current of *I* flowing into node A. By symmetry consideration, the RF current of each segment is I/2. Therefore, the complex power is given by adding up the complex power in each segment, resulting in

 $P = (R + 1 / j\omega C)I^*I$. This can be equated to $P = ZI^*I / 2$, where Z is the impedance of

this unit cell. It is immediately seen that the impedance of the unit cell is given by $Z = R + (1 / j\omega C)$. Due to the symmetry of the circuit network, the surface impedance (the impedance per unit area) can be shown to equal to the impedance per unit cell as given by the result presented above. Because a thin dielectric film with an in-plane

permittivity ϵ_{\parallel} and thickness t has a surface impedance $1/j\omega\epsilon_{\parallel}t$, it follows that the effective in-plane permittivity of the ADL is given by

$$\epsilon_{\parallel} = \varepsilon_{\parallel} / \left(1 + j\omega\tau \right), \tag{4.1}$$

where $\varepsilon_{\parallel} = C/t$, $\tau = RC$, and t is the overall thickness of the ADL. In the structures studied in this thesis, it can be shown that the product $\omega \tau$ is negligible in the frequency range of interest so that $\epsilon_{\parallel} \approx \varepsilon_{\parallel}$. The $\omega \tau$ term actually contributes to the loss tangent of the ADL as will be discussed later. For now, we are interested in the real part of the dielectric constant and just safely ignore this term. Since the thickness t_d of the dielectric layer is extremely small (several tens of nanometers), one can neglect the fringe field and estimate the value of *C* according to

$$C = \varepsilon_d A / t_d, \qquad (4.2)$$

where ε_d is the dielectric constant of the dielectric layer and A is the overlap area between two squares partially facing each other (Fig.4.3). As a result,

$$\varepsilon_{\parallel} = C / t = \varepsilon_d A / t_d t . \tag{4.3}$$

It is important to note that A is in square microns whereas t_d and t do not exceed several tens, respectively, hundreds of nanometers. Therefore, ε_{\parallel} is typically several orders of magnitude larger than ε_d . The high value of ε_{\parallel} can be exploited to shield the Si substrate from RF electric fields induced by a CPW transmission line as discussed in the previous section.

The effective dielectric constant ε_{\perp} of the ADL in the direction normal to its plane can be easily estimated when the spacing between the patterns is much smaller than their size. For perpendicular electric fields, the ADL then practically behaves as two continuous metal layers separated by a dielectric film. Here we consider a general situation where N dielectric layer with dielectric constant of ε_d is sandwiched by (N+1)metal layers (as shown in Fig.4.7). The total capacitance of the structure is just N capacitors in series and is given by $C = \varepsilon_d A / Nt_d$, where A is the overlap area and td is the thickness of one dielectric layer. On the other hand, the total capacitance can also be calculated by defining an effective dielectric constant ε_{eff} of the overall structure, and resulting in $C = \varepsilon_{eff} A/t$ where t is the thickness of the complete structure. By comparing the two expressions, one finds $\varepsilon_{eff} = \varepsilon_d t / Nt_d$. In our ADL case, N is simply equal to 1 and this results in $\varepsilon_{\perp} \approx \varepsilon_d t / t_d$. Since t and t_d do not differ by orders of magnitude, ε_{\perp} is much smaller than ε_{\parallel} .



Figure 4.7: A simple model to calculate the perpendicular effective dielectric constant of general ADL structure.

The dielectric loss of the ADL is characterized by the total loss tangent which includes the dielectric loss of the insulating dielectric material as well as the pattern resistance (*R* in Fig.4.4). Substituting ε_d by $\varepsilon_d(1 - j \tan \delta_d)$ in Eq. (4.2) and combined with Eq. (4.1), it can be shown that the total dielectric loss of the ADL is given by

$$\tan \delta = \tan \delta_d + \tan \delta_M, \qquad (4.4)$$

where $\tan \delta_M = \omega \tau = \omega RC$ is the loss brought by the pattern resistance.

Now consider the magnetic losses. Under influence of a magnetic field perpendicular to the patterns, planar eddy currents start to flow in the plane of the patterns. Such currents, in turn, generate a dipolar magnetic field which opposes the original field. This effect can be represented by an effective susceptibility χ_{\perp} , i.e. the ratio between the volume average of the induced magnetic dipoles and the original field. For patterns with

a thickness far below the skin depth inside the metal, an approximation assuming a constant magnetic field yields

$$\gamma = \frac{\chi_{\perp}}{1 + \chi_{\perp}} = -j\eta_M f \,\omega\mu_0 \sigma_M D^2 \,, \tag{4.5}$$

where D is the diameter of the patterns, f is their volume filling factor, and η_M is a dimensionless, geometry-dependent factor. For square patterns $\eta_M \sim 1/32$. Detailed derivation is shown in Appendix C.

4.3.2 Shielding effect of ADL

The shielding effect of the ADL on the electric field of a Si-based CPW line (Fig. 4.3c) can be demonstrated using the quasi-TEM approximation [14] where the (transverse) electric field is expressed as the gradient of an electrostatic potential $\varphi(x,y)$ on the line cross section. To simplify the analysis we assume an unbounded substrate and an infinitely thick Si layer. Solving the spectral domain Laplace equation (see Appendix D), the Fourier transform (in the *x*-direction) of φ inside the Si layer (*y*<0) can be expressed as

$$\tilde{\varphi}_{\rm si}(k,y) = \frac{\exp(|k|y) \int_{-\infty}^{\infty} V(x) \exp(ikx) dx}{A(k) \cosh(|k|h) + B(k) \sinh(|k|h)},\tag{4.6}$$

where $V(x) = \varphi(x, h+t)$ is the potential on the (top) oxide surface, *h* is the oxide layer thickness (see Fig.4.3c), and

$$A(k) = \cosh(|q|t) + (\varepsilon_{\rm si} / \varepsilon_{\rm e})\sinh(|q|t)$$
(4.7)

$$B(k) = \varepsilon_{\text{ox}}^{-1} \left[\varepsilon_{\text{si}} \cosh(|q|t) + \varepsilon_{\text{e}} \sinh(|q|t) \right]$$
(4.8)

where ε_{ox} is the dielectric constant of the oxide layer, $\varepsilon_{e} = (\varepsilon_{\parallel} \ \varepsilon_{\perp})^{1/2}$, $q = k(\varepsilon_{\parallel} / \varepsilon_{\perp})^{1/2}$, and $\varepsilon_{si} = \epsilon_{si} - j\sigma_{si} / \omega$ with ϵ_{si} the dielectric constant and σ_{si} the conductivity of Si. The (Fourier transformed) electric field inside Si is $(\tilde{E}_{x}, \tilde{E}_{y}) = (ik, -|k|)\tilde{\varphi}_{si}$.

Eqs. (4.6-4.8) show that, for any non-uniform field ($k \neq 0$), a high value of ε_{\parallel} reduces $\tilde{\varphi}_{si}$ and, therefore, the electric field and power dissipated inside the conductive Si. In particular, since the lateral variation of the field over distances comparable to the ADL

thickness (<1 μ m) is negligible in typical CPW configurations, it suffices to consider the limit *kt*<<1, where

$$A(k) \approx 1 + (\varepsilon_{\rm si} / \varepsilon_{\perp}) |k| t , \qquad (4.9)$$

$$B(k) \approx \varepsilon_{\text{ox}}^{-1} (\varepsilon_{\text{si}} + \varepsilon_{\parallel} \mid k \mid t) .$$
(4.10)

Evidently, increasing ε_{\parallel} leads to an increase in B(k) and, therefore, a reduction of $\tilde{\varphi}_{si}$. (Although V(x) is given below the CPW conductors, a rigorous treatment of this problem requires its self-consistent computation inside the CPW slots. But the argument presented is sufficient to show the shielding effect.)

Note that the magnetostatic equations of the quasi-TEM approach [14] are not influenced by the dielectric properties of the medium. Therefore, as long as the quasi-TEM approximation holds (i.e. at not too high frequencies), the effect of the ADL on the magnetic field of the CPW is expected to be negligible.

4.4 Experimental results of CPW lines

4.4.1 CPW with ordinary ADL shields

CPW transmission lines were fabricated on a 5-10 Ω ·cm n-type silicon wafer in a standard aluminum CMOS production line. The ADL shields were built using two 100 nm-thick aluminum layers isolated by a 30 nm-thick PVD-sputtered aluminum oxide film with a relative dielectric constant of 10. For each CPW line, three different ADL structures were implemented by patterning the metal layers into 3 x 3, 5 x 5, and 7 x 7 μ m² squares. In all cases the squares were laterally separated by 1 μ m-wide gaps. Taking the 7 x 7 case as an example, this structure yields an overlap area of $A = 9 \mu$ m², resulting

in an effective in-plane dielectric constant of $\varepsilon_{\parallel} = \varepsilon_d A / t_d t \approx 13000$ for the ADL.

The ADL was isolated from the silicon substrate by a 0.75 μ m thermally grown silicon oxide layer which is the typical distance from substrate to the first metal layer in CMOS integration. On top of the shield structure, a 9 μ m PECVD silicon oxide was deposited and the CPW line was fabricated using the top metal level (3 μ m-thick aluminum). The micrograph of one CPW line after processing is shown in Fig.4.8. The shield structure is shown in the enlarged area. CPW lines with different signal line width

(w) and signal to ground spacing (s) were designed to achieve different characteristic impedances.



Figure 4.8: Micrograph of a CPW transmission line with an ADL substrate shield. The signal line width and signal to ground spacing of the CPW are denoted by *w* and *s*, respectively. The enlarged portion of the ADL shield clearly shows the two (overlapping) metallic square lattices.

The propagation constant (γ) and characteristic impedance (Z_0) of the CPW lines (of length *l*) were found from (see Appendix E)

$$\gamma = (1 / l) \operatorname{arctanh}\left(\sqrt{Z_{\text{short}} / Z_{\text{open}}}\right)$$
 (4.11)



Figure 4.9: Attenuation vs. frequency of a CPW line ($w=30 \ \mu m$, $s=30 \ \mu m$) shielded by ADL's with 3x3, 5x5, and 7x7 μm^2 patterns. Results for an unshielded line are shown for comparison. Dashed lines show the HFSS simulation results with the ADL modeled as an anisotropic dielectric. Taking into account a ~0.3 μm reduction of the ADL pattern size due to lithography issues, a value of $\varepsilon_{\parallel} \sim 700$, 4200, and 10000 was calculated for the three shields, respectively, and used in the HFSS simulations. $\varepsilon_{\perp} \sim 75$ in all cases. A Si resistivity of 7 Ω cm was assumed in the simulations.

$$Z_0 = \sqrt{Z_{\text{open}} \cdot Z_{\text{short}}} , \qquad (4.12)$$

where Z_{open} and Z_{short} denote the impedance of the open- and short terminated lines, respectively, extracted from S-parameter data obtained using a HP8510C vector network analyzer. Other transmission line parameters (i.e. capacitance per unit length *C*, inductance per unit length *L*, serious resistance per unit length *R* and shunt conductance per unit length *G*) can also be extracted from S-parameter measurement as derived in

Appendix E. As shown in Appendix E, for integrated CPW line, the metal loss is modeled as the serious resistance per unit length and the substrate loss is modeled as the shunt conductance per unit length.

Fig.4.9 shows the attenuation constant $\alpha = (20 \log e) \operatorname{Re} \gamma$ of a shielded CPW ($w=30 \mu m$, $s=30 \mu m$) in comparison with that of an identical, but unshielded line. The effectiveness of shielding clearly increases with the ADL pattern size. At frequencies below 30GHz, the ADL with the largest patterns (7 x 7 μm^2) yields a nearly three fold reduction of α compared to the unshielded device. HFSS simulations based on the effective description of the ADL yield lower values of α compared to the experimental data (Fig.4.9). Especially at high frequencies, this can be attributed to the inability of the dielectric model to account for the frequency dependant eddy current loss inside the individual patterns of the ADL. Nevertheless, the overall agreement is reasonable below 20 GHz.

Fig.4.10 compares other extracted transmission line parameters of shielded CPW lines with the same geometric configurations ($w=30 \mu m$, $s=30 \mu m$), but different ADL pattern sizes. Unshielded line results are also shown for comparison. As shown in Fig.4.10(a), ADL shield manifest itself by significantly reducing the substrate loss. The substrate loss reduction effect increases with the ADL pattern size. A six-fold substrate loss reduction near 30 GHz can be achieved by applying the ADL with a 7x7 μm^2 pattern size. Fig.4.10(b) compares the inductance per unit length among different CPW lines. As mentioned in section 4.3.2, the ADL behaves mainly like an effective dielectric layer, the magnetostatic equations of the quasi-TEM approach are not influenced by the existence of ADL. Effect of ADL on the magnetic field, i.e. the extracted inductance per unit length, of the CPW line is expected to be negligible. Therefore, as illustrated in Fig.4.10(b), different CPW lines show almost identical inductance per unit length values throughout the whole frequency range.

The effect of ADL on the capacitance per unit length value is shown in Fig.4.10(c) and can be understood with the help of a simplified lumped element model. Fig.4.11 shows the resistive-capacitive equivalent network of a CPW line with and without a shield. The oxide capacitance is denoted by C_{ox} whereas C_{sub} accounts for the capacitance due to electric field penetrating into Si. The resistors R_{sub} account for the substrate conductivity and are responsible for the substrate loss. In the case of CPW with the ADL shield, the substrate network is shunted by equivalent ADL capacitors. Because of the



Figure 4.10: Extracted (a) substrate loss, (b) inductance per unit length, (c) capacitance per unit length and (d) characteristic impedance vs. frequency of a CPW line ($w=30 \mu m$, $s=30 \mu m$) shielded by ADL's with 3x3, 5x5, and 7x7 μm^2 patterns. Results for an unshielded line are shown for comparison. It shows that the substrate loss alone has about six-fold reduction of a 7x7 μm^2 ADL shielded CPW line compared with the unshielded line at high frequencies. The ADL has little effects on the inductance per unit length value due to the mainly dielectric properties of the ADL. In contrast, ADL modifies the capacitance per unit length value as expected by its different effective in-plane dielectric constant. As a result, the characteristic impedance also changes at high frequencies due to the insertion of the ADL.

high effective (in-plane) dielectric constant of the ADL, this capacitance may be much larger than the Si capacitance. This will reduce the current flowing through the substrate resistors (shielding effect) but will also lead to an increase in the total capacitance of the CPW line. Note that the inductance per unit length is not affected by the inclusion of a high-k shield. Consequently, the inclusion of the shield will cause the characteristic impedance, which can be simplified as $\sqrt{L/C}$ when the losses are low, of the line to drop. This effect will become stronger with increasing of the pattern size, as shown in Fig.4.10(d).



Figure 4.11: Cross section view and simplified lumped-element models of a CPW line on silicon with (a) and without (b) ADL shield.



Figure 4.12 Attenuation (a) and real part of the characteristic impedance (b) of shielded (solid markers) and unshielded (hollow markers) CPW lines with the signal line width w and signal to ground spacing s.

Fig.4.12(a) shows the attenuation versus frequency for three shielded CPW lines with different values of w and s. Results for identical, but unshielded lines are also shown for

89

comparison. A two- to three fold reduction of line attenuation is observed for all lines up to 30 GHz. Note that Fig.4.12(a) shows the total line attenuation including the conductor loss. Otherwise, results for the substrate loss show an up to six-fold reduction due to insertion of the ADL (for example the one shown in Fig.4.10a). These results are comparable with those found by Cheung et al [8].

Fig.4.12(b) compares the characteristic impedances of the shielded and unshielded lines with different geometric configurations. All the shielded lines have an ADL with 7 x 7 μ m² patterns. In contrast with the unshielded lines, the characteristic impedance of the shielded CPW's changes little as function of frequency and has a very small imaginary component. This is due to the ADL masking the influence of the complex permittivity

 $\varepsilon_{si} = \epsilon_{si} - j\sigma_{si} / \omega$ of silicon which has a large, frequency-dependent imaginary part. Note

that, even in presence of the ADL, a wide range of characteristic impedances can be realized without the need for extremely narrow signal lines or large signal-ground spacing. This is in contrast with results obtained using a floating metallic shield [8]. In both cases, the shield structures introduce a low impedance path, which are the ADL and the metallic stripes respectively, between the parasitic capacitances formed between the CPW conductors and the shields, leading to an increase of capacitance per unit length of the CPW line and a decrease of the characteristic impedance. However, due to its high conductivity, the latter has a much bigger effect on the characteristic impedance. As a result, CPW lines with high characteristic impedance are difficult to be realized using conventional metallic shield while still keeping the metallic loss or signal to ground space low. For ADL shielded CPW lines, a tradeoff between high efficiency of shield and high characteristic impedance can be made. Note that, in our experiments, the ADL lowers the characteristic impedance of the CPW lines up to $\sim 15\%$ due to an increase in the line capacitance per unit length. Therefore, ADL-based CPW lines with a wide range of characteristic impedances can be realized without the need for extremely narrow signal lines or large signal-ground spacing.

Based on the experimental results, it is clear that the shielding effectiveness of the ADL can be further increased by increasing its in-plane dielectric constant ε_{\parallel} . This can be achieved by increasing the pattern size, provided that the flow of parasitic eddy currents inside individual patterns can be effectively suppressed, which will be the content of the next section (section 4.4.2). ε_{\parallel} can also be increased by reducing the thickness of the

dielectric layer, or using a dielectric with a high intrinsic dielectric constant such as TiO_2 . Finally, in view of the many metallization layers available in modern IC processes, one

can stack several ADL's on top of each other to increase the shielding effect (see section 4.4.3).

It is worth mentioning that, unlike conventional shielding structures [6-8], the geometrical patterning of an ADL does not depend on the layout of individual CPW lines (e.g. their direction). Therefore, a predefined uniform ADL shield can be built underneath the whole passive (sub-) circuit, which simplifies the mask design and fabrication process.

4.4.2 CPW with micro-patterned ADL shields

Although the application of ADL shields results in significant reduction of the substrate loss, the overall performance gain is limited by the eddy currents generated inside the metallic patterns of the ADL itself. At microwave frequencies, eddy currents flow inside the individual metallic particles of the ADL in response to the perpendicular component of the magnetic field of the CPW. These currents, which undergo ohmic dissipation inside the metallic patterns, manifest themselves by increasing the conductor loss (i.e. resistance per unit length) of the transmission line. This effect, which becomes more prominent at higher frequencies, is not accounted for by the dielectric model of the ADL (see Fig.4.9). Using smaller patterns reduces the eddy current loss, but at the expense of a lower effective dielectric constant and, thus, less effective shielding.

In this section, a new type of ADL will be demonstrated. The new type of ADL shield, while maintaining a very high shielding efficiency, is much less prone to eddy current loss. This is achieved by micro-patterning the individual metallic elements of the ADL as shown in Fig.4.13. Employing these shields, we have realized CPW lines on an ordinary Si substrate with an overall attenuation comparable to identical lines built on a high resistivity Si (HRS) wafer. This provides an alternative to the expensive high-resistivity substrates as the vehicle for high-quality microwave transmission lines.

The principles of the eddy current loss reduction by micro-patterning of the metallic elements can be understood by considering the eddy current distribution in a metallic square under influence of a perpendicular magnetic field. Fig. 4.14 shows simulation results of current density distribution of metallic squares with and without



Figure 4.13: Top view (left) of a micro-patterned ADL. The micro-patterned ADL is realized by etching slots near the edges of each square pattern (dashed lines in the left figure). The schematic layout of an integrated CPW line equipped with an ADL shield is shown as well (right).

micro-patterning under uniform magnetic field perpendicular to the paper. As shown in the figure, the current density is relatively large near the edges where it dominates the ohmic loss (which is quadratic in current). Cutting narrow slots close to the edges of the metallic square reduces the flow of these edge currents, resulting in a lowering of the total ohmic dissipation inside the pattern. Note that by placing the slots outside the overlap area between squares on different lattices, the in-plane dielectric constant of the ADL will not be affected.

Integrated CPW lines with ADL shields were fabricated using similar processing steps as described in the previous section. The CPW lines were made by 3 µm-thick aluminum (Al) metallization on a standard 5-10 Ω ·cm Si wafer, isolated by a 9 µm-thick silicon oxide (SiO₂) layer. The ADL shields were built on top of Si using a 30 nm-thick, sputtered aluminum oxide (Al₂O₃) film sandwiched between two 100 nm-thick, patterned Al layers. Both conventional and micro-patterned ADL shields were fabricated for comparison. In the first case, conventional ADL was built using 15x15 µm² Al squares laterally separated by 1µm-wide gaps. In the latter case, each 15x15 µm² square was additionally patterned by etching slots perpendicular to its sides (Fig.4.13). In both cases the overlap area between metallic elements on the two lattices was $A = 49 \ \mu m^2$ which, with $\varepsilon_d \sim 10$, yields a relative in-plane dielectric constant of $\varepsilon_{\parallel} \sim 70000$.

To compare the performance of the shielded lines to those implemented on a



Figure 4.14: Calculated eddy current density distribution in a single, 100 nm-thick metallic (Al) element of a conventional (left) and micropatterned (right) ADL in response to a perpendicular, magnetic field (*B*). The field is assumed to be uniform for simplicity. Calculations were based on the quasi-static approximation which is applicable to elements much smaller than the electromagnetic wavelength. At the frequency range of interest in this work (<25 GHz) the element thickness remains smaller than the skin depth in the Al metal (>500 nm), leading to an almost frequency-independent current distribution.

high-resistivity substrate, identical CPW lines were fabricated on a HRS wafer (1-5 k Ω ·cm) with surface passivation [15]. The surface passivation of the HRS wafer was carried out by Ar implantation to amorphorise the surface of the HRS substrate. As a result, the detrimental effects of conductive surface channels can be minimized [15]. A 0.5 μ m PECVD silicon oxide was used as the isolation layer between the CPW conductors and the substrate. The line parameters (characteristic impedance, attenuation) were extracted using the formula in Appendix E from the S-parameter measurements carried out on a HP8510 network analyzer.

The total attenuation, substrate loss, real part of characteristic impedance and phase velocity of a CPW line with a signal line width of 30 μ m, and signal to ground spacing of 10 μ m are shown in Fig.4.15 up to 25 GHz. Compared to an unshielded line on standard Si (U-CPW in Fig.4.15), significant reduction of total attenuation is observed with the ADL, both with (μ ADL-CPW) and without (ADL-CPW) micro-patterning, as shown in Fig.4.15(a). Fig.4.15(b) presents the substrate loss of the CPW line and shows more than a ten-fold reduction within most of the measurement frequency range, due to the use of the ADL. At relatively low frequencies (<5 GHz), ADL-CPW and μ ADL-CPW exhibit



Figure 4.15: Measured (a) total attenuation, (b) substrate loss, (c) characteristic impedance and (d) phase velocity as a function of frequency for an unshielded CPW (U-CPW), a CPW with a conventional ADL shield (ADL-CPW), and a CPW with a micropatterned shield (μ ADL-CPW), all built on a 5-10 Ω ·cm Si substrate. Results for an identical line on a HRS wafer are shown for comparison (HRS-CPW). The line has a signal width *w* of 30 μ m and a signal to ground space *s* of 10 μ m.

similar attenuation values. However, as the frequency increases, the μ ADL-CPW shows remarkably lower attenuation, despite the fact that the substrate losses are almost identical as clearly illustrated in Fig.4.15(b). This deviation is due to the suppression of the eddy current loss of the ADL by micro-patterning, which leads to the reduction of the conductor loss of the line. Fig.4.15(a) also shows the attenuation of the HRS-based CPW (HRS-CPW) which is only slightly lower than that of μ ADL-CPW (2.7- vs. 3.6 db/cm at

25 GHz) despite the latter being built on a standard Si substrate. Note that the characteristic impedances of the shielded and HRS-based lines were all close to ~46 Ω , as shown in Fig.4.15(c) so that a comparison of attenuation values is justified. The conductor loss of the μ ADL-CPW was found to be reduced by 40% at 25 GHz compared to the ADL-CPW which proves the efficiency of micro-patterning in reducing the eddy current loss of the ADL. This, in combination with the high shielding efficiency of the ADL, causes the overall loss of μ ADL-CPW to be low and comparable to the HRS-based device.

It is also noticed that the phase propagation velocity of the shielded lines (not shown here) is almost constant over the whole frequency range. This is in contrast with the dispersive behavior of the unshielded line which is caused by the losses in the standard Si substrate and leads to a rising phase velocity as function of frequency. Compared to the unshielded line, the shielded lines exhibit a phase velocity which is slightly lower (up to 15%) because of the increase in the effective dielectric constant of the environment brought about by the insertion of the ADL.

4.4.3 CPW with multi-layered ADL shields

As mentioned in section 4.4.1, the shielding effect can be enhanced by stacking several ADL layers on top of each other. Fig.4.16 shows schematically the cross section view of the multi-layered ADL. On one hand, the stacking of ADL increases the total thickness of the ADL media. And on the other hand, it also increases the effective in-plane dielectric constant as we shall discuss below.



Figure 4.16: Cross section view of multi-layered ADL.

The thickness dependence of the shield efficiency can be understood by considering Eq. (4.6-4.10). By increasing the thickness t of the ADL, both A(k) and B(k) increase, resulting in a decrease of $\tilde{\varphi}_{si}(k, y)$ and therefore the electric field $(\tilde{E}_x, \tilde{E}_y)$ in Si. For a

95

multi-layer ADL, the effective in-plane dielectric constant can be estimated by considering a 3D lumped-element equivalent circuit, which is similar to the one shown in Fig.4.4 (In Fig. 4.6, a 2D lumped-element network is used to model single ADL with two metal layers and one ordinary dielectric in between. While for a equivalent lumped-element circuit model of multi-layer ADL, the 2D network is stacked on top of each other to form a 3D equivalent circuit.) The effective in-plane dielectric constant of a multi-layer ADL is then given by

$$\varepsilon_{\parallel} = \frac{NC}{t} = \frac{N\varepsilon_d A}{t_d [(N+1)t_M + Nt_d]} \quad , \tag{4.13}$$

where N is the total number of ordinary dielectric layers and t is the overall ADL thickness, ε_d and t_d denote the dielectric constant and thickness of the isolating layers, A is the overlap area between metallic patterns, and t_M is the thickness of the metallic layers. In the above equation we have used $C = \varepsilon_d A / t_d$. If the number of isolating layers is 1, i.e. the ADL structure we have considered in the previous sections, one finds

$$\varepsilon_{\parallel} = \frac{C}{t} = \frac{\varepsilon_d A}{t_d (2t_M + t_d)}, \qquad (4.14)$$

which is exactly Eq. (4.3), based on the fact that the total ADL consist 2 metal layers and one dielectric layer. By dividing both the numerator and denominator on the right side of Eq. (4.13), one has

$$\varepsilon_{\parallel} = \frac{\varepsilon_d A}{t_d \left(\frac{N+1}{N} t_M + t_d\right)}.$$
(4.15)

By comparing Eq. (4.14) and Eq. (4.15), it is obvious that by increasing N, or equivalently the number of ADL layers, one has larger value of the in-plane effective dielectric constant. Substituting the values of t_M and t_d in our experiments, it is found that a ~28% percent increase of ε_{\parallel} can be achieved by adding only one metal and one dielectric layers (N=2 in this case) on single ADL.

The fabrication of CPW lines with multi-layered ADL is straight-forward and similar to the processing presented in the previous sections, except for repeating the patterned metal and isolating dielectric layer certain times. CPW lines with single ADL shield



Figure 4.17: Extracted total attenuation (left column) and substrate loss (right column) for CPW lines ($w=30 \text{ }\mu\text{m}$, $s=30 \text{ }\mu\text{m}$) with different number of ADL shield layers. The ADL metal pattern sizes are 3x3 (a-b), 5x5 (c-d) and 7x7 (e-f) μm^2 .

(SADL-CPW, N=1), double ADL shield (DADL-CPW, N=2) and triple ADL shield (TADL-CPW, N=3) were fabricated on the same wafer. For comparison, unshielded lines (U-CPW) were made as well. The ADL metal square sizes were chosen to be 3 x 3, 5 x 5, and 7 x 7 μ m². Still transmission line parameters were extracted from S-parameter measurement on a HP8510 network analyzer.

Fig. 4.17 compares total attenuation and substrate losses of CPW lines with different ADL pattern sizes and different number of ADL layers. In most of the cases, the total attenuation decreases with increasing the number of ADL layers, which is attributed to the decreases of substrate losses. Apparently, by stacking more ADL layers, the shield is more effective, as expected, due to the higher in-plane effective dielectric constant and higher thickness of the ADL media. However, at high frequencies, as shown in the figure, for the case of $7x7 \mu m^2$, the total loss increases slightly with the number of ADL layers. This is mainly due to the contribution of the eddy current losses in ADL patterns. As can be understood by the aid of Fig. 4.16, the number of metal blocks per unit volume increases with ADL layer numbers, resulting in more eddy current losses per unit volume. This effect becomes more prominent when the pattern size of ADL is large, as in the case of $7x7 \text{ um}^2$, and can be ignored for the pattern sizes of 3x3 and $5x5 \text{ um}^2$. Therefore, by applying multi-layered ADL shields with relatively small pattern sizes, the shield efficiency can be increased at high frequencies while keeping the eddy current loss insignificantly. This is demonstrated by comparing Fig. 4.17 (c) and (e), CPW line with $5x5 \text{ }\mu\text{m}2$ TADL shield shows less loss at 25GHz than all CPW lines with $7x7 \text{ }\mu\text{m}^2$ ADL shield.

4.5 Applications of shielded CPW lines

Transmission lines are building blocks of a wide range of microwave components including branch-line couplers, quarter wavelength transformers and transmission line filters [1]. With the ADL shield developed, the losses of these devices can be, in principle, significantly reduced. In this section, we focus on branch-line couplers and ring hybrids built from shielded CPW lines to demonstrate the loss reduction effects.

4.5.1 Principles of directional couplers

Directional couplers are passive reciprocal four-port networks [1]. The operation of a directional coupler can be illustrated with the aid of Fig. 4.18. As shown in the figure, among the four ports, one (port 1) is regarded as the input, one (port 2) is regarded as the

through port (where most of the incident signal exits), one (port 3) is regarded as the coupled port (where a fixed fraction of the input signal appears), and an isolated port (port 4), which is usually terminated. All four ports are ideally matched, and the circuit is ideally lossless. Directional couplers can be realized in transmission line (microstrip, stripline, coax and CPW) and waveguide forms. They are used for sampling a signal, sometimes both the incident and reflected waves which is an important part of a network analyzer [1].



Figure 4.18: Schematic view of directional couplers and power flow conventions.

It can be shown that, for an ideal directional coupler, its S-parameter matrix takes the form

$$[s] = \begin{bmatrix} 0 & S_{12} & S_{13} & 0 \\ S_{12} & 0 & 0 & S_{24} \\ S_{13} & 0 & 0 & S_{34} \\ 0 & S_{24} & S_{34} & 0 \end{bmatrix}.$$
 (4.16)

In real case, since the directional couplers are required to operate over a certain frequency range, it is not possible to obtain an ideal performance over the whole frequency range. The following S-parameter based quantities are commonly used to characterize a real directional coupler [1]:

Coupling =
$$C = 10 \log \frac{P_1}{P_3} = -20 \log |S_{13}| dB$$
, (4.17)

Directivity =
$$D = 10 \log \frac{P_3}{P_4} = 20 \log \frac{|S_{13}|}{|S_{14}|} dB$$
, (4.18)

Isolation =
$$I = 10 \log \frac{P_1}{P_4} = -20 \log |S_{14}| dB$$
, (4.19)

Insertion loss =
$$IL = 10 \log \frac{P_1}{P_2} = -20 \log |S_{12}| dB$$
. (4.20)

Hybrid couplers are special cases of directional couplers that are designed for a 3dB coupling. Hybrids come in two types, 90° or quadrature hybrids and 180° hybrids. The two kinds of hybrids are simulated and experimentally realized by CPW lines with ADL shields to demonstrate the loss reduction effects and will be the contents of the following two sections.

4.5.2 Branch-line couplers

90° or quadrature hybrids are hybrid couplers with a 90° phase difference in the outputs of the though and coupled ports. They are often made in microstrip or stripline forms as shown in Fig. 4.19, and are also known as branch-line couplers. The operation of an ideal branch-line coupler is as follows. With all ports matched, power entering port 1 equally distributes to port 2 and 3, with no power coupled to port 4. Thus the S-parameter matrix has the form of

$$[s] = \frac{-1}{\sqrt{2}} \begin{bmatrix} 0 & j & 1 & 0 \\ j & 0 & 0 & 1 \\ 1 & 0 & 0 & j \\ 0 & 1 & j & 0 \end{bmatrix}.$$
 (4.21)

The frequency response of an ideal branch-line coupler with central operation frequency of 20 GHz is shown in Fig. 4.20. Here only the magnitudes of S-parameters (in dB) are shown. Without losses, at design frequency (20 GHz), powers inputted at port 1 are equally (3 dB) split between port 2 and port 3, as can be read from the |S12| and |S13| values at 20 GHz in the figure. Also perfect isolation ($|S14| = -\infty dB$) and return loss ($|S11| = -\infty dB$) are achieved at design frequency. All of these quantities, however, degrade quickly as the frequency departs from the central design frequency.

In reality, however, especially for branch-line couplers built on Si, losses are inevitable which results in significant deviation from the ideal behavior. To see the effect of substrate loss, branch-line couplers built from Si-based CPW lines were simulated in



Figure 4.19: Geometry of a branch-line coupler in microstrip form. Power entered at port 1 are equally distributes to port 2 and 3, with no power coupled to port 4.



Figure 4.20: Magnitudes of *s*-parameters (in dB) of an ideal branch-line coupler as a function of frequency.

ADS both with and without the high-k ADL shield. The design frequency was 20 GHz. The system characteristic impedances (characteristic impedance of the transmission lines)
in both cases were 50 Ω . Due to modifications of characteristic impedance and phase speed by application of ADL shields, the geometric layouts are slightly different in the two designs. The layout of the branch-line coupler in its CPW form is schematically shown in Fig. 4.21. As illustrated in the cross sectional view, an additional metal layer of 0.6µm in thickness, which is 3µm away from but connected to the top CPW metal layer by metallic vias, is implemented to prevent floating of the central, rectangular-shaped ground conductor. In the S-parameter simulation, the resistivity of the Si substrate is supposed to be 10 Ω cm.



Figure 4.21: Schematic top view (a) and cross section view (b) of directional coupler with CPW configuration.

Fig. 4.22 compares the simulated S-parameters of the branch-line coupler with (solid markers) and without (hollow markers) the ADL shield. Without the ADL shield, the coupler shows an insertion loss of 6.3 dB and coupling of 7.4 dB respectively at 20 GHz. Compared with the ideal coupler behavior, 3.3 dB and 4.4 dB of loss is generated during the power transfer from input port (port 1) to through port (port 2) and coupled port (port 3), which is caused by the substrate loss and metal loss of the CPW lines. With the ADL shield in place, an insertion loss of 4.3 dB and a coupling of 4.4 dB at 20 GHz can be achieved. The 2 dB loss reduction at the through port and 3 dB loss reduction at the



Figure 4.22: Simulated magnitudes of *s*-parameters (in dB) of CPW branch-line couplers on Si with (solid markers) and without (hollow markers) ADL shields.

coupled port are due to the shielding effect of the ADL.

Based on the simulation results, CPW based branch-line couplers with and without ADL shields were designed and fabricated on ordinary Si wafer. The fabrication processes are similar to the ones described in section 4.4.1, except that the 9 μ m PECVD oxide layer was deposited in two steps. After the first 6 μ m oxide deposition, a metal layer of 0.6 μ m in thickness was grown and patterned to form the underneath bridge as shown in Fig. 4.21(b). The other 3 μ m oxide was then formed and vias were opened before the last metallization step to form the CPWs. Due to the limitation of our measurement facilities, 2-port S-parameter measurement could be performed with the other two ports open. However, it can be shown that, under this circumstance, with z-parameters of each pair of two ports, the four port z-parameter matrix can be easily constructed by simply combining the two port z-parameters. The four port z-parameter matrix was then converted back to S-parameter matrix to extract the characteristic quantities of the branch-line couplers.



Figure 4.23: Measured magnitudes of *s*-parameters (in dB) of CPW branch-line couplers on Si with (solid markers) and without (hollow markers) ADL shields.

Fig. 4.23 shows the extracted magnitudes of S-parameters from measurements. Due to the uncertainty in the substrate resistivity (nominal value: 5-10 Ω cm, 10 Ω cm was assumed in simulation), deviation of characteristic impedances as well as phase velocities from simulations can be expected. Therefore, in the experiments, the minimum values of S11 and S14 were achieved at different frequencies. Nevertheless, the loss reduction by applying ADL shields is clearly demonstrated if one looks at S12 and S13 values in a certain bandwidth near the central design frequency.

4.5.3 Ring hybrids

 180° or quadrature hybrids are hybrid couplers with a 180° phase difference in the two output ports. It can also be operated so that the outputs are in phase. Ideally, referred to Fig. 4.18, a signal applied to port 1 is equally split into two in-phase components at ports 2 and 3. While input at port 4 will be evenly split into two components at ports 2 and 3 with 180° phase differences. 180° hybrids are often used as a combiner. With input signals at ports 2 and 3, the sum of the inputs will be formed at port 1, while the difference will be formed at port 4. Therefore, ports 1 and 4 are also referred to as the sum and difference ports. The scattering matrix for an ideal 180° hybrid takes the following form:



Figure 4.24: Geometry of a ring hybrid in microstrip form. Power entered at port 1 are equally distributes to port 2 and 3, with no power coupled to port 4.

$$[s] = \frac{-j}{\sqrt{2}} \begin{bmatrix} 0 & 1 & 1 & 0\\ 1 & 0 & 0 & -1\\ 1 & 0 & 0 & 1\\ 0 & -1 & 1 & 0 \end{bmatrix}.$$
 (4.22)

The 180° hybrids can be realized in different forms. In silicon technologies, the ring hybrid, also known as rat-race, is usually the choice of fabrication due to its planar structure. A ring hybrid in microstrip form is schematically shown in Fig. 4.24 and the frequency response of an ideal ring hybrid is shown in Fig. 4.25. At design frequency (20 GHz), power inputted in port 1 is evenly (3 dB) split into port 2 and 3 without any power returned back. Also port 4 is perfectly isolated. It is also shown in Fig. 4.25 that S-parameters are symmetric about the design frequency.

Ring hybrids using CPW lines were designed and simulated in ADS to demonstrate the loss reduction effect of ADL shields. The layout is schematically shown in Fig. 4.26, where, similar to the branch-line coupler, additional metal bridge and through dielectric vias are implemented. Fig. 4.27 compares the magnitudes of simulated S-parameters of ring hybrids with (solid markers) and without (hollow markers) ADL shields. In the unshielded case, the insertion loss and coupling are found to be 6.7dB and 6.1dB at 20 GHz, respectively. Also notice the obvious asymmetry of all S-parameters about the



Figure 4.25: Magnitudes of *s*-parameters (in dB) of an ideal ring hybrid as a function of frequency.



Figure 4.26: Schematic top view of the layout of a ring hybrid in CPW form. Additional metal bridge under the CPW layer is added and connected to CPW grounds by through dielectric vias.

central design frequency, which is due to high substrate loss at high frequencies. When ADL shields are inserted between the Si substrate and the CPW lines, the ring hybrid shows a 4.0dB insertion loss and 3.8dB coupling, which are 2.7dB and 2.3dB lower than the unshielded case. Besides, the shielded ring hybrid shows a more symmetric behavior about the design frequency.



Figure 4.27: Simulated magnitudes of *s*-parameters (in dB) of CPW ring hybrids on Si with (solid markers) and without (hollow markers) ADL shields as a function of frequency.

4.6 Application of ADL shields to spiral inductors

On-chip spiral inductors are nowadays viewed as key components in monolithic RF/microwave circuits. But, as the same time, they are one of the main performance limiting components of integrated RF circuits. The quality factor (Q-factor) of spiral inductors is limited by the conductor loss in the spiral coils as well as substrate loss primarily caused by the unwanted currents flowing in the conductive Si substrate [16]. Similar to the integrated transmission lines, efforts have been made to reduce the substrate loss by using electric shields [17]. In this section, we investigate the application of the high-k ADL shields for the reduction of the substrate loss of integrated spiral coils.

4.6.1 Principles of operation

Substrate loss reduction of spiral inductors by applying ADL shields can be understood by considering the simplified equivalent lumped element model of Fig. 4.28 [16, 18-19]. which also shows the geometry (cross sectional view) of the inductors with and without the ADL shields. The spiral coil itself is represented by the inductance L_s and series resistance R_s which accounts for the conductor loss. C_{ox} represents the oxide capacitance between spiral coil and the substrate. The Si substrate is modeled by C_{si} and R_{si} with the latter accounting for the losses in the substrate. For simplicity, here we ignore the inter-wire capacitance. It should be pointed out that due to its inherent asymmetry of spiral inductor layout, the shunt elements in the π -type model are not necessarily symmetric (i.e. for example, C_{ax1} is not equal to C_{ax2}). The quality factor (Q) of the spiral inductor is defined as the ratio between the imaginary part and real part of input impedance at port 1 when port 2 is grounded [15]. When the ADL layer is inserted between the spiral coil and the substrate, extra shunt capacitances represented by C_{ADL} in Fig. 4.28(d) is added to the lumped element model. This provides a low impedance route for RF current to flow through, bypassing the substrate resistance. However, a side effect of this extra capacitance is that it will also decrease the self resonance frequency of the inductor. Therefore, a tradeoff has to be made to get sufficient shield effect while keeping the self resonance frequency far enough from the working frequency.

4.6.2 Results and discussions

To test the effectiveness of ADL shields for spiral inductors, simulations were carried out in ADS. First, a conventional spiral inductor built on a SiO₂/Si substrate was simulated and the values of the lumped elements in Fig. 4.28(c) were extracted from the simulated S-parameters. Different values of C_{ADL} in a certain range (which can be estimated by the geometry of the spiral coil and the in-plane dielectric constant of ADL) were then assigned in Fig.4.28(d). The quality factor (Q) as a function of frequency was then calculated from the S-parameter simulations with and without C_{ADL} .

Fig.4.29 shows the simulated quality factor Q as a function of frequency for different values of C_{ADL} . Without the shield, the maximum value of quality factor (Q_{max}) is 13.6 at 1.9 GHz. When C_{ADL} is 0.1 pF, a Q_{max} of 15.5 (14% increase), is obtained at 2.4 GHz. With C_{ADL} of 0.5 pF, we have a Q_{max} of 23.9 at 3.9 GHz, an almost 76% improvement. It is noticed that the frequency at which Q_{max} is reached increases with increasing C_{ADL} . It can also be observed from Fig.4.29 that the self resonance frequency of the inductor (the frequency at which Q=0) decreases with increasing ADL capacitance. This is because the







Figure 4.28: Cross section views of spiral inductors (a) on Si substrate without ADL shields and (b) with ADL shields. Also shown simplified equivalent lumped element models of spiral inductors (c) on Si substrate without ADL shields and (d) with ADL shields.

total shunt capacitance in the π -type equivalent model of Fig.4.28 increases as C_{ADL} increases.



Figure 4.29: Simulated quality factor of a 1.5nH spiral inductor as a function of frequency with different values of C_{ADL} .

For experimental verification, spiral inductors with square shape coils and equipped with ADL shields were fabricated on a 5-10 Ω cm Si substrate. The metallic element size of the ADL shield was chosen to be 15x15 μ m² to achieve a high in-plane dielectric constant . Due to the strong magnetic field generated by the spiral, which is expected to induce relatively large eddy currents in the ADL patterns, spiral inductors with ordinary ADL (ADL-Ind) and micro-patterned ADL (μ ADL-Ind) were fabricated for comparison (refer to Fig. 4.14). The ones without any ADL shield (U-Ind) were made as well on the same wafer.

Fig. 4.30 shows the extracted quality factor and inductance values for a ~ 2 nH inductor. Compared with unshielded coil, the shielded devices show an improvement of the *Q*-factor in a wide frequency range. In particular, a 17% increase of the Q_{max} is achieved by the use of micro-patterned ADL shields. This can be understood by considering the equivalent lumped element model as shown in Fig. 4.28. The quality factor *Q* of an inductor without ADL shields can be expressed as



Figure 4.30: Measured quality factor and inductance value of a 2nH spiral inductor as a function of frequency with ordinary (ADL-Ind) and micro-patterned (μ ADL-Ind) ADL shields. Results for un-shielded spiral inductor (U-Ind) are shown as well for comparison.

$$\frac{1}{Q} = \frac{1}{Q_s} + \frac{1}{Q_c}, \qquad (4.23)$$

where $Q_s = \omega L_s / R_s$ and

$$Q_{c} = \frac{1 + \omega^{2} R_{si}^{2} C_{si} (C_{ox} + C_{si})}{\omega R_{si} C_{ox}}.$$
 (4.24)

Since we ground port 2 when calculating the quality factor, the resistance and capacitance values are taken from the left shunt branch in Fig. 4.28(b). At relatively low frequencies (roughly speaking, below 1 GHz), the overall quality factor is limited by the conductor loss which is represented by Q_s and has a small value. Therefore, in this frequency range, inclusion of the ADL shield does not strongly affect Q. As frequency ramps up, Q_s increases and becomes comparable to Q_c so that the influence of Q_c on the overall Q becomes more predominant. The effect of the ADL shields can be modeled as an extra

capacitance C_{ADL} added to C_{si} in Eq. 4.24. It is apparent that this will lead to an increment of Q_c and therefore an improvement of Q. It is also noticed in Fig. 4.30 that, compared with the original ADL shield, the micro-patterned one has slightly higher Q over most of the frequency range. This again can be understood from the equivalent circuit model shown in Fig. 4.28. The ohmic loss due to eddy current flow inside the ADL metallic patterns can be lumped into the series resistance R_s in Fig.4.28(c). By cutting slots in the metallic element, eddy current flow is effectively suppressed, leading to a reduction of R_s as compared to ordinary ADL shields. Therefore, the quality factor Q_s of the series branch is improved, so as to the overall quality factor Q of the inductor. The frequency at which Q_{max} is achieved shifts to higher values when ADL shields are applied, which is in agreement with our simulation results. It is also clear from the figure that the self-resonance frequencies of the spiral inductors are reduced by applying ADL shields as expected.

4.7 Conclusions

In this chapter, we presented and investigated a novel method for reducing the substrate loss of Si-based integrated transmission lines. The method exploits the shielding effect of an ADL with a high in-plane dielectric constant. The ADL comprises two patterned metal layers separated by a thin conventional dielectric, and can be readily built in a standard IC process. It was found that the ADL shield significantly lowers the attenuation, but only moderately changes the characteristic impedance of the transmission lines. The shield effectiveness of ADL was experimentally shown to be improved by increasing the metallic element size, while micro-patterning the ADL to suppress the eddy current loss. By this way, CPW built on ordinary conductive silicon showed comparable total attenuation as the one fabricated on HRS substrate. We also showed that the shield effectiveness can be improved by stacking a few ADL layers. Loss reduction of CPW based microwave devices, i.e. branch-line coupler and ring hybrid, by applying ADL shields were demonstrated by simulation and experiment. Application of ADL shields to spiral inductors was also analyzed with the help of a simplified lumped element model and was experimentally demonstrated.

References

- [1] D. Pozar, "Microwave Engineering", Wiley, New York, 2005.
- [2] I.D. Robertson, S. Lucyszyn, "RFIC and MMIC Design and Technology", Antony Rowe Ltd, Chippenham, Wiltshire, 2001.
- [3] J.N. Burghartz, D.C. Edelstein, K.A. Jenkins, Y.H. Kwart, "Spiral inductors and transmission lines in silicon technology using copper-damascene interconnects and low-loss substrate", IEEE Trans. Microwave Theory Tech., Vol.45, No.10, pp. 1961-1968, 1997.
- [4] V. Milonovic, M. Gaitan, E.D. Bowen, M.E. Zaghloul, "Micromachined microwave transmission lines in CMOS technology", IEEE Trans. Microwave Theory Tech., Vol. 45, No. 5, pp. 630-635, 1997.
- [5] R. Dekker, P.G.M. Baltus, H.G.R. Maas, "Substrate transfer for RF technology", IEEE Trans. Electron Devices, Vol. 50, No. 3, pp. 747-757, 2003.
- [6] C.P. Yue, S.S. Wong, "On-chip spiral inductors with patterned ground shields for Si-based RF ICs", IEEE J. Solid-State Circuits, Vol. 33, No. 5, pp. 743-752, 1998.
- [7] R. Lowther, S.G. Lee, "On-chip interconnect lines with patterned ground shields", IEEE Microwave and Guided Wave Lett., Vol. 10, No. 2, pp. 49-51, 2000.
- [8] T.S.D. Cheung, J.R. Long, "Shielded passive devices for silicon-based monolithic microwave and millimeter-wave integrated circuits", IEEE J. Solid-State Circuits, Vol. 41, No. 5, pp. 1183-1200, 2006.
- [9] Y. Ma, B. Rejaei, Y. Zhuang, "Artificial dielectric shields for integrated transmission lines", IEEE Microwave Wireless Compon. Lett., Vol. 18, No. 7, pp. 431-433, 2008.
- [10]Y. Ma, B. Rejaei, Y. Zhuang, "Low-loss on-chip transmission lines with micro-patterned artificial dielectric shields", Electronic Lett., Vol. 44, No. 15, pp. 913-915, 2008.
- [11] M.A.T. Sanduleanu, G. Zhang, J.R. Long, "31-34GHz low noise amplifier with on-chip microstrip lines and inter-stage matching in 90-nm baseline CMOS", Proceedings of RFIC, 2006, pp. 143-146.
- [12] J.C. Peuzin, J.C. Gay, "Demonstration of the waveguiding properties of an artificial surface reactance", IEEE Trans. Microwave Theory Tech., Vol. 42, No. 9, pp.

1695-1699, 1994.

- [13] I. Awai, H. Kubo, T. Iribe, D. Wakamiya, A. Sanada, "An artificial dielectric material of huge permittivity with novel anisotropy and its application to a microwave BPF", in Int. Microwave Symp. Dig. 2003, IEEE MTT-S, Vol. 2, pp. 1085-1088.
- [14] R. Marques, M. Horno, "Propagation of quasi-static modes in anisotropic transmission lines: application to MIC lines", IEEE Trans. Microwave Theory Tech., Vol. 33, No. 9, pp. 927-932, 1985.
- [15] B. Rong, J.N. Burghartz, L.K. Nanver, B. Rejaei, M. van der Zwan, "Surfacepassivated high-resistivity silicon substrates for RFICs", IEEE Electron Device Lett., Vol. 25, No. 4, pp. 176-178, 2004.
- [16] J.N. Burghartz, B. Rejaei, "On the design of RF spiral inductors on silicon", IEEE. Trans. Electron Devices, Vol. 50, No. 3, pp. 718-729, 2003.
- [17] C.P. Yue, S.S. Wong, "On-chip spiral inductors with patterned ground shields for Si-based RF IC's", IEEE J. of Solid State Circuits, Vol. 33, No. 5, pp.743-752, 1998.
- [18] C.P. Yue, C. Ryu, J. Lau, T.H. Lee, S.S. Wong, "A physical model for planar spiral inductors on silicon", Electron Devices Meeting, 1996, International (IEDM), pp. 155-158, 1996.
- [19] J.R. Long, M.A. Copeland, "The modeling, characterization, and design of monolithic inductors for silicon RF IC's", IEEE J. Solid-State Circuits, Vol. 32, No. 3, 1997.

Chapter 5

On-chip Artificial Dielectric Resonators for Microwave Applications

115

5.1 Introduction

Microwave resonators are important components in microwave circuits, and are used in a variety of applications, including filters, oscillators, frequency meters and tuned amplifiers [1]. From a physical point of view, energy is stored inside a resonator through electromagnetic fields. Resonance occurs at certain frequencies when the time average stored electric and magnetic energies are equal. The input impedance of a resonator at resonance is purely real since the reactance is proportional to the difference between electric and magnetic energy storage.

The quality factor (Q-factor or simply Q) is an important figure of merit to characterize a resonance circuit, and is defined as

$$Q = \omega \frac{\text{average energy stored}}{\text{power loss}},$$

$$= \omega \frac{W_e + W_m}{P_i},$$
(5.1)

where ω is the angular frequency, W_e and W_m denote average electric and magnetic energy stored in the resonance circuit respectively, and P_i is power dissipated in the resonance circuit itself[1]. To a very good approximation, the quality factor can also be expressed as

$$Q = \frac{f_0}{\Delta f} = \frac{f_0}{|f_1 - f_2|},$$
(5.2)

where f_0 is the resonance frequency, f_1 and f_2 are half power frequencies.

In order to be applied in a microwave circuit, a microwave resonator must deliver power to an external load. The power loss due to the presence of an external load results in the external quality factor Q_e , which is defined as

$$Q_e = \omega \frac{W}{P_e}, \qquad (5.3)$$

where *W* is the total electric and magnetic energy stored in the resonance circuit, and P_e is the power delivered to the external load. We can also define the loaded quality factor, Q_L , which is the total *Q* of the system including power losses both internal and external to the resonator. Since the total power loss is simply the sum of the internal power loss of P_i and the external power loss P_e , one obtains

$$\frac{1}{Q_{\rm L}} = \frac{1}{Q_{\rm e}} + \frac{1}{Q_{\rm 0}},\tag{5.4}$$

where Q_0 is the quality factor of the resonator itself as defined in Eq. (5.1), also known as unloaded Q.

Microwave resonators can be realized in different forms, i.e. lumped element resonators (*RLC* resonant circuit), transmission line resonators, cavity resonators (also known as waveguide resonators) and dielectric resonators. The lumped element resonators are simply the parallel or series *RLC* resonant circuits where lumped inductor and capacitor are parallel or series connected to form the resonant circuit. This realization is normally simple and gives a compact circuit. In microwave integrated circuits the quality factor of the resonator is limited by the lumped inductor, especially at high frequencies [2]. Therefore, lumped element resonators are normally realized on low loss substrates [3] and/or at relatively low frequencies [4], where substrate loss is small [2]. Even in that case, however, the quality factor of the integrated inductor (and, thus, the resonator) can hardly exceed a value of 50 due to conductor loss which is further enhanced by the skin effect at high microwave frequencies. An alternative to lumped-element resonator is provided by distributed element resonators which are usually based on transmission lines. From transmission line theory, it follows that a short-or open-circuited transmission line can resonate at frequencies where its electric length becomes a multiple of a quarter of the propagation wavelength on the line. The unloaded quality factor for such a resonator is given by

$$Q_0 = \frac{\beta}{2\alpha}, \qquad (5.5)$$

where β is the propagation constant and α is the attenuation constant of the line [1]. For MMIC application on Si, on-chip transmission lines can be realized by a microstrip or coplanar waveguide (CPW). However, the quality factor of these lines is limited by substrate loss in conductive Si and metal loss mainly in the signal line, especially at high frequencies (tens of GHz) As shown in Cheung's work, the quality factor of a 50 Ω microstrip line on a 10 Ω ·cm Si substrate is below 15 at 30 GHz. And even on a semi-insulated alumina substrate, a CPW line of 50 Ω shows a quality factor not exceeding 40 below 40 GHz [6].

In term of quality factor, nothing can surpass the cavity resonator, which is constructed by using a section of a rectangular or cylindrical waveguide. Because of radiation loss of an open-ended waveguide, waveguide resonators are usually short circuited at both ends, forming a closed, hollow box which stores electric and magnetic energies. Although power is dissipated in the metallic walls of the cavity, very high *Q*-factors (tens of thousands) can be obtained [1]. However, although cavity resonators are extensively used in discrete or a hybrid microwave system, they are difficult to integrate with the standard IC technology. This is because the lateral dimensions of a cavity resonator should at least equal have the electromagnetic wavelength in air which is quite large compared to chip dimensions. Besides, their fabrication requires substrate micromachining techniques [7,8] which significantly complicates the process integration and therefore increases the fabrication cost.

Whereas in a cavity resonator the metallic walls are responsible for the field confinement, a dielectric resonator employs the high (relative) dielectric constant of a small disc or cube of low loss dielectric material. The high dielectric constant of the resonator material ensures that most of the electromagnetic field is contained within the resonator, with little fringing or leakage from the sides and ends of the dielectric resonator. Due to the high dielectric constant, the size of dielectric resonator is considerably smaller than that of an empty resonant cavity operating at the same frequency. Conductor losses are absent in a dielectric resonators are usually sealed on solid state devices to form certain microwave systems such as wireless front-end transceivers. Because of their high Q factor, dielectric resonators are needed in such system to have low insertion loss, high out of band rejection, and high channel to channel isolation. However, compared with the solid state devices, dielectric resonators are much more expensive and area consuming. Therefore, for MMIC applications, it will be attractive if one could integrate dielectric resonator on chip and shrink the size significantly, while still keep the high Q factor.

The size of the dielectric resonator is approximately proportional to the wavelength at resonance frequency, which is inversely proportional to the square root of dielectric constant. Obviously, to make the dielectric resonator compact, dielectric materials with very high dielectric constant are needed. However, even for ferroelectric materials, in their thin film form, a dielectric constant of 1000 is hardly achieved. Besides, as we remarked in the previous chapter, there is no guarantee that the dielectric loss of these materials remains low at microwave frequencies.

Could we then use artificial dielectric constructions, as we did in the previous section, to emulate high-k dielectrics? In this chapter, we will try to answer this question by investigating a new type of on-chip resonator, based on artificial dielectric layer (ADL) disk. The resonator may be excited by a conventional metallic structure, i.e., a metal ring surrounding the ADL disk. To better understand the operation of the device, a physical model based on Maxwell's equations will be deduced, in which we model the ADL as an effective material, i.e. a continuous anisotropic media with effective permittivity and permeability tensors. It will be shown that every resonance mode of the ADL disk can be modeled as a normal parallel *RLC* resonator and coupled to the self inductance of the metal ring through a mutual inductance. We will then compare the results obtained with the full-wave numerical calculations carried out using HFSS (High Frequency Simulation



Figure 5.1: schematic of the ADL disk resonator. A conductive, current-carrying ring encloses a thin disk of artificial dielectric with the radius *R*. The ring and the disk are placed on top of a substrate which is assumed to be unbounded in the horizontal plane. The current flowing through the ring is considered as an impressed source, inducing an

electromagnetic field with the components E_{θ}, H_r, H_z in the system.

System), as well as experimental data from ADL disk resonators fabricated on a low-loss substrate (AF 45 glass wafer).

5.2 Theoretical analysis of artificial dielectric resonator

5.2.1 Configuration and theoretical model

As shown in Fig. 5.1, we assume a cylindrically symmetric configuration comprising a dielectric (ADL) disk of radius R encircled by a metallic ring carrying a current with the density

$$\boldsymbol{J}^{i} = J^{i}_{\theta}(\boldsymbol{r}, \boldsymbol{z})\hat{\boldsymbol{\theta}}$$
(5.6)

with $\hat{\theta}$ the unit vector in the azimuthal direction. We neglect the influence of the ADL disk on J^i , viewing the latter as an impressed source. The ring and the disk are assumed to be placed above a laterally unbounded, stratified substrate.

Inside the ADL disk the distribution of the electric (E) and magnetic (H) fields is

governed by the Maxwell equations

$$\frac{\partial E_{\theta}}{\partial z} = j\omega\mu_0 H_r \tag{5.7}$$

$$\frac{\partial}{r\partial r} \left(rE_{\theta} \right) = -j\omega\mu_{\perp}H_{z} \tag{5.8}$$

$$\frac{\partial H_r}{\partial z} - \frac{\partial H_z}{\partial r} = j\omega\varepsilon_{\parallel}E_{\theta} + J_{\theta}^i, \qquad (5.9)$$

where the indices r, θ, z represent the field components in the radial, azimuthal, and vertical directions. Here we model the ADL disk as an effective material with its effective in-plane dielectric constant ε_{\parallel} , and effective perpendicular permeability μ_{\perp} . The latter differs from the vacuum permeability μ_0 because of the flow of local eddy currents in the individual metallic patterns comprising the ADL. For the moment, we neglect the effect of μ_{\perp} , and assume the latter to be equal to μ_0 .

Instead of directly solving the Maxwell equations inside the ADL, we view the latter as a volume source of electric polarization currents given by

$$J_{\theta}^{P} = j\omega(\varepsilon_{\parallel} - \varepsilon_{0})E_{\theta}.$$
(5.10)

Assume now that $(-j\omega\mu_0 r')G(r, z | r', z')$ is the Green's function describing the azimuthal electric field induced at r, z by an infinitesimally thin ring of current placed above the substrate at r', z' (in the absence of the ADL disk). At any point in space, the total electric field generated by the impressed and polarization currents can be written as

$$E_{\theta}(r,z) = -j\omega\mu_{0} \int_{\Omega_{D}} G(r,z \,|\, r',z') J_{\theta}^{P}(r',z') ds' + E_{\theta}^{i}(r,z)$$
(5.11)

$$E^{i}_{\theta}(r,z) = -j\omega\mu_{0} \int_{\Omega_{i}} G(r,z \mid r',z') J^{i}_{\theta}(r',z') ds', \qquad (5.12)$$

where Ω_D and Ω_i denote the cross sections of the disk and the current source in the *r*-*z* plane (as shown in Fig. 5.2), respectively, and ds' = r'dr'dz'. In the above equation E_{θ}^i is the (external) field generated by the impressed source alone. Note that *G* is frequency dependent, but this is not explicitly shown. By considering the points inside the ADL, and combining Eqs.(5.10-5.11), one obtains the two-dimensional integral



Figure 5.2: Cross section of the system in the r-z plane. The cross section of the ADL disk and the current-carrying ring are denoted by Ω_D and Ω_i , respectively.

equation:

$$E_{\theta}(r,z) = \omega^2 \mu_0 \Delta \varepsilon \int_{\Omega_D} G(r,z \mid r',z') E_{\theta}(r',z') ds' + E_{\theta}^i(r,z), \qquad (5.13)$$

where $\Delta \varepsilon = \varepsilon_{\parallel} - \varepsilon_0$. Solving this equation for a given external field yields the distribution of the electric field inside the disk.

Modal expansion

Consider now the eigenvalue problem defined on Ω_D by the equation

$$\int_{\Omega_{D}} G(r, z \mid r', z') \psi_{k}(r', z') ds' = \lambda_{k} \psi_{k}(r, z) , \qquad (5.14)$$

where ψ_k and λ_k are the (complex) eigenfunctions and eigenvalues of the integral operator G. (They both depend on frequency as does G.) Since G is symmetric (it does not change under the simultaneous replacements $r \leftrightarrow r', z \leftrightarrow z'$), it follows that ψ_k form an orthogonal set which, upon proper normalization of ψ_k , yields

$$\int_{\Omega_D} \psi_k(r,z) \psi_m(r,z) ds = \delta_{km}, \qquad (5.15)$$

where δ_{km} is the Kronecker delta. Furthermore, we assume that the set of the

eigenfunctions ψ_k is complete so that every arbitrary function defined on Ω_D can be expanded as a series in ψ_k . Therefore, we can write:

$$E_{\theta}(r,z) = \sum_{k=0}^{\infty} c_k \psi_k(r,z) .$$
 (5.16)

Upon substituting Eq. (5.16) into Eq. (5.13), and using Eqs. (5.14-15), solving for a_k , and returning to Eq. (5.16) one obtains

$$c_k = \frac{-j\omega\mu_0 A_k}{1 - \lambda_k \omega^2 \mu_0 \Delta \varepsilon},$$
(5.17)

where

$$A_{k} = \int_{\Omega_{D}} \int_{\Omega_{i}} \psi_{k}(r, z) G(r, z \mid r', z') J_{\theta}^{i}(r', z') ds ds' .$$
 (5.18)

From Eq. (5.16), we then have

$$E_{\theta}(r,z) = \left(-j\omega\mu_{0}\right) \sum_{k=1}^{\infty} \frac{A_{k}\psi_{k}(r,z)}{1 - \lambda_{k}\omega^{2}\mu_{0}\Delta\varepsilon} \quad (5.19)$$

5.2.2 Equivalent circuit model

To have a better understanding of the operation of the ADL-disk resonator, it is instructive represent the latter by an equivalent circuit model. This will be achieved by first calculating the impedance of the ring, from which an equivalent circuit model will be developed.

To calculate the impedance Z of the ring, consider the total complex power P delivered to the system by the impressed source, which is given by

$$P = -\frac{1}{2} \int_{\Omega_i} E_\theta \left(J_\theta^i \right)^* ds = \frac{1}{2} Z |I|^2, \qquad (5.20)$$

where I is the total current flowing through the ring. Using Eqs. (5.10-12, 5.20), the impedance of the ring can be expressed as

$$Z = Z^i + Z^D \quad , \tag{5.21}$$

where

$$Z^{i} = \frac{j\omega\mu_{0}}{|I|^{2}} \iint_{\Omega_{i}\Omega_{i}} \left[J_{\theta}^{i}(r,z) \right]^{*} G(r,z \mid r',z') J_{\theta}^{i}(r',z') ds ds'$$
(5.22)

is the self-impedance of the current ring (when no disk is present). The term Z^{D} represents the contribution of the ADL disk to the impedance, and is given by

$$Z^{D} = -\frac{\omega^{2} \mu_{0} \Delta \varepsilon}{|I|^{2}} \int_{\Omega_{i} \Omega_{D}} \left[J_{\theta}^{i}(r,z) \right]^{*} G(r,z \mid r',z') E_{\theta}(r',z') ds ds'.$$
(5.23)

Using the expansion (5.19), Eq. (5.23) can be written as

$$Z^{D} = \frac{\omega^{2} \mu_{0} \Delta \varepsilon}{|I|^{2}} \sum_{k=1}^{\infty} \frac{(j \omega \mu_{0}) A_{k} \overline{A}_{k}}{1 - \lambda_{k} \omega^{2} \mu_{0} \Delta \varepsilon}, \qquad (5.24)$$

$$\overline{A}_{k} = \int_{\Omega_{i}} \int_{\Omega_{D}} \left[J_{\theta}^{i}(r,z) \right]^{*} G(r,z \mid r',z') \psi_{k}(r',z') ds ds' .$$
(5.25)

The above result can be represented by the equivalent lumped-element model of Fig. 5.3 as we discuss below. Consider the current carrying ring as the primary winding of a transformer with an infinite number of independent secondary windings (corresponding to different eigenmodes of the disk) all coupled to the primary winding. Let us define the impedance matrix of the transformer as

$$\boldsymbol{Z}_{tr} = j\omega \begin{bmatrix} L_{p} & M_{1} & M_{2} & M_{3} & \cdots \\ M_{1} & L_{s,1} & 0 & 0 & 0 \\ M_{2} & 0 & L_{s,2} & 0 & 0 \\ M_{3} & 0 & 0 & L_{s,3} & 0 \\ \vdots & 0 & 0 & 0 & \ddots \end{bmatrix},$$
(5.26)

where $L_p = Z^i / j\omega$ is the self inductance of the primary winding, $L_k = \mu_0 \lambda_k / t$ is the self inductance of the *k*-th secondary winding, and $M_k = \mu_0 \left(\overline{A_k A_k} / |I|^2 t\right)^{1/2}$ is the mutual inductance between the latter and the primary winding. (The division by the disk thickness *t* is to ensure the correct physical dimension for the inductances, but is arbitrary otherwise. Any other length scale could have been used instead of *t*.) It can

then be shown that the impedance Z (Eq. 5.21) is identical to the impedance seen at the input of the primary winding with all secondary windings terminated by capacitances all equal to $C = t\Delta\varepsilon$. The conductive ring is thus magnetically coupled to an infinite number of parallel *LC* resonators.

Note that the eigenvalues λ_k and, therefore, the inductances L_k are generally complex and frequency-dependent. The imaginary part of $L_k = L'_k - jL''_k = (\mu_0 / t)(\lambda'_k - j\lambda''_k)$ is related to the polarization loss of the dielectric substrate, as well as radiation into space and surface waves of the substrate. The capacitance *C* is also generally complex due to the polarization loss of the artificial dielectric disk, i.e. $C = C' - jC'' = t [(\varepsilon'_{\parallel} - \varepsilon_0) - j\varepsilon''_{\parallel}]$.



Figure 5.3: Equivalent circuit of an ADL disk resonator. Note that, in general, all the component values are complex and frequency dependent.

Provided that the losses are small (i.e. $L'_k \gg L''_k$, $C' \gg C''$) each eigenmode of the disk will resonate at a frequency $\omega = \omega_k$ satisfying the equation

$$\omega^2 L'_k(\omega)C' = 1. \tag{5.27}$$

The quality factor Q of the resonator is then given by

$$\frac{1}{Q} = \frac{1}{Q_k^L} + \frac{1}{Q^C}, \qquad (5.28)$$

where

$$Q_k^L = \frac{L'_k(\omega_k)}{L''_k(\omega_k)} = \frac{\lambda'_k(\omega_k)}{\lambda''_k(\omega_k)}$$
(5.29)

$$Q^{C} = \frac{C'}{C''} = \frac{\varepsilon_{\parallel}' - \varepsilon_{0}}{\varepsilon_{\parallel}''} \approx \frac{\varepsilon_{\parallel}'}{\varepsilon_{\parallel}''} = (\tan \delta)^{-1}.$$
(5.30)

The approximation in the last equation is justified since $\varepsilon'_{\parallel} \gg \varepsilon_0$, and $\tan \delta$ is the effective loss tangent of the ADL.

Now we would like to include the effect of the magnetic losses by perturbation theory. To that end, consider the perpendicular magnetization M_z defined by

$$M_z = \chi_\perp H_z \,, \tag{5.31}$$

where $\chi_{\perp} = \mu_{\perp} / \mu_0 - 1$ is the perpendicular susceptibility of the ADL. Like the electric polarization, the effect of M_z can be taken account of by an equivalent current which is now given by $J_{\theta}^M = -\frac{\partial M_z}{\partial r}$. The extra electric field generated by this current can once again be represented in terms of the Green's function *G*:

$$E_{\theta}^{M}(r,z) = \gamma \int_{\Omega_{D}} \frac{\partial}{r'\partial r'} \left[r'G(r,z \mid r',z') \right] \frac{\partial}{r'\partial r'} \left[r'E_{\theta}(r',z') \right] ds'$$
(5.32)

$$\gamma = \frac{\chi_{\perp}}{\chi_{\perp} + 1} = 1 - \frac{\mu_0}{\mu_{\perp}}, \qquad (5.33)$$

where we have used partial integration, together with Eqs. (5.8) and (5.31). The total electric field at any point in space is now given by adding the field E_{θ}^{M} to the right hand side of Eq. (5.13):

$$E_{\theta}(r,z) = \omega^{2} \mu_{0} \Delta \varepsilon \int_{\Omega_{D}} G(r,z \mid r',z') E_{\theta}(r',z') ds' + E_{\theta}^{M}(r,z) + E_{\theta}^{i}(r,z) .$$
(5.34)

We next insert the expansion (5.16) into the above equation, multiply both sides by the

eigenfunctions $\psi_k(r,z)$ and perform an integration over the cross section of the ADL disk on the r-z plane. This results in the matrix equation

$$\sum_{m=1}^{\infty} \left[(1 - \lambda_k \omega^2 \mu_0 \Delta \varepsilon) \delta_{km} - \gamma \lambda_k Q_{km} \right] c_m = -j \omega \mu_0 A_k , \qquad (5.35)$$

where

$$Q_{km} = \int_{\Omega_D} \frac{\partial}{r\partial r} \left[r \psi_k(r, z) \right] \frac{\partial}{r\partial r} \left[r \psi_m(r, z) \right] ds .$$
 (5.36)

If γ is small, the second term on the l.h.s. of Eq. (5.35) can be considered as a perturbation.¹ Up to the first order in the latter, the solution to the above equation can be written as

$$c_{k} = \frac{\left(-j\omega\mu_{0}\right)}{1 - \lambda_{k}\omega^{2}\mu_{0}\Delta\varepsilon - \gamma\lambda_{k}Q_{kk}} \left(A_{k} + \gamma\lambda_{k}\sum_{m(\neq k)}\frac{Q_{km}A_{m}}{1 - \lambda_{m}\omega^{2}\mu_{0}\Delta\varepsilon - \gamma\lambda_{m}Q_{mm}}\right).$$
(5.37)

In order to calculate the impedance of the ring, we once again use Eq. (5.20). The contribution of the field generated by the electric polarization currents can again be found using Eq. (5.23). However, one should not forget the effect of the field generated by the magnetization which yields the additional impedance

$$Z^{M} = -\frac{1}{|I|^{2}} \int_{\Omega_{i}} E^{M}_{\theta} \left(J^{i}_{\theta}\right)^{*} ds .$$
 (5.38)

Using Eqs. (5.16) and (5.32), the above expression can be rewritten as

$$-\frac{\gamma}{|I|^2} \sum_{k=1}^{\infty} a_k \iint_{\Omega_I \Omega_D} \left[J_{\theta}^i(r,z) \right]^* \frac{\partial}{r' \partial r'} \left[r' G(r,z \mid r',z') \right] \frac{\partial}{r' \partial r'} \left[r' \psi_k(r',z') \right] ds ds' .$$
(5.39)

¹ Actually numerical results show that $\lambda_k \sim c_k R t$ where c_k of the order of one or smaller. On the other hand, $Q_{mk} \sim d_{mk} / R^2$ where d_{mk} is again of the order of one or smaller. The perturbation is thus of the order of $\gamma(t/R)$. Therefore, for thin films, i.e. $t \ll R$, perturbation theory remains valid even if γ may not be so small. Since r', z' belong to Ω_D , the function G for $r, z \in \Omega_i$ can be expanded as

$$G(r, z \mid r', z') = \sum_{m} \psi_{m}(r', z') \int_{\Omega_{D}} G(r, z \mid r'', z'') \psi_{m}(r'', z'') ds'', \qquad (5.40)$$

which, after substitution in Eq. (5.39), leads to

$$-\frac{\gamma}{|I|^2} \sum_{k=1}^{\infty} \sum_{m=1}^{\infty} \overline{A}_m Q_{mk} c_k . \qquad (5.41)$$

Adding up Z^{D} and Z^{M} , and using Eq. (5.37), one finds that up to the first order in γ :

$$Z^{D} + Z^{M} = \frac{\left(j\omega\mu_{0}\right)}{\left|I\right|^{2}} \sum_{k=1}^{\infty} \frac{\left(\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{kk}\right)B_{k}\overline{B}_{k}}{1 - \lambda_{k}\left(\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{kk}\right)}$$
(5.42)

$$B_{k} = A_{k} + \frac{(\gamma/2)}{\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{kk}} \sum_{m(\neq k)} \frac{Q_{km}A_{m}}{1 - \lambda_{m}\left(\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{mm}\right)}$$
(5.43)

$$\overline{B}_{k} = \overline{A}_{k} + \frac{(\gamma/2)}{\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{kk}} \sum_{m(\neq k)} \frac{Q_{km}\overline{A}_{m}}{1 - \lambda_{m}\left(\omega^{2}\mu_{0}\Delta\varepsilon + \gamma Q_{mm}\right)}.$$
(5.44)

The circuit interpretation of this result is straightforward. The coupling between the primary and secondary turns of the transformer is now given by $M_k = \mu_0 \left(\overline{B}_k B_k / |I|^2 t\right)^{1/2}$. But, most important, in addition to the capacitance *C*, each secondary turn sees a shunt admittance (refer to Fig. 5.3)

$$Y_k = -\frac{\gamma Q_{kk} t}{j \omega \mu_0} \quad . \tag{5.45}$$

The real part of $Y_k = Y'_k - jY''_k$ represents a shunt conductance, accounting for the magnetic losses of the resonator. The overall quality factor Q of the device is then given by

$$\frac{1}{Q} = \frac{1}{Q_k^L} + \frac{1}{Q^C} + \frac{1}{Q_k^M}, \qquad (5.46)$$

127

where

$$Q_k^M = \frac{\omega C'}{Y'_k} = \frac{\omega(\varepsilon_{\parallel}' - \varepsilon_0)t}{Y'}.$$
(5.47)

5.2.3 Numerical simulations

On any substrate, the Green's function G can be written as

$$G(r,z \mid r',z') = \int_{0}^{\infty} \left[e^{-p_{0}|z-z'|} + \Gamma(k)e^{-p_{0}(z+z')} \right] J_{1}(kr)J_{1}(kr')\frac{kdk}{2p_{0}},$$
(5.48)

where $p_0^2 = k^2 - k_0^2$ with $k_0^2 = \omega^2 \mu_0 \varepsilon_0$, $J_1(z)$ is the first order Bessel function of the first kind, and $\Gamma(k)$ describes the reflection coefficient of a wave with the in-plane wave number k impinging on the surface of the substrate at z = 0. In case of a homogeneous substrate with the thickness D and dielectric constant ε_s , we have

$$\Gamma(k) = \frac{k_s^2 - k_0^2}{p_0^2 + p_s^2 + 2p_0 p_s \coth(p_s D)},$$
(5.49)

where $p_s^2 = k^2 - k_s^2$ in which $k_s^2 = \omega^2 \mu_0 \varepsilon_s$. If the substrate is bounded from below by a ground plane (which is assumed to be perfectly conductive for simplicity), then

$$\Gamma(k) = \frac{p_0 \tanh(p_s D) - p_s}{p_0 \tanh(p_s D) + p_s}.$$
(5.50)

Our aim is now to solve the eigenvalue problem (5.14). This will be numerically carried out employing the Galrekin's technique. To that end, we consider a finite set of test functions $f_n(r,z)$ defined on the cross section of the ADL, and expand each eigenfunction as

$$\psi_k(r,z) = \sum_n u_n^k f_n(r,z).$$
 (5.51)

We substitute this expansion into (5.14), multiply both sides by $rf_m(r,z)$, and carry out an integration over the disk cross section Ω_D . The result is the (generalized) matrix eigenvalue problem

$$\sum_{n} \overline{\bar{G}}_{mn} u_{n}^{k} = \lambda_{k} \sum_{n} \overline{\bar{W}}_{mn} u_{n}^{k} , \qquad (5.52)$$

where

$$\overline{\overline{G}}_{mn} = \int_{\Omega_D} \int_{\Omega_D} f_m(r,z) G(r,z \mid r',z') f_n(r',z') ds ds'$$
(5.53)

$$\overline{\overline{W}}_{mn} = \int_{\Omega_D} f_m(r,z) f_n(r,z) ds .$$
(5.54)

To simplify the calculation we divide the cross section of the disk in rectangular segments and use the test functions

$$f_m(r,z) = \begin{cases} r^{-1} & \text{if } (r,z) \in \Omega_m \\ 0 & \text{otherwise} \end{cases},$$
(5.55)

where Ω_m is the surface of the *m*-th rectangular segment.

Using this method, we have computed ψ_k and λ_k for different disk geometries,

and on different substrates. We restrict ourselves to ADLs with a thickness t not exceeding several microns. Even with in-plane dielectric constants of the order of 10^4 - 10^5 , the electromagnetic wavelength in such a material exceeds by several orders of magnitude the film thickness. Thus the variation of the field over the ADL thickness is negligible. We focus on the eigenvalue with the largest real part corresponding to the mode with the largest partial inductance and, thus, lowest resonance frequency. This we call the first mode or the dominant mode designated by the subscript 1, from which we extract the equivalent inductance and quality factor of the lowest resonance mode of the ADL resonator. Fig. 5.4 and Fig. 5.5 show the extracted inductance (real part) and quality factor of the dominant resonance mode of the ADL resonator as a function of frequency on high resistivity silicon (HRS) substrates. There several remarks to be made:

The disk diameter should be limited to few millimeters at most for on-chip applications. Especially for small disks, λ'_1 and, thus, the (real) inductance L'_1 only slightly changes at frequencies below 50 GHz (the range of interest in this work). Disks of larger dimensions show a drop in L'_1 at higher frequencies. The imaginary component λ''_1 , by contrast, increases rapidly with frequency in all cases, leading to a



Figure 5.4: Equivalent inductance (left) and quality factor (right) of the lowest mode of the ADL disk resonator as a function of frequency for different values of the radius (*R*). Simulations were carried out assuming a high-resistivity silicon substrate ($\varepsilon_s = 12$) with a thickness of 0.5 mm. No ground plane was assumed.

sharp drop in the inductive quality factor $Q_1^L = \lambda_1' / \lambda_1''$.



Figure 5.5: Equivalent inductance (left) and quality factor (right) of the lowest mode of the ADL disk resonator as a function of frequency for different values of the radius (R). Simulations were carried out assuming a grounded high-resistivity silicon substrate with a thickness of 0.5mm. The thickness of the ADL disk was 1 μ m.

The imaginary part λ_1'' is related to the losses caused by radiation into free space, leakage of microwave power into the surface-wave modes of the substrate, and the

polarization loss inside the substrate. The last factor is found to be negligible on insulating substrates with a small loss tangent ($<10^{-2}$). By far the most important loss contribution, however, comes from the surface wave effects. The ADL resonance is caused by the formation of TE standing waves inside the disk. Such modes naturally couple to the TE modes of the underlying substrate. If the substrate is not grounded, the cutoff frequency of the lowest TE mode is just zero, and TE waves can freely propagate on the substrate at any frequency, carrying away the energy stored inside the resonator. By contrast, if a ground plane is inserted underneath, a finite cutoff results (>100 GHz in 0.5mm-thick silicon) below which no propagation takes place. By comparison of Fig. 5.4 and Fig. 5.5, it is clear that grounding the substrate yields a major increase in the inductive quality factor Q_1^L . It also decreases L_1' due to the image currents in the conductive ground plane. But this effect is not remarkable when the disk radius is less than the substrate thickness.

If the substrate is not insulating, i.e. it has a finite resistivity, the inductive quality factor can be severely limited. Fig. 5.6 shows the inductive quality factor of an ADL resonator on silicon substrate as a function of frequency with different resistivity. In this case the major loss contribution comes from the generation of eddy currents, and their ohmic dissipation inside the substrate. For disk resonators built on silicon with a radius less than 1 mm, this effect becomes observable when the substrate resistivity becomes less than 100 Ω cm. The interesting point, however, is that even on low resistivity silicon substrates (10 Ω cm) Q_1^L values above 50 can be achieved if the frequency is not too high. Given the difficulty of the realization of conventional lumped elements with quality factors exceeding even 20 on a low-resistivity wafer, this result is particularly promising.

We next consider the resonance frequency $f_1 = \omega_1 / 2\pi$ of the first mode as function of the ADL (in-plane) dielectric constant for different values of the disk radius (as shown in Fig. 5.7). As expected, f_1 increases with increasing ε' . With 1µm-thick ADL films, resonance frequencies below 50 GHz can be obtained by using a ε'_{\parallel} above a few thousand. Comparison of Fig. 5.5 and Fig. 5.6 shows that the inductive quality factors above 1000 can be easily achieved by choosing the right disk dimension. In practice, this means that the overall quality factor of the resonator is mainly limited by the dielectric and magnetic losses of the ADL material which should be evaluated from its tangent loss and complex susceptibility.



Figure 5.6: Inductive quality factor of the first mode of an ADL disk (radius 0.5 mm, thickness 1µm) on a silicon substrate (0.5 mm-thick, with a ground plane) as function of frequency for different vaues of substrate resistivity ρ .



Figure 5.7: Resonance frequency of the first mode of the ADL disk as function of the in-plane dielectric constant ε'_{\parallel} for different disk radii. A grounded high-resistivity silicon substrate with a thickness of 0.5mm was assumed. The ADL thickness was 1 μ m.

5.2.4 Effects of electric and magnetic polarization loss of ADL

The electric polarization loss of the ADL can be estimated using a 3D lumped-element equivalent circuit consisting of capacitors C and series resistors R, as described in section 4.3.1 and section 4.4.3. The series resistance describes the ohmic loss arising inside the metallic patterns as (RF) current travels along the patterns and the capacitances coupling them. The effective in-plane dielectric constant of the ADL is given by

$$\varepsilon_{\parallel}' = \frac{(N-1)C}{t} = \frac{(N-1)\varepsilon_d'A}{t_d[Nt_M + (N-1)t_d]},$$
(5.56)

where N is the total number of metal layers and t is the overall ADL thickness, ε'_d and t_d denote the dielectric constant and thickness of the isolating layers, A is the overlap area between metallic patterns, and t_M is the thickness of the metallic layers. In the above equation we have used $C = \varepsilon'_d A / t_d$. If the number of layers is large, one finds

$$\varepsilon_{\parallel}^{\prime} \approx \frac{\varepsilon_d^{\prime} A}{t_d \left(t_M + t_d \right)} \,. \tag{5.57}$$

The loss tangent of the ADL is given by

$$\tan \delta = \tan \delta_d + \tan \delta_M, \qquad (5.58)$$

where $\tan \delta_d$ is the loss tangent of the isolating dielectric layers and

$$\tan \delta_{_M} = \omega RC \tag{5.59}$$

is the loss brought about by the pattern resistance. Since

$$R = \frac{\eta}{\sigma_M t_M},\tag{5.60}$$

with η a geometrical factor and σ_M the metal conductivity, one has from Eqs. (5.56-59):

$$\tan \delta_{M} = \eta (1 + t_{d} / t_{M}) \frac{\omega \varepsilon_{\parallel}'}{\sigma_{M}}.$$
 (5.61)

Now consider the magnetic losses. Under influence of a magnetic field perpendicular to the patterns, planar eddy currents start to flow in the plane of the patterns. Such currents, in turn, generate a dipolar magnetic field which opposes the original field. This effect can be represented by an effective susceptibility χ_{\perp} , i.e. the ratio between the volume average of the induced magnetic dipoles and the original field. For patterns with a thickness far below the skin depth inside the metal, an approximation assuming a constant magnetic field yields (see Appendix C for a detail derivation)

$$\gamma = \frac{\chi_{\perp}}{1 + \chi_{\perp}} = -j\eta_M f \,\omega \mu_0 \sigma_M D^2 \,, \qquad (5.62)$$

where *D* is the diameter of the patterns, *f* is their volume filling factor, and η_M is a dimensionless, geometry-dependent factor. Fig. 5.8 shows some pattern geometries and their associated geometrical factors η_M . The quantity η_M can be significantly reduced by micro-patterning the individual metallic squares. Note, however, that while η_M decrease, the pattern series resistance and, thus, η increases. Careful calculations are required to optimize the pattern shape.



Figure 5.8: Some patterns and their associated susceptibility coefficients.

From Eq. (5.45),

$$Y_k = -\frac{\gamma Q_{kk} t}{j \omega \mu_0} = \left(\eta_M f\right) \sigma_M D^2 Q_{kk} t .$$
(5.63)

The magnetic quality factor becomes

$$Q_k^M = \frac{\omega C'}{Y_k'} = \frac{\omega(\varepsilon_{\parallel}' - \varepsilon_0)}{\eta_M f D^2 \operatorname{Re}[Q_{kk}] \sigma_M} \approx \frac{1}{\eta_M f D^2 \operatorname{Re}[Q_{kk}]} \left(\frac{\omega \varepsilon_{\parallel}'}{\sigma_M}\right), \quad (5.64)$$

where Eq. (5.47) has been used plus the fact that $\varepsilon'_{\parallel} \gg \varepsilon_0$. Note that the capacitive

quality factor is proportional to $\sigma_M / \omega \varepsilon'$, while the magnetic quality factor is inversely proportional to this quantity. Thus (if the frequency dependence of $\operatorname{Re} Q_{kk}$ can be neglected, as was the case in our numerical experiments), Q^C decreases with frequency, whereas Q_k^M increases with frequency. The combined quality factor Q^{CM} is found from

$$\frac{1}{Q^{CM}} = \tan \delta_d + \frac{1}{Q^C} + \frac{1}{Q_k^M}, \qquad (5.65)$$

and has a maximum value given by

$$Q_{\max}^{CM} = \left(\tan \delta_d + 2\sqrt{\eta_M \eta f (1 + t_d / t_M) D^2 \operatorname{Re}[Q_{kk}]}\right)^{-1}, \qquad (5.66)$$

which is reached when

$$\frac{\omega \varepsilon_{\parallel}'}{\sigma_{_M}} = \sqrt{\frac{\eta_M f D^2 \operatorname{Re}[Q_{_{kk}}]}{\eta(1 + t_d / t_M)}}.$$
(5.67)

Note that the volume filling factor of the metallic patterns (f) can be written as

$$f = f_a \frac{t_M}{t_M + t_d}, \qquad (5.68)$$

where f_a is the surface filling factor of those patterns. Therefore, Eq. (5.66) can be written as

$$Q_{\max}^{CM} = \left(\tan \delta_d + 2\sqrt{\eta_M \eta f_a D^2 \operatorname{Re}[Q_{kk}]}\right)^{-1}.$$
(5.69)

Numerical experiments show that, for the lowest mode of the resonator, if the frequency is not too high, $\operatorname{Re}[Q_{11}] \sim v_1 / R^2$ where v_1 is almost a constant (~6). This yields

$$Q_{\max}^{CM} = \left[\tan \delta_d + 2(D/R) \sqrt{\eta_M \eta f_a v_1} \right]^{-1}.$$
(5.70)

It shows that the highest attainable quality factor is only determined by the geometrical
factors and not the material constants. Note, however, that this conclusion is only valid within the perturbative description of the magnetic losses and the approximate result Eq. (5.62).

5.2.5 Restrictions of the model

Since it assumes an impressed current source, the model presented above does not account for ohmic loss in the current-carrying ring. This can be accounted for by adding a resistance R_p in series with the primary winding of the transformer in the equivalent circuit model of Fig. 5.3. R_p can be approximately calculated by considering the flow of the electric current inside an isolated ring, since we neglect the effect of the environment on the ring current distribution anyway. This resistance includes the effect of the non-uniform distribution of current at high frequencies (skin effect) and is frequency-dependent.

A more serious shortcoming of the model used, however, is the assumption of a constant current along the metallic ring. In reality, the excitation current does not remain constant and charges accumulate along the ring. This does not affect the intrinsic behavior of the eigenmodes of the ADL disk discussed above. But a current varying along the ring breaks the rotational symmetry of the system. Thus, additional modes with an electric field varying in the azimuthal direction may also be excited. These modes (similar to whispering gallery mode or WGM of cylindrical dielectric rods [9]) have a very high resonance frequency compared to the low-lying rotationally symmetric modes considered in the previous section. Nevertheless, their resonance tails may still affect the resonator at low frequencies.

Although the model presented in the previous sections can be generalized to include the WGMs, this will significantly complicate the calculation. Instead, to account for the role of charge accumulation, we employ a perturbative approach. Consider the situation with a constant current flowing through the ring as the zero'th order (unperturbed) solution of the problem. Due to the symmetry of the system, this current brings about an electric potential along the ring which linearly varies with distance. From this potential distribution, we can calculate the accumulated charge on the ring and, subsequently, its effect on the impedance. Yet, even this calculation is not straightforward at high frequencies: charges on the ring will induce polarization charges on the ADL disk which, at high frequencies, generate polarization currents. A full description of the latter again needs the inclusion of rotationally non-symmetric eigenmodes. Therefore, to simplify the calculation, we adopt an electrostatic approach and include the effect of the charges by adding a shunt capacitance at the input of the resonator. Strictly speaking, this approach is valid at low frequencies only, but we will later justify its use by comparison with full wave simulations performed using commercial software. But before performing this calculation, let us see what the consequences will be for the behavior of the resonator.

Consider the ADL resonator close to its first resonance. A shunt capacitance C_p has now been added to its input to account for the charging effects. A series resistance R_p accounts for conductor loss in the ring. For simplicity, we neglect the effect of the higher modes. The resulting equivalent circuit is now given by Fig. 5.9 (the subscript 1, denoting the lowest mode, has been dropped). It consists of two *LC* resonators coupled by a mutual inductance.



Figure 5.9: Equivalent circuit of the ADL resonator including a shunt capacitance C_p .

If the imaginary parts of L, C, L_p, C_p, M as well as the resistance R_p are small, the resonance frequencies of this circuit follow from the solution of the simple algebraic equation:

$$\left(1 - \frac{\omega_D^2 \omega_p^2}{\omega_M^4}\right) \omega^4 - \left(\omega_D^2 + \omega_p^2\right) \omega^2 + \omega_D^2 \omega_p^2 = 0, \qquad (5.71)$$

where $\omega_D^2 = \omega_1^2 = 1/L'C'$ is the intrinsic frequency of the resonance of the lowest mode of the ADL disk,

$$\omega_p^2 = 1 / L_p' C_p' \tag{5.72}$$

is the resonance frequency of the primary inductor in parallel with the shunt capacitance, and

$$\omega_M^2 = 1/\left(M'\sqrt{C_p'C'}\right). \tag{5.73}$$

Eq. (5.71) has two solutions corresponding to the two coupled resonators. They are given by

$$\omega_{\pm}^{2} = \frac{\omega_{D}^{2} + \omega_{p}^{2} \pm \sqrt{(\omega_{D}^{2} - \omega_{p}^{2})^{2} + 4\omega_{D}^{2}\omega_{p}^{2}\kappa^{2}}}{2(1 - \kappa^{2})},$$
(5.74)

where

$$\kappa = \frac{\omega_D \omega_p}{\omega_M^2} = \frac{M'}{\sqrt{L'_p L'}}$$
(5.75)

is the magnetic coupling factor of the transformer. It is instructive to introduce the coupling factor k between the two resonator, i.e.,

$$k = \frac{2\omega_D \omega_p \kappa}{\omega_D^2 - \omega_p^2},$$
(5.76)

which depends not only on the magnetic coupling factor κ , but also on the differences between the resonance frequencies of the two resonators, and consider the cases of weak coupling ($k^2 \ll 1$) and strong coupling ($k^2 \gg 1$).

Weak coupling (k2<<1)

Expansion of the solution (5.74) then shows that the resonances of the ADL disk and the primary ring are shifted to

$$\omega_D^2 \approx \frac{\omega_D^2}{1 - \kappa^2} + \frac{k^2 (\omega_D^2 - \omega_p^2)}{4(1 - \kappa^2)} \\ \omega_p^2 \approx \frac{\omega_p^2}{1 - \kappa^2} - \frac{k^2 (\omega_D^2 - \omega_p^2)}{4(1 - \kappa^2)}$$
(5.77)

Thus, if $\omega_p > \omega_D$ ($\omega_p < \omega_D$), the disk resonance frequency will be shifted to lower (higher) values. This shift is reduced if the magnetic coupling becomes smaller, or if the difference between the resonance frequencies of the two *LC* resonators becomes large.

Strong coupling (k2>>1)

In this limit, the magnetic coupling is high and/or the resonance frequencies of the two resonators are very close to each other. Solving Eq. (5.74) in this circumstance yields

$$\omega_{\pm}^{2} \approx \frac{\omega_{D}^{2} + \omega_{p}^{2} \pm 2\omega_{D}\omega_{p}\kappa}{2(1 - \kappa^{2})}.$$
(5.78)

When κ is close to one, the two resonance frequencies are highly separated and could be far different than the original ones.

5.3 Full wave simulations of ADL resonators

To verify the validity of the theoretical model developed in previous section, full wave simulation was carried out using commercially available simulation software, HFSS, which employs the finite element method. The ADL disk is modeled as an anisotropic effective material, with the in-plane dielectric constant ε_{\parallel} and the out-of-the-plane dielectric constant ε_{\perp} which we calculated in section 4.3.1. The magnetic properties of the ADL can be derived as follows. Because the thickness of the metallic layer is much smaller than its lateral dimensions and the magnetic field is mainly perpendicular to the ADL disk, the eddy current loss due to horizontal magnetic field is negligible for the dominant resonance mode, which implies that the relative permeability and magnetic loss tangent perpendicular to ADL disk is given by the negative imaginary part of Eq. (5.62). And therefore, the real part of perpendicular permeability is given by $\mu'_{\perp} = 1/(1 + \tan^2 \delta_{\text{Mag}})$, where $\tan \delta_{\text{Mag}}$ is the magnetic loss tangent of ADL. In the HFSS simulation, we first assume a magnetic loss tangent value of 1. The choice of this value is based on Eq. (5.62) which is repeated here for convenience.

$$\gamma = \frac{\chi_{\perp}}{1 + \chi_{\perp}} = -j\eta_M f \,\omega\mu_0 \sigma_M D^2 = -j \tan \delta_{\text{Mag}}$$

Here a micro-patterned ADL structure is assumed to minimize the magnetic loss, i.e. η_M

taking the value of 0.0077 (Fig.5.8). The volume filling factor f is calculated based on ADL structure described in chapter 4 with ADL metallic pattern size of $5x5 \ \mu\text{m}^2$, and the same pattern size is assigned to D. The frequency is fixed at 30 GHz and the conductivity of Al film is assumed to be $3x10^7$ S/m. This yields a value of the magnetic loss tangent to be 0.73. Therefore, the choice of 1 for magnetic loss tangent is justified in this case. However, for larger pattern size and/or ADL without micro-patterning, the magnetic loss

will be higher than this value. For example, for ADL pattern of $11x11 \ \mu m^2$, without micro-patterning, the magnetic loss tangent could be as high as 19. (We will come back to this point later, but the simulation presented here will suffice to study the performance of ADL resonator in a more or less optimized scenario.) The magnetic properties of the ADL disk can thus be expressed by the permeability tensor:

$$\mu = \mu_0 \begin{bmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 0.5(1-j) \end{bmatrix}.$$
 (5.79)

To characterize the ADL resonator from simulations, we first transform the resulting *S*-parameters into *Z*-parameters and then calculate $Z_s = Z_{11} + Z_{22} - Z_{12} - Z_{21}$, which is the input impedance seen with the resonator excited differentially. This is to eliminate the un-wanted impedance through the substrate and can be understood by considering the T-network representation of the resonator [10]. The *Q*-factor is calculated using Eq. (5.2), while the resonance frequency f_0 is determined by the frequency at which the real part of input impedance Z_s reaches its maximum value, f_1 and f_2 are frequencies where Re(Z_s) has half of the peak value. This is justified by considering the input impedance for a parallel *RLC* resonator (our ADL resonator can be modeled as a parallel *RLC* resonator as shown in the previous section). Near the resonance, the input impedance can be approximated by the following expression [1]:

$$Z_{\rm in} \simeq \frac{R}{1 + 2jQ\Delta f / f_0}, \qquad (5.80)$$

where Δf is the frequency deviation from the resonance and f_0 is the resonance frequency. The real part of Z_{in} is

$$\operatorname{Re}(Z_{\text{in}}) = \frac{R}{1 + 4Q^2 \left(\Delta f / f_0\right)^2}.$$
(5.81)

At resonance, the input impedance is just *R*. By setting the $\text{Re}(Z_{\text{in}})$ to *R*/2, and solving Eq. (5.81), one finds two Δf values which correspond to two frequency values of f_1 and f_2 respectively. The quality factor Q is therefore expressed as in Eq. (5.2).



Figure 5.10: HFSS simulation results for ADL disk resonators (R=0.75 mm) with different values of in-plane effective dielectric constants, coupled with ring inductor.

Fig. 5.10 shows the simulated real parts of Z_s (= $Z_{11}+Z_{22}-Z_{12}-Z_{21}$) as a function of frequency for different in-plane dielectric constants ε_{\parallel} of the ADL for disk resonators with R=0.75 mm coupled with a ring inductor (Fig. 5.1). Clear resonances can be identified in the simulation frequency range for ε_{\parallel} higher than 10000. The trend of lower resonance frequency for higher ε_{\parallel} is observed as expected. The Q-factor of 140 is achieved for ε_{\parallel} equals to 20000 with magnetic loss tangent tan $\delta_{\text{Mag}} = 1$, and dielectric loss tangent tan $\delta = 0.002$ for ADL material. To verify our theoretical models developed in the previous section, we compare the resonance frequencies obtained from the HFSS simulations and those from calculation based on the physical model. To that end, we re-calculate the resonance frequencies for different ADL disk geometries on AF45 glass substrate, which is shown in Fig. 5.11. By a comparison between Fig. 5.10 and Fig. 5.11, it is observed that the first resonance frequencies of ADL resonator with different in-plane dielectric values are all shifted to higher end. Taking the case of $\varepsilon_{\parallel} = 20000$ as first resonance frequency of ADL disk example, the resonator with an



Figure 5.11: Resonance frequency of the first mode of the ADL disk as function of the in-plane dielectric constant for different disk radii. A grounded AF45 glass substrate with a thickness of 0.7 mm was assumed. The ADL thickness was 1 μ m.

R=0.75 mm is around 28.3 GHz, while in Fig. 5.10 it is found to be 33.1 GHz from the HFSS simulation result, more than 15% differences. This discrepancy is related to the shift in resonance frequency caused by the magnetic coupling between the current-carrying ring and the ADL disk as discussed in 5.2.5. Notice that the geometric configurations of ADL disk coupled with the ring inductor intrinsically implies a not weak magnetic coupling. To illustrate the influence of magnetic coupling factor on the two resonance frequencies, calculation of the resonance frequency shifts by solving Eq. (5.74) is carried out. In this calculation, we assume the ADL disk and the ring inductor have their first resonance frequencies at 30 GHz and 12 GHz respectively and sweep the magnetic coupling factor. Result is shown in Fig. 5.12. It is inferred from the figure that the ADL disk (ring inductor) resonance frequency shifts to higher (lower) end as the magnetic coupling factor increases. When the magnetic coupling factor is 0.5, the shift of ADL disk resonance frequency could be as high as ~20%, while the ring inductor has a shift of only ~2%. The small shift of the resonance frequency of the ring inductor is also observed (not shown here) in the HFSS simulation.



Figure 5.12: Calculated resonance frequency shifts as a function of magnetic coupling factor. Assume the ADL disk and the ring inductor have a first resonance frequency of 30 GHz and 12 GHz respectively.

To demonstrate the 'real' resonance frequency of the ADL disk, we look into a weak magnetic coupling structure. This could be implemented by a dipole, a microstrip transmission line or putting the ring outside the ADL disk. Here for simplicity, we choose to couple the ADL disk with a dipole as schematically shown in Fig. 5.13. A full wave HFSS simulation is carried out for ADL disks of R=0.5 mm and R=0.75 mm with the same ADL material parameters and results are shown in Fig. 5.14. In both cases, clear resonances are observed. Notice that the peak impedances at resonance frequencies are much lower than the ring coupled ones due to the very weak coupling as expected. Besides the first resonance mode, higher modes corresponding to the WHG modes are also clearly visible. By a comparison with Fig. 5.11, it is found that the first resonance frequencies of ADL disks with the two different radii well match the results obtained from the theoretical model. Taking two cases for illustration, the first resonance mode of ADL disk with R=0.5 and an in-plane dielectric constant of 25000 is found to be 31.1 GHz, while it is exactly 31.1 GHz from the calculation (Fig. 5.11). For R=0.75 mm with in-plane dielectric constant of 20000, the simulation and calculation results for the first resonance frequencies are both around 28.3 GHz. Therefore, the theoretical model fits



Figure 5.13: Schematic view of a weakly coupled ADL disk resonator, i.e., coupled with a dipole structure.

very well with the HFSS simulation results, which indicates the validity of the physical model presented in the previous section and can be used as a guideline for ADL disk resonator design.

Besides the major resonances, we also notice that minor resonance peaks are also present in Fig. 5.10 for the ring coupled ADL resonator. By comparison Fig. 5.10 with Fig. 5.14, some of the minor peaks at higher frequency end can be explained as the higher resonance modes of the disk resonator. However, the resonance lower than the first disk resonance frequency can not be explained this way. Taking the case of ε_{\parallel} =20000 as an

example, the major resonance which corresponds to the first resonance mode of the ADL disk occurs at ~33.1 GHz, while a weak resonance peak at ~30 GHz also be observed. Since the ring itself has its second resonance at around 37 GHz (as indicated by the curve with in-plane dielectric constant of the ADL to be one), this minor peak might be related to the coupling between ADL resonator and this higher mode of the ring. To verify this hypothesis, a HFSS simulation is performed by modifying the ring geometry which has lower second self-resonance frequency. The result (not shown here) shows that this modification has little influence on the ADL disk resonance, but shifts the minor peak to its lower end. Therefore, the hypothesis mentioned above is justified.



Figure 5.14: HFSS simulation results for ADL resonators of R=0.5 mm (top) and R=0.75 mm (bottom) with different values of in-plane effective dielectric constants, coupled with a dipole structure.

5.4 Experimental results and discussions

Preliminary experiments were carried out to demonstrate the possibility of realizing high-O ADL resonators. ADL resonators were built on a 500 µm glass substrate (AF 45) to minimize the substrate loss in the first place. The fabrication process is quite similar to the multi-layer ADL shields which were described in section 4.4.3. Here we have 8 Al layers (100 nm for each layer) isolated by 7 PVD sputtered Aluminum oxide films (30nm for each layer). This gives an ADL material of $\sim 1 \mu m$ in thickness. The sizes of metallic squares were chosen to be 3x3, 5x5, 7x7, 9x9 and $11x11 \ \mu m^2$ to achieve different values of effective in-plane dielectric constant of the ADL material. Using Eq. (5.56), these pattern sizes yield effective in-plane dielectric constant values of approximately 2300, 9300, 20000, 37000 and 58000 respectively. Based on our numerical experiments, ADL disks with radii of 0.25, 0.5 and 0.75 mm were fabricated. After the formation of ADL disks, a 1 µm silicon oxide layer was deposited to form an etching stop layer. Finally an Al layer of 3 um thick was sputtered and patterned to form the conducting ring shown in Fig. 5.1 which can excite resonance of the ADL resonator. To suppress the undesired surface-wave mode in the substrate, $1.4 \mu m$ Al is applied on the backside of the glass wafer which will be grounded during the measurement. For comparison, micro-patterning ADL was also fabricated on the same wafer.

Two port S-parameter measurements were performed on a HP8510 vector network analyzer. To eliminate the influence of the measurement pads, open and short structures were fabricated on the same wafer and de-embedded. The extraction of the resonance frequency and quality factor is quite similar as that of HFSS simulation and will not be repeated here.

Fig. 5.15 shows the measured Re (Z_s) as a function of frequency for selected different ADL patterns. The dominant resonance mode of ADL resonator with larger ADL metallic pattern size is clearly observed. The resonance frequencies of the ADL resonators with and without micro-patterning with the same metallic pattern sizes are close to each other, which implies that the effective in-plane dielectric constants are more or less the same. However, the quality factor, as computed for the pattern size of 9x9 μ m², is estimated to be 25, which is in sharp contrast with the results shown in the HFSS simulations of Fig 5.10. This could be resulted from an under-estimation of the magnetic and/or dielectric loss of the ADL material. As mentioned in the previous section, the magnetic loss tangent for large ADL pattern without mciro-patterning could be as high as 19. A HFSS



Figure 5.15: Measurement results for ADL resonators (R=0.75 mm) with different ADL patterns.

simulation shows that, with magnetic loss tangent $\tan \delta_{\text{Mag}} = 15$ while the dielectric loss tangent is the same ($\tan \delta = 0.002$) as calculated in Fig. 5.10, the Q-factor of ADL disk ($R=0.75 \text{ mm}, \epsilon_{\parallel}' = 20000$) is as low as ~25. On the other hand, with ADL dielectric loss

tangent tan $\delta = 0.02$, while keeping the magnetic loss tangent the same (tan $\delta_{Mag} = 1$) as

calculated in Fig. 5.10, the Q-factor is around 30, which is also comparable with our experimental result. However, note also that micro-patterning the ADL elements does not yield a higher Q-factor as expected (it even lowers the Q-factor!). This implies that the eddy-current loss is not a dominant factors in the devices investigated. The low Q-factor observed should then be mainly due to dielectric loss which comprises the dielectric loss of the insulating dielectric (RF sputtered aluminum oxide in our case) and the pattern resistance [see Eq. (5.58)]. The higher than expected effective dielectric loss of the ADL may be related to a number of factors as argued below.

149

Normally, aluminum oxide has a very low tangent loss at microwave frequencies (~0.0002 or lower for alumina substrate [11]). However, the aluminum oxide in our experiment is prepared by reactive RF sputtering of Al target with O_2 plasma, which has poor step coverage especially near the edges of the metallic ADL patterns. This results in very thin Al oxide film near these edges which can introduce electron tunneling effects between two Al layers. This effect plus the fact that the film is extremely thin (30nm) (there is little data in literature on the microwave loss tangent of Al oxide thin film) may lead to an increase of tangent loss of the film. A separate measurement of the loss tangent of the aluminum oxide film will be helpful to identify this issue.

The tangent loss induced by the pattern resistance is given by Eq. (5.59). ADS simulations give a pattern resistance of ~1 Ω assuming 9x9 μ m² patterns and an Al conductivity of 3.3×10^7 S/m (which is the conductivity of bulk sputtered Al in our sputtering system) This yields a loss tangent of 0.009 at 30 GHz. However, a close look at the SEM images of the cross section of the ADL resonator (Fig. 5.16), shows local narrowing effect of the metallic layer caused by the crystallization and grain growth during the film deposition. This effect can lead to significant decrease of local conductivity of Al and increase of the pattern resistance R. Therefore, the loss tangent $\tan \delta_{M}$ due to pattern resistance will be higher than 0.009. The above arguments are further justified by the fact that with micro-patterning, the quality factor becomes even less. This can be qualitatively explained by considering the pattern resistance R when cutting slots in the square patterns. Since R is also proportional to a geometric factor η as shown in Eq. (5.60), the micro-patterning in ADL manifests itself by increasing R due to local narrowing near the center of the metallic pattern, as can be seen from Fig. 5.8. Therefore, a higher tangent loss due to metallic pattern resistance can be expected for micro-patterned ADL resonator, which overwhelms the eddy current loss reduction when compared with an un-patterned ADL resonator.

To overcome this problem and achieve higher quality factor, one obvious solution is to increase the conductivity of the metal layer by using copper for example. Another way is to make the metallic layer thicker. The geometry of the metallic pattern in ADL can also be further optimized, however a tradeoff between a lower geometric factor η for pattern resistance *R* and a sufficient suppression of eddy current loss has to be made. It is also suggested to use dielectric material which has higher dielectric constant (such as TiO₂ or ferroelectric materials) as the isolating material in ADL structure, substituting the 150

Al oxide film. One obvious benefit by using such material is that the film thickness can be a few times higher than the Al oxide case, to obtain the same in-plane dielectric constant of ADL. This will eliminate the possibility of local electron tunneling as mentioned before. On the other hand, the metallic pattern size can also be decreased which leads to less pattern resistance as well as less eddy current loss.



Figure 5.16: SEM image of cross section of ADL resonator. Eight layers of the metallic patterns in ADL are resolved. The local narrowing of the metallic layer is also shown.

5.5 Conclusions

In this chapter, a novel type of on-chip microwave resonator was proposed based on a thin disk of artificial dielectric material, which can potentially achieve very high quality factor. Based on an effective material description of ADL, we developed a theoretical model to analyze the resonance modes of the disk resonator. The dielectric loss (due to the insulating dielectric inside the ADL as well as the resistance of the ADL metallic patterns) and the magnetic loss (caused by the eddy currents inside the patterns) were taken into account in the model. Numerical simulations show that very high quality factor

can be realized on semi-insulating substrates. This prediction was further verified by full wave simulations carried out by in commercially available software HFSS. Results show quality factors exceeding 100. However, although the resonance behavior of the ADL disk was also seen in the experiments carried out, a much lower quality factor was observed. We attribute this degradation to the higher than expected resistance of the individual metallic patterns of the ADL.

References

- [1] D. Pozar, "Microwave Engineering", Wiley, New York, 2005.
- [2] J.N. Burghartz, B. Rejaei, "On the design of RF spiral inductors on silicon", IEEE. Trans. Electron Devices, Vol. 50, No. 3, pp. 718-729, 2003.
- [3] Y.C. Chiang, C.C. Lin, Y.J. Chan, "The design of a narrowband MIC lumped-element coupled-resonator bandpass filter", Microw. And Opt. Tech. Lett., Vol. 18, No. 6, pp. 414-417, 1998.
- [4] T. Pan, A. Baldi, E. Davies-Venn, R.F. Drayton, B. Ziaie, "Fabrication and modeling of silicon-embedded high-Q inductors", J. Micromech. Microeng., Vol. 15, No. 4, pp. 849-854, 2005.
- [5] M.A.T. Sanduleanu, G. Zhang, J.R. Long, "31-34GHz low noise amplifier with on-chip microstrip lines and inter-stage matching in 90-nm baseline CMOS", Proceedings of RFIC, 2006, pp. 143-146.
- [6] T.S.D. Cheung, J.R. Long, "Shielded passive devices for silicon-based monolithic microwave and millimeter-wave integrated circuits", IEEE J. Solid-State Circuits, Vol. 41, No. 5, pp. 1183-1200, 2006.
- [7] C. Kim, I. Song, C. Song, C. Cheon, Y. Kwon, S. Lee, "A micromachined cavity resonator for millimeter-wave oscillator applications", Sensors and Actuators A, Vol. 83, No. 1-3, pp. 1-5, 2000.
- [8] A.R. Brown, P. Blondy, G.M. Rebeiz, "Microwave and millimeter-wave high-Q micromachined resonators", Int. J. RF and Microwave CAE, Vol. 9, No. 4, pp.

326-337, 1999.

- [9] D. Cros, P. Guillon, "Whispering gallery dielectric resonator modes for W-band devices", IEEE Trans. Microwave Theory Tech., Vol. 38, No. 11, pp. 1667-1674, 1990.
- [10] B. Rejaei, A. Akhnoukh, M. Spirito, L. Hayden, "Effect of a local ground and probe radiation on the microwave characterization of integrated inductors", IEEE Trans. Microwave Theory Tech., Vol. 55, No. 10, pp. 2240-2247, 2007.
- [11] R.A. Woode, E.N. Ivanov, M.E. Tobar, D.G. Blair, "Measurement of dielectric loss tangent of alumina at microwave frequencies and room temperature", Electron. Lett., Vol. 30, No. 25, pp. 2120-2122, 1994.

Chapter 6

Conclusions and Recommendations

6.1 Conclusions

The applications of ferroelectric materials and artificial dielectric materials in monolithic microwave integrated circuit (MMIC) have been proposed and investigated. For ferroelectric materials, both material development and device optimization have been addressed. It is for the first time, to the best of our knowledge, that a special kind of artificial dielectric material is designed and implemented in MMIC. Applications of this artificial dielectric in MMIC are proposed and realized experimentally.

153

In the following, the main results of this work are summarized. Recommendations to future work will be proposed as well.

• It was shown that crack free and smooth surface barium strontium titanate (BST) thin film have been successfully prepared by chemical solution deposition (CSD)

method on platinum coated silicon substrates. Based on the impedance measurement of parallel plate tunable capacitor with Al/BST/Pt configuration, asymmetric characteristics of both capacitance and loss with respect to the bias voltage applied between the Pt and Al electrodes is observed and is found to be attributed to the formation of an ohmic contact and a Schottky barrier at the Al/BST and Pt/BST interfaces, respectively. (*chapter 2*)

- It was proposed to realize ferroelectric thin film tunable microwave devices in parallel-plate configuration, to maximize its tunablity, by substrate transfer technology (SST) to lift the stringent requirement on the metallic electrodes. However, varactors and microstrip transmission lines fabricated in this way were found to be short circuited. This was found to be resulted by micro-cracks originating from the SiO₂/BST interface due to the large thermal expansion mismatch. (*chapter 2*)
- The tunability of ferroelectric varactors utilizing thick (>1µm) ferroelectric films can be increased by implementing nano-dot electrodes. Using a semi-analytic model, it was found that the tuning factor of a single nano-dot varactor built on a thick BST layer can reach values higher than 2, even with a dc bias voltage as low as 5V. Arrays of nano-dot varactors were then proposed to achieve sufficient capacitance densities. Their behavior as function of the dot radius and separation, as well as the thickness of the isolating layer were studied and it was found that the nano-dot array can maintain the high tuning factor of a single dot, provided that the inter-dot distance does not become too small. (*chapter 3*)
- A novel method for reducing the substrate loss of Si-based integrated CPW transmission lines was introduced by exploiting the shielding effect of a thin film layer with a high in-plane dielectric constant. This can be realized either by ferroelectric material with super high dielectric constant or an artificial dielectric layer proposed by us which can be modeled as an effective media with extremely high dielectric constant. The ADL comprises two patterned metal layers separated by a thin conventional dielectric, and can be readily built in a standard IC process. It was found that the ADL shield significantly lowers the attenuation, but only moderately changes the characteristic impedance of the transmission lines. (*chapter 4*)

- It was experimentally shown that the shield effectiveness of ADL can be improved by increasing the metallic element size, while micro-patterning the ADL to suppress the eddy current loss. By this way, it was demonstrated that CPW built on ordinary conductive silicon can achieve comparable total attenuation as the one fabricated on HRS substrate. We also showed that the shield effectiveness can alternatively be improved by stacking a few ADL layers. Loss reduction of CPW based microwave devices, i.e. branch-line coupler and ring hybrid, by applying ADL shields were demonstrated by simulation and/or experiment. The ADL shield was also shown, by simulation and experiment, to potentially be applied to on-chip spiral inductors to optimize their quality factors. (*chapter 4*)
- On-chip microwave resonator based on ADL was proposed and theoretically investigated. It was shown by full wave simulation that the quality factor of ADL resonators can potentially be as high as more than 100. The ADL resonators were also experimentally demonstrated with clear resonance characteristics. The degradation of quality factor in experiment was attributed to high resistance loss in ADL metallic patterns. (*chapter 5*)

6.2 Recommendations

Since BST a very good candidate material for tunable microwave devices, and by substrate transfer technology, the complete use of the tunability of BST can be achieved, further investigation to eliminate micro-crack formation during deposition is highly recommended. Applying seed layers other than SiO₂, which has lower lattice mismatch with BST such as titanium oxide, is one candidate to obtain crack free BST film. Yet it has to be verified by further experiments.

Although the tunability enhancement of ferroelectric thick film varactors is theoretically investigated, experiments have to be designed and carried out to verify this effect. However, to fabricate the dot arrays in nanometer scale, advanced lithography, such as e-beam lithography, has to be applied. One possible quick way of test is to use an atomic force microscopy (AFM) working at its capacitance measurement mode. Since the head of the cantilever in an AFM system is usually in tens of nanometer range, if one can apply certain bias voltage on it, the varactor characteristic can be obtained by measuring capacitance as function of DC bias. It should be mentioned that, however, the dot array

varactor can not be tested in this way.

In addition to the discussed micro-patterning and multi-layered ADL shield to enhance the shielding efficiency, the insulating dielectric can also be substituted by material with higher dielectric constant such as TiO_2 . Another benefit to implement TiO_2 is that for the same effective in-plane dielectric constant value, a smaller ADL pattern size will be chosen, therefore, reducing the eddy current loss. Furthermore, the micro-patterning ADL is not optimized. Calculation is necessary to find the best patterning shape to minimize the eddy current loss and also minimize the metallic pattern resistance.

Although ADL resonators with clear resonances have been successfully fabricated, it is just a proof of principle. Compared with simulation results, the quality factor is substantially low. To diagnose this issue, further investigation has to be carried out. First, it is necessary to measure the dielectric loss of the insulating dielectric, i.e. the 30nm aluminum oxide layer. To gather a complete understanding of the ADL media, devices have to be designed to measure the effective dielectric constant and loss tangent of the ADL with certain accuracy. Furthermore, ADL resonators on HRS substrate were not successfully demonstrated with unknown reasons. More efforts need to be given to find the root causes, because Si is such an important substrate for future MMIC.

Appendix A Ferroelectric Phase Transition

In this appendix, a brief introduction to the phenomenological theory, also known as Landau-Ginzburg-Devonshire (LGD) theory [1] of ferroelectric phase transition is given. This is indeed a macroscopic theory without regard to any atomic structure. Nevertheless, it is helpful to understand the macroscopic phenomena related to spontaneous polarization and ferroelectric phase transitions.

In general, many approaches can be taken to understand the origin of spontaneous polarization and the ferroelectric phase transition. From a lattice dynamics point of view, a displacive ferroelectric transition occurs if a transverse optical phonon mode "softens" until its frequency reaches zero and wavelength becomes infinite at Curie temperature [2]. In another word, the soft mode vibration locks in a displacive distortion, which creates spontaneous polarization necessary for ferroelectricity.

In contrast to the microscopic theory of soft phonon mode, another way to look at the same problem is through phenomenological or macroscopic theory. The phenomenological theory of ferroelectrics is indeed thermodynamic. In general, the thermal, electrical and mechanical properties of a crystal are related to each other. The main task of a phenomenological theory is to explain the properties of a crystal by as few parameters as possible, in order to clarify the relationship between observed properties. Among the possible thermodynamic functions, the elastic Gibbs energy G_1 , which is obtained from the internal energy U by the following transformation $G_1 = U - TS - X_i x_i$, is usually used for ferroelectric phase transition, where T is temperature, S is entropy, X_i and x_i are strain and stress of the crystal respectively [3].

For simplicity, in the following discussion, only a one dimensional problem is considered. It will nevertheless reveal the essential characteristics of ferroelectric phase transition. Following Landau, the elastic Gibbs free energy can be expressed as a power series of the order parameter, the electric polarization P (the order parameter is the macroscopic parameter that appears in the less symmetric phase in a symmetry-breaking phase transition):

$$G_{1} = G_{10} + \frac{1}{2}\alpha P^{2} + \frac{1}{4}\beta P^{4} + \frac{1}{6}\gamma P^{6}, \qquad (A.1)$$

where G_{10} is the elastic Gibbs free energy when polarization is zero and α , β , γ depend, in general, on temperature. The use of G_1 is because, in the first place, it gives the relations between *E* and *P* under ordinary experimental conditions (constant temperature and constant stress) and, in the second place, is related to the Gibbs free energy *G* by the

expression of $G_1 = G - E \cdot P$. In the absence of the electric field, i.e., E = 0, the two are

the same. Notice that the series only contains even powers of *P* because the free energy does not change with polarization reversal. It should also be mentioned that the value of γ should be positive since otherwise the state of infinite spontaneous polarization will be the stable phase. This formulation should be applied for both the paraelectric and ferroelectric phase over the whole temperature range [3].

It then can be shown that the electric field E can be found by the first derivative of G_1 :

$$E = \frac{\partial G_1}{\partial P} = \alpha P + \beta P^3 + \gamma P^5.$$
 (A.2)

Particularly, when E=0 one obtains

$$P(\alpha + \beta P^2 + \gamma P^4) = 0.$$
 (A.3)

whose solution is given by

$$P_s^2 = \frac{-\beta \pm \sqrt{\beta^2 - 4\alpha\gamma}}{2\gamma}, \qquad (A.4)$$

If the right hand side of the above relation is positive, and if the substitution of P_s in Eq. (A.1) leads to a free energy less than G_{10} , then one has a stable spontaneously polarized state. Otherwise, the paraelectric phase is the stable state of the system [3].

Depending on whether β is positive or negative, a second order or first order phase transition will be found (the *n*th order phase transition is one in which the (*n*-1)th derivative of Gibbs energy is continuous while the *n*th shows a jump at the transition temperature).

In the case of $\beta > 0$, (A4) yields an acceptable solution only if $\alpha < 0$ and

$$P_s^2 = \frac{-\beta + \sqrt{\beta^2 - 4\alpha\gamma}}{2\gamma} \,. \tag{A.6}$$

If $|\alpha|\gamma \ll \beta^2$, which is the case of most real ferroelectric crystals, we obtain:

$$P_s^2 = -\alpha \,/\,\beta \,. \tag{A.7}$$

By substituting this equation into the power series for the elastic Gibbs energy, and taking $G=G_1$ when E=0 and $G_0=G_{10}$ into consideration, we have

$$G - G_0 = -\frac{\alpha^2}{4\beta} \,. \tag{A.8}$$

Therefore, we see that the above solution of P_s gives a smaller Gibbs free energy than when $P_s=0$, and thus the spontaneously polarized phase becomes stable.

As stated earlier, the coefficients of the power series are generally temperature dependent. One frequently expresses α as a function of temperature in the form of Taylor expansion of temperature including only the first order term:

$$\alpha = \frac{1}{\varepsilon_0 C} (T - T_0). \tag{A.9}$$

The requirement for the occurrence of spontaneous polarization is that $\alpha < 0$, and so that $T < T_0$. In this case, we have $T_0 = T_c$. When the temperature is above the Curie temperature, for weak electric fields, we can express the reciprocal of dielectric susceptibility as

$$\chi^{-1} = \varepsilon_0 \left(\frac{\partial E}{\partial P}\right)_{P=0} = \varepsilon_0 \alpha = \varepsilon_0 \frac{T - T_0}{\varepsilon_0 C} = \frac{T - T_0}{C}.$$
 (A.10)

And the relative dielectric constant is given by

$$\varepsilon_r = 1 + \chi = 1 + \frac{C}{T - T_0}$$
 (A.11)

Since normally in ferroelectric materials $\chi >>1$, we can ignore the 1 term, which just

gives the Curie-Weiss law.

Besides, by substituting Eq. (A.9) into Eq. (A.7), one finds for $T \le T_0$

$$P_s = \sqrt{\frac{T_0 - T}{\varepsilon_0 \beta C}} , \qquad (A.12)$$

which implies a continuous change of spontaneous polarization to zero at the Curie point. Furthermore, substituting Eq. (A.9) into Eq. (A.8), one obtains

$$G - G_0 = -\frac{1}{4\beta} (\frac{1}{\varepsilon_0 C})^2 (T - T_0)^2, \qquad (A.13)$$

which clearly shows a second order transition as defined previously.

For the case of $\beta < 0$, a similar analysis can be carried out. Here we restrict ourselves to a brief outline of the results : (1) The Curie-Weiss law is still valid; (2) The Curie point or phase transition temperature T_c does not coincide with the Curie-Weiss temperature T_0 , but given by

$$T_c = T_0 + \frac{3}{16} \cdot \frac{\beta^2 \varepsilon_0 C}{\gamma}; \qquad (A.14)$$

(3) The phase transition is first order and the spontaneous polarization changes abruptly near T_c . [3]

Finally, it is worth mentioning that the phenomenological theory can also be used to discuss the influence on polarization by strain and stress and is also applied to thin ferroelectric films.

References

- [1] C. Basceri, S.K. Streiffer, A.I. Kingon, R. Waser, "The dielectric response as a function of temperature and film thickness of fiber-textured (Ba,Sr)TiO3 thin films grown by chemical vapor deposition", J. Appl. Phys., Vol 82, No. 5, pp. 2497-2504, 1997.
- [2] R. Waser (Ed.), "Nanoelectronics and Information Technology", Wiley-VCH, Weinheim, Germany, 2003.
- [3] T. Mitsui, I. Tatsuzaki, E. Nakamura, "An Introduction to the Physics of Ferroelectrics", Gordon and Breach Science Publishers, London, 1976.

Appendix B

Process Flow for the Fabrication of Microwave Ferroelectric Devices

The following steps are on silicon wafer:

- 1. **Substrate**: Silicon wafer; in this work: p-type (100) Si, resistivity= $2-5\Omega$ -cm, thickness= $525\pm15\mu$ m; but other common Si substrates can be used as well.
- 2. Coating and baking: 1.4µm Shipley SPR 3012 photoresist, 95°C, 1 min.
- 3. Alignment and exposure: Define the alignment mark for substrate transfer.
- 4. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 5. Alignment and exposure: Define the alignment mark.
- 6. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 7. Si etching: 1µm Si trench etching, Trikon Omega 201 plasma etcher.
- 8. Cleaning: Photoresist strip in oxygen plasma, clean in 100% HNO₃, 10 min,

161

rinse in demineralized water, clean in 65% HNO₃ (110°C), 10 min, rinse in demineralized water, dry.

- 9. Thermal oxidation: 500nm oxidation of silicon, 1100°C, 200 min.
- 10. Ferroelectric layer deposition: 100nm BST by CSD method.
- 11. Coating and baking: 2µm Shipley SPR 3017 photoresist, 95°C, 1 min.
- 12. Alignment and exposure: Open the ground metal contact.
- 13. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 14. Ferroelectric layer wet etching: 100nm BST wet etching, 2.5% HF.
- 15. **Cleaning**: Photoresist strip in acetone (40°C), clean in PRS 3000 (90°C), 20 min, rinse in demineralized water, dry.
- 16. **First metallization**: 4μm Al sputter deposition, room temperature, Trikon Sigma sputter coater.
- 17. Coating and baking: 2µm Shipley SPR 3017 photoresist, 95°C, 1 min.
- 18. Alignment and exposure: Define the ground pattern.
- 19. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 20. Al etching: 4µm Al etch, 25°C, Trikon Omega plasma etcher.
- 21. **Cleaning**: Photoresist strip in oxygen plasma, clean in PRS 3000 (90°C), 20 min, rinse in demineralized water, dry.

The following steps are on AF45 glass wafer:

- 1. Cleaning: Clean in 100% HNO₃, 10 min, rinse in demineralized water, dry.
- 2. Al sputtering: 100nm Al sputtering, Trikon Sigma sputter coater.
- 3. Coating and baking: 1.4µm Shipley SPR 3012 photoresist, 95°C, 1 min.
- 4. Alignment and exposure: Define the alignment mark for substrate transfer.
- 5. **Development**: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 6. Al wet etching: 100nm Al etch, Merck selectipur, 45 sec.
- 7. Cleaning: Photoresist strip in acetone (40°C), clean in 100% HNO₃, 10 min,

rinse in demineralized water, dry.

 Oxide deposition: 3μm PECVD SiO2 deposition on both sides, 400°C. Novellus PECVD system.

The following steps are substrate transfer processing:

- Oxide deposition: 1µm PECVD SiO2 deposition on Si wafer, 400°C. Novellus PECVD system.
- 2. **Primer treatment**: Coat primer on both Si and AF45 wafers, keep in close box for 24 hours.
- 3. Glueing: Manually spread glue on Si wafer.
- 4. **Stick wafers**: Manually stick AF45 wafer on Si wafer. Visually inspect the alignment marks on both wafers.
- 5. **Pressure on wafers**: Supply 0.2 bar pressure on the wafers, 5 min.
- 6. UV light treatment: Supply UV light on the wafers, 5 min.
- 7. Cleaning: Acetone, room temperature, 1 min.
- 8. **Baking**: Bake the wafers on hotplate, 250°C, 2min.
- 9. Cleaning: Clean in 100% HNO₃, 10 min, rinse in demineralized water, dry.

The following steps are on the backside of the silicon side of the glued wafer unless specified:

- 1. Coeating and baking (on front side AF45): 2μm Shipley SPR 3017 photoresist, 95°C, 1 min.
- 2. Oxide removal: 500nm SiO2 etch, BHF (1:7), rinse in demineralized water, dry.
- 3. **Cleaning**: Photoresist strip in acetone (40°C), clean in 100% HNO₃, 10 min, rinse in demineralized water, dry.
- 4. **HF dip**: Dip in 0.55% HF, 30 sec, rinse in demineralized water, dry.
- 5. Si etching: Etch completely the Si wafer, 25% TMAOH (80°C), 14 hours, rinse in demineralized water, dry.

- 6. Cleaning: Clean in PRS 3000 (90°C), 20 min, rinse in demineralized water, dry.
- 7. **Al coating on front side**: For wafer detection in the following steps. 1.8μm Al sputtering, Trikon Sigma sputter coater.
- 8. Coating and baking: 1.4µm Shipley SPR 3012 photoresist, 95°C, 1 min.
- 9. Alignment and exposure: Open the contact for second metal layer.
- 10. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 11. Oxide etching: 500nm SiO2 wet etching, BHF (1:7).
- 12. **Cleaning**: Photoresist strip in acetone (40°C), clean in PRS 3000 (90°C), 20 min, rinse in demineralized water, dry.
- 13. **Second metallization**: 1.4µm Al deposition, using low thermal budget recipe, Trikon Sigma sputter coater.
- 14. Coating and baking: 1.4µm Shipley SPR 3012 photoresist, 95°C, 1 min.
- 15. Alignment and exposure: Define the top metal patterns.
- 16. Development: 115°C 1 min, Shipley MF322 developer, 100°C 1 min.
- 17. Al etching: 1.4 μm Al wet etching, Merck selectipur, rinse in demineralized water, dry.
- 18. **Cleaning**: Photoresist strip in acetone (40°C), clean in PRS 3000 (90°C), 20 min, rinse in demineralized water, dry.

Appendix C

Derivation of effective magnetic susceptibility of ADL

165

In this appendix, we derive the approximate effective magnetic susceptibility of ADL by using the quasi-static approximation of Maxwell's equations. Consider a thin metallic pattern of arbitrary shape under a uniform RF magnetic field. The assumption of uniformity of the magnetic field is justified as the size of the ADL patterns does not exceed the micron range which is several orders of magnitude smaller than the (effective) electromagnetic wavelength at microwave frequencies. Note also that this magnetic field may be the superposition of an externally applied field (e.g., by other current-carrying structures) and an internal field induced by the currents in the patterns themselves.

Due to the magnetic field, currents will be induced inside the metallic pattern according to the Lenz's law. Suppose the thickness of the metallic layer is much smaller than the skin depth at frequency of interest, like the thin Al layer in our ADL structure. The density of the eddy currents induced will be uniform along the thickness of the metal. Besides, one may neglect the vertical component of the current density vector so that a two dimensional approximation will suffice. In Fig. C.1, we show a single micro-patterned ADL under a uniform RF magnetic field. The coordinate system used in

the following derivation is also shown.

Next, consider the Maxwell equations in frequency domain. By combining the curl equation of electric field and the ohm's law inside the metallic pattern, we have,

$$\frac{\partial J_y}{\partial x} - \frac{\partial J_x}{\partial y} = -j\omega\sigma_M B_z, \qquad (C.1)$$

where $J = \hat{x}J_x + \hat{y}J_y$ is the current density within the metallic region, ω is the angular

frequency, σ_M is the conductivity of the metal and B_z is the normal component of the magnetic field. Inside the metallic region (Ω in Fig. C.1) where there are no volume charges, the current continuity equation reads,

$$\nabla \cdot \boldsymbol{J} = \frac{\partial J_x}{\partial x} + \frac{\partial J_y}{\partial y} = 0.$$
 (C.2)

Since J is divergence free and has only x and y components, it can be represented in terms of a scalar potential ψ :

$$\boldsymbol{J} = \hat{\boldsymbol{x}} \frac{\partial \psi}{\partial y} - \hat{\boldsymbol{y}} \frac{\partial \psi}{\partial x} \,. \tag{C.3}$$

Substituting eq. (C.3) into eq. (C.1), one ends up with a Poison equation for ψ :

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}\right)\psi = j\omega\sigma_M B_z.$$
 (C.4)

In order to find a unique solution, this equation has to be supplemented by boundary conditions for the potential function at the edges of the metallic pattern. Since the structure is small compared to the electromagnetic wavelength, we adopt the quasi-static approximation in which the accumulation of electric charge on the surface of a conductor are neglected. Thus, we require the normal component of the current density vector to vanish on the boundary curve of the pattern which we denote as C. Now we define two direction unit vectors \hat{n} and $\hat{\tau}$, which are the unit vectors normal and along the boundary C, as shown in Fig. C.1. By doing so, the boundary condition reads,

$$\hat{\boldsymbol{n}} \cdot \boldsymbol{J} = 0$$
 along C. (C.5)

Substituting eq. (C.3) into eq. (C.5) and using the relation between the two unit vectors, one finds,

$$\hat{\boldsymbol{\tau}} \cdot \nabla \boldsymbol{\psi} = 0$$
 or $\frac{\partial \boldsymbol{\psi}}{\partial \boldsymbol{\tau}} = 0$ along C. (C.6)

This implies that the potential function ψ does not change on C, thus it can be taken to be a constant on this boundary curve. For a single pattern, the value of this constant will be arbitrary as it will neither affect the Poisson equation nor the current density. To simplify the calculation we demand $\psi = 0$ on C. The Poison's equation eq. (C.4) combined with this boundary condition can then be solved numerically.



Figure C.1 A single ADL pattern under normal magnetic field. Also the coordinate system is shown.

Once this equation is solved, we consider the magnetic moment induced by the flow of the eddy current calculated above in a single metallic pattern, which is given by [1],

$$\boldsymbol{m} = \frac{1}{2} \int_{V} \boldsymbol{r} \times \boldsymbol{J} d\boldsymbol{v} \,. \tag{C.7}$$

Substituting eq. (C.3) into eq. (C.7), one finds,

$$\boldsymbol{m} = -\hat{\boldsymbol{z}} \frac{t}{2} \int_{\Omega} \boldsymbol{r} \cdot \nabla \boldsymbol{\psi} dA , \qquad (C.8)$$

where dA is differential area on Ω . Using the fact that $\mathbf{r} = \frac{1}{2}\nabla r^2$, and Green's first identity [1] combined with eq. (C.4), we have

$$\boldsymbol{m} = -\hat{\boldsymbol{z}}\frac{t}{4}\left[\oint_{C} r^{2}(\hat{\boldsymbol{n}}\cdot\nabla\psi)dl - j\omega\sigma_{M}B_{z}\int_{\Omega}r^{2}dA\right].$$
 (C.9)

Note that in this equation, *m* linearly depends on B_z as ψ linearly depends on B_z

which was considered to be uniform.

Next, we calculate the magnetization of ADL. To that end, we define a volume filling factor f, which is the ratio of the volume of one ADL pattern to that of a unit cell. The magnetization then reads,

$$\boldsymbol{M} = f \frac{\boldsymbol{m}}{V_s}, \qquad (C.10)$$

where V_s is the volume of a single ADL pattern and is given by

$$V_s = t \int_{\Omega} dA \,. \tag{C.11}$$

The magnetization is also in the z-direction and a linear function of B_z . Let γ denote the ratio $\mu_0 M_z / B_z$. For a square pattern with the side D, $\gamma = -j\eta_M f \omega \mu_0 \sigma_M D^2$ where η_M is a dimensionless, geometry-independent factor which is calculated by solving the Poison's equation.(C.4)

We now have a relationship between the volume magnetization of an ensemble of patterns on the one hand, and the normal component of the RF magnetic field on the other hand. The effective susceptibility of ADL in z direction $\chi_{\perp} = M_z / H_z$ can now be evaluated with the help of

$$\boldsymbol{B} = \boldsymbol{\mu}_0 \left(\boldsymbol{H} + \boldsymbol{M} \right). \tag{C.12}$$

which leads to $\chi_{\perp} = \gamma / (1 - \gamma)$.

Note that we did not consider the case where the magnetic field is tangential to the plane of the pattern. This is because the patterns are very thin and the eddy current induced will not be significant in this case.

References

[1] D. Jackson, "Classical Electrodynamics 3rd Edition", Wiley, New York, 1999.

169

Appendix D

Derivation of shielding effect of ADL

In this appendix, we theoretically investigate the substrate shielding effect of an artificial dielectric layer (ADL) inserted in between a CPW transmission line and the Si substrate. The geometry of the problem is illustrated in Fig. (D.1). The calculation is based on the quasi-TEM approximation in which the electric and magnetic fields surrounding the transmission line are separately treated using the electro-, respectively, magneto-static approximations. This is justified for CPW lines at not too high frequencies, i.e. the frequency range of our interest.

In the quasi-TEM approach the transverse electric field can be expressed as the gradient of an electrostatic potential $\varphi(x,y)$ on the line cross section, which is governed by Laplace equation,

$$\nabla \cdot (\overline{\overline{\varepsilon}} \cdot \nabla \varphi) = 0, \qquad (D.1)$$

where the second rank permittivity tensor $\overline{\overline{\epsilon}}$ has been used for the sake of generality. In the Si and oxide regions, where permittivity is a scalar, eq. (D.1) reads

$$\left(\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}\right)\varphi(x, y) = 0, \qquad (y \le 0, t \le y \le t + h) \tag{D.2}$$

where t is the thickness of the ADL and h is the thickness of the oxide layer. In ADL, due to its anisotropy of the dielectric properties, the permittivity is represented as a second rank tensor, and Laplace equation (D.1) reads,



Figure D.1 Geometrical layout of CPW with ADL shield. The thickness of the ADL and oxide are t and h respectively.

$$\left(\frac{\varepsilon_{\parallel}}{\varepsilon_{\perp}}\frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}\right)\varphi(x, y) = 0, \qquad (0 \le y \le t), \qquad (D.3)$$

where ε_{\parallel} and ε_{\perp} are the in-plane (in x-direction) and out-of-plane (in y direction) permittivity respectively.

To simplify the analysis, we assume an unbounded substrate (in x-direction) and transform eq. (C.2-3) to the spectral domain by applying a Fourier transform with respect to x to the potential:

$$\tilde{\varphi}(k,y) = \int_{-\infty}^{\infty} \varphi(x,y) \exp(jkx) dx$$
 (D.4)

where k is the wave number, The equations for the Fourier transformed potential

 $\tilde{\varphi}(x,y)$ are

$$(-k^2 + \frac{\partial^2}{\partial y^2})\tilde{\varphi}(k, y) = 0, \qquad (y \le 0, t \le y \le t + h)$$
(D.5)

$$(-q^{2} + \frac{\partial^{2}}{\partial y^{2}})\tilde{\varphi}(k, y) = 0, \qquad (0 \le y \le t)$$
(D.6)

with $q = k(\varepsilon_{\parallel} / \varepsilon_{\perp})^{1/2}$. The Fourier transformed electric field can be written in terms of $\tilde{\varphi}(k, y)$ as

$$(\tilde{E}_x, \tilde{E}_y) = (ik, -\frac{\partial}{\partial y})\tilde{\varphi}(k, y).$$
 (D.7)

Since the thickness of the Si substrate (>500 μ m) is much larger than the signal line width and signal to ground spacing of the CPW lines under consideration (tens of μ m), the Si substrate is assumed to be infinitely thick for simplicity. This leads to the following boundary condition for the potential

$$\tilde{\varphi}(k, y = -\infty) = 0 \tag{D.8}$$

Let us denote the electrostatic potential on top of the oxide surface by $\varphi(x,t+h) = V(x)$ which reads in its Fourier transformed form,

$$\tilde{\varphi}(k,t+h) = \int_{-\infty}^{+\infty} V(x) \exp(jkx) dx.$$
 (D.9)

By combining eq.(D.8-9) and the boundary conditions of the continuity of tangential electric field \tilde{E}_x and normal electric displacement field \tilde{D}_y at Si/ADL and ADL/oxide interfaces, one can solve the spectrum domain Laplace equation (D.5-6) in the three regions. The Fourier transformed electrostatic potential $\tilde{\varphi}_{si}(k, y)$ in Si substrate then reads,

$$\tilde{\varphi}_{\rm si}(k,y) = \frac{\exp(|k|y) \int_{-\infty}^{\infty} V(x) \exp(ikx) dx}{A(k) \cosh(|k|h) + B(k) \sinh(|k|h)} \tag{D.10}$$

$$A(k) = \cosh(|q|t) + (\varepsilon_{\rm si} / \varepsilon_{\rm e})\sinh(|q|t)$$
(D.11)

$$B(k) = \varepsilon_{\text{ox}}^{-1} \left[\varepsilon_{\text{si}} \cosh(|q|t) + \varepsilon_{\text{e}} \sinh(|q|t) \right], \qquad (D.12)$$

where ε_{ox} is the dielectric constant of the oxide layer, $\varepsilon_e = (\varepsilon_{\parallel} \varepsilon_{\perp})^{1/2}$, and $\varepsilon_{si} = \epsilon_{si} - j\sigma_{si} / \omega$ with

 ϵ_{si} the dielectric constant and σ_{si} the conductivity of Si.
Appendix E

Extraction of Transmission Line Parameters

In this appendix, the method we adopt to extract the transmission line parameters from the scattering (S-) parameter measurements or simulations throughout this work is briefly outlined. For the measurement, a HP-8510 network analyzer, after a line-reflect-match (LRM) calibration procedure, was used to obtain two-port S-parameters. The S-parameters were transformed to impedance (Z-) parameters by the following relation

$$[Z] = Z_s([U] + [S])([U] - [S])^{-1},$$
(E.1)

where Z_s is the system impedance (50 Ω in our case), [Z], [U] and [S] are the 2x2 impedance, unitary and S-parameter matrices respectively [1]. The 2x2 admittance matrix

$$[Y] = [Z]^{-1}. (E.2)$$

For a transmission line of length l with characteristic impedance of Z_0 and the complex propagation constant γ , we define two impedance values of interests, i.e., Z_{open} and Z_{short} to be the input impedances from one port when the other port is open and short circuitry respectively. It is trivial to show that Z_{open} and Z_{short} can be extracted from the impedance and admittance matrices of the transmission line respectively as

$$Z_{\text{open}} = Z_{11} = \frac{Z_0}{\tanh(\gamma l)}, \qquad (E.2)$$

$$Z_{\text{short}} = \frac{1}{Y_{11}} = Z_0 \tanh(\gamma l)$$
 (E.3)

Therefore, the characteristic impedance Z_0 and the propagation constant γ can be expressed as [2]

$$Z_0 = \sqrt{Z_{\text{short}} \cdot Z_{\text{open}}} , \qquad (E.4)$$

$$\gamma = \frac{1}{l} \operatorname{arctanh}(\sqrt{Z_{\text{short}} / Z_{\text{open}}}).$$
(E.5)

The attenuation (α) and propagation (β) constants can then be found from

$$\gamma = \alpha + j\beta , \qquad (E.6)$$

while the attenuation constant α is usually translated into dB form, i.e.

$$\alpha = (20\log e)\operatorname{Re}\gamma . \tag{E.7}$$

Fig. E.1 shows an equivalent circuit model of a transmission line. Its parameters can be extracted from

$$\gamma = \sqrt{(R + j\omega L) \cdot (G + j\omega C)} \quad , \tag{E.8}$$

$$Z_0 = \sqrt{(R + j\omega L) / (G + j\omega C)}, \qquad (E.9)$$

by

$$R = \operatorname{Re}(\gamma \cdot Z_0), L = \operatorname{Im}(\gamma \cdot Z_0) / \omega, G = \operatorname{Re}(\gamma / Z_0), C = \operatorname{Im}(\gamma / Z_0) / \omega. \quad (E.10)$$

where ω is the angular frequency. The transmission line parameters thus can be extracted from S-parameter measurement. It is worth mentioning that, for CPW transmission line on Si, the series resistance *R* models the metal loss of CPW signal line and the substrate loss is counted by the shunt conductance term *G*.



Figure E.1: Equivalent circuit model of a transmission line. L, R, C and G denote the series inductance, resistance and shunt capacitance, conductance (all per unit length), respectively.

References

- [1] D. Pozar, "Microwave Engineering", Wiley, New York, 2005.
- [2] Y. Zhuang, M. Vroubel, B. Rejaei, E. Boellaard, J.N. Burghartz, "Investigation of microstrips with NiFe magnetic thin film (I): experiment", Trans. Magn. Soc. Japan, Vol. 2, No. 5, pp. 367-370, 2002.

Summary

Title: Ferroelectric Materials and Artificial Dielectric Layer Structures for Microwave Integrated Circuit Technologies

By: Yue Ma

In traditional hybrid microwave circuits, microwave passive components like capacitors and inductors are directly built on the interconnect board. This leads to high total cost of microwave circuits and hinders their miniaturization. For that reason, monolithically integrated passive components are gradually replacing the discrete ones, but are limited by their large chip area consumption, high RF/microwave losses and limited functionality. This thesis is dedicated to address these issues by applications of ferroelectric materials and artificial dielectrics.

Chapter 1 provides an overview of challenges associated with passive components implemented within the standard Silicon (Si) IC technologies. Ferroelectric materials and artificial dielectric materials are then introduced and applications to microwave components are reviewed.

In Chapter 2, the preparation of BST ferroelectric thin films by the chemical solution deposition method is presented. Morphological and electrical characteristics of deposited BST thin films on a platinized Si substrate are experimentally studied. Substrate transfer technique is proposed for microwave tunable components which eliminates the use of platinum, a metal which is incompatible with the mainstream Si technology. Possible causes of the short circuited devices fabricated by the substrate transfer method are discussed and recommendations for further research are given.

Chapter 3 deals with the tunability enhancement of ferroelectric-based components by

177

smart device structure design, i.e. the implementation of nano-dot electrodes. Physical principles and theoretical models are presented for nano-dot varactors based on an electrostatic approximation. It is shown that the electric field near the nano-dot electrode is mainly determined by the dot size, rather than the voltage applied to it. Therefore, the tunability is enhanced by the large electric field near the nano-dot electrode. A more realistic varactor design by forming a nano-dot array is proposed and numerical simulations are carried out which show a high tuning range with a relatively low applied voltage.

In Chapter 4, a new artificial dielectric layer (ADL) structure is introduced which contains a conventional thin dielectric film sandwiched between two patterned metal layers. An effective medium model of the ADL structure is derived. The principles of the shielding effect of ADL on CPW transmission lines are shown by solving the Fourier transformed Laplace equations, which is based on a quasi-TEM approximation. It is also experimentally shown that by micro-patterning of ADL or stacking a few ADL layers, the shielding effect can be further increased. Applications of ADL shield, such as directional couplers and spiral inductors are also experimentally studied.

In Chapter 5, a new microwave resonator based on ADL is proposed and theoretically investigated. An equivalent circuit model is developed by solving the Maxwell's equations using Green's functions. Experimental results are compared with full wave simulation of the proposed ADL resonator. The degradation of the quality factor achieved by experiments compared with simulation is attributed to the resistance loss within ADL.

Chapter 6 outlines the major conclusions of this work and recommendations for possible future research topics related to this thesis.

Samenvatting

Titel: Ferroelectric Materials and Artificial Dielectric Layer Structures for Microwave Integrated Circuit Technologies

Door: Yue Ma

Passieve microgolf componenten, zoals condensatoren en inductoren, worden in traditionele hybride microgolf circuits direct op de printplaat geplaatst. Dit gaat gepaard met hoge kosten en bemoeilijkt miniaturisatie. Om deze reden vervangen monolithisch geïntegreerde passieve componenten steeds meer de discrete componenten. Echter, monolithisch geïntegreerde passieve componenten vereisen een groot chip oppervlak, hebben een groot RF/microgolf verlies en hebben beperkte functionaliteit. Deze thesis beschrijft hoe door het toepassen van ferroelektrische materialen deze nadelen kunnen worden voorkomen.

Hoofdstuk 1 geeft een overzicht van de uitdagingen voor passieve componenten die gemaakt worden met standaard IC technologie op basis van silicium (Si). Ferroelektrische materialen en kunstmatige diëlectrica worden vervolgens geïntroduceerd en hun toepassingen in microgolf componenten besproken.

In hoofdstuk 2 wordt de fabricage van dunne BST ferroelektrische lagen met chemical solution deposition gepresenteerd. Morfologische- en elektrische karakteristieken van gedeponeerde BST lagen op een platinized silicium substraat worden experimenteel bestudeerd. Voor variabele microgolf componenten wordt een substraat overdracht techniek voorgesteld die het gebruik van platinized silicium substraten voorkomt. Platina is namelijk een metaal dat niet compatibel is met gangbare Si technologie. Mogelijke oorzaken van kortsluiting in devices die zijn gefabriceerd met de substraat transfer techniek worden besproken. Ook worden aanbevelingen gedaan voor toekomstig

onderzoek.

Hoofdstuk 3 gaat over de vergroting van de verstelbaarheid van ferroelektrische componenten door slim device ontwerp, bijvoorbeeld door de implementatie van nano-dot elektrodes. Fysische principes en theoretische modellen gebaseerd op een elektrostatische benadering worden gepresenteerd voor nano-dot varicaps. Aangetoond wordt dat het elektrische veld bij de nano-dot elektrode voornamelijk wordt bepaald door de dot-grootte en in mindere mate door de aangelegde spanning. De instelbaarheid van de varicap wordt hierdoor vergroot. Een ontwerp van een nano-dot varicap array wordt voorgesteld met een capaciteit groot genoeg voor praktische toepassing. Numerieke simulaties laten een groot instelbereik zien bij een relatief klein spanningsbereik.

Hoofdstuk 4 introduceert een nieuwe kunstmatige diëlectricum laag (artificial dielectric layer, ADL) die bestaat uit een conventioneel diëlectricum tussen twee gepatroneerde metaallagen. De ADL structuur wordt gemodelleerd met een macroscopische beschrijving. Gecombineerd met een oplossing van Fourier getransformeerde Laplace vergelijkingen gebaseerd op een quasi-TEM benadering beschrijft dit model het afschermingeffect van de ADL structuur op CPW transmissielijnen. Experimenteel wordt aangetoond dat met het patroneren van de ADL, of het stapelen van een verscheidene ADL lagen, de afscherming verder kan worden verhoogd. Verder worden toepassingen van ADL afscherming in één-richtings transmissie koppelingen en spiraalinductoren experimenteel bestudeerd.

In hoofdstuk 5 wordt een nieuwe op ADL gebaseerde microgolf resonator voorgesteld en theoretisch onderzocht. Een model van een equivalent circuit wordt ontwikkeld middels het oplossen van de Maxwell vergelijkingen met behulp van Green's functies. Experimentele resultaten worden vergeleken met simulatie van de voorgestelde ADL resonator. Het verschil tussen de experimenteel bepaalde kwaliteitsfactor en de gesimuleerde kwaliteitsfactor wordt toegeschreven aan het weerstandsverlies binnen de ADL.

Hoofdstuk 6 geeft een overzicht van de belangrijkste conclusies en geeft aanbevelingen voor mogelijk toekomstig onderzoek dat is gerelateerd aan deze thesis.

Acknowledgements

This thesis is the result of my research in the High-frequency Technologies and Components (HiTeC) group, Delft University of Technology. Looking back on the four exiting and enjoyable years of my PhD period, I would like to take this opportunity to express my gratitude to many people for their encouragement, contributions and support.

First of all, I would like to thank Prof. Behzad Rejaei, my supervisor and co-promotor, a great source of inspiration, for his encouragements and stimulating suggestions to this thesis. I am deeply impressed by his intelligence to find novel and simple solutions for difficult problems. Behzad, you are the best physicist I've ever met and you never stopped surprising me with your endless brilliant physical insights and rigorous mathematical derivations.

I am very thankful to my promoter, Prof. Joachim N. Burghartz, for offering the position in his group and introducing me into the world of microwave integrated circuits and passive microwave components. Also I can hardly imagine how this thesis would have been finished without your encouragement.

I would like to express my special gratitude to Dr. Yan Zhuang, my former supervisor. Thanks for introducing me to such a warm and productive group. I would never forget the hours we spent together in the clean room and measurement laboratories. I am also grateful that we become life time long friends even after you left for the United States.

Special thanks to all the staff and engineers of the DIMES laboratories. In particular, I would like to thank Atef Akhnoukh for all his help on the high frequency measurements. To Silvana Milosavljevic and Tom Scholtes, with whom I followed the cleanroom courses. I would also like to offer my gratitude to all other engineers of DIMES for all their support during my research work.

It is my great pleasure to work with all my (ex-)colleagues at TU Delft: My former office

mates Pedram Khalili, Huseyin Sagkol and Jacopo Iannacci, as well as Dr. Marina Vroubel, Dr. Leo de Vreede, Jun Tian, Cong Huang, Han Yan, Koen Buisman, Sinaga Saoer, Edmund Neo, Gabriel Macias Montero, Sebastien Sosin, Theodoros Zoumpoulidis, Hsien-Chang Wu and many others.

I am also grateful to all my friends from our small but evolutionary Chinese community, inside and outside the university. In particular, my special thanks go for Qiang Xu, Xiaoli Wang, Ming He, Haiyan Cao, Yu Song, Jie Zhu, Yuhui Wang, Mengyue Wu, Yusong Pang, Hongyan Wang, Jianghua Chen and Wanrong Wang. Thanks for their help on settling my life in the Netherlands and all the pleasant times we've spent together.

Special thanks go for my new neighbors, Annemarie and Andrew, who helped me with the Dutch translation of my propositions and the settling of the new enjoyable life in Delfgauw. Together with Arjen Storm and Vincent Kuiper, my current colleagues, for translating the summary part into Dutch.

Finally, I would like to thank for my family, for their constant support and patience over the past years. Last but not least, my deepest gratitude goes to my wife, Na Li, for all her encouragement, understanding and support, for all that she has given to me, including our beloved daughter.

About the Author

Yue Ma was born on October 16, 1976 in Tianjin, P. R. China. He received the bachelor and master degrees both in materials science and engineering from Tsinghua University in 2000 and 2003 respectively. He worked for Applied Materials China as a process support engineer afterward. In January 2005, he joined the High-frequency Technologies and Components Group in Delft University of technology as a PhD candidate. There he worked on the implementation of ferroelectric and artificial dielectric materials in monolithic microwave integrated circuit technologies. He is working for MAPPER Lithography B.V. as an experimental physicist since March 2009.